ONLINE CONSUMER ENGAGEMENT AND PURCHASE INTENTION: A STUDY ON INDIAN FMCG SECTOR

A THESIS

Submitted to Delhi Technological University for the award of the degree of

DOCTOR OF PHILOSOPHY

By

ANURAG TIRUWA

2K13/PHDDSM/03

Under the guidance of:

PROF. RAJAN YADAV

Professor

Delhi School of Management

Delhi Technological University

PROF. P. K. SURI

Professor

Delhi School of Management

Delhi Technological University



DELHI SCHOOL OF MANAGEMENT DELHI TECHNOLOGICAL UNIVERSITY SHAHBAD DAULATPUR, DELHI-110042 (INDIA) SEPTEMBER 2019

COPYRIGHT NOTICE

© DELHI TECHNOLOGICAL UNIVERSITY, DELHI 2019 ALL RIGHTS RESERVED

CANDIDATE'S DECLARATION

I, hereby certify that the thesis titled "Online Consumer Engagement and Purchase Intention: A Study on Indian FMCG Sector", submitted in fulfillment of the requirements for the award of the degree of Doctor of Philosophy is an authentic record of my research work carried out under the guidance of Prof. (Dr.) Rajan Yadav and Prof. (Dr.) P. K. Suri. Any material borrowed or referred to is duly acknowledged.

The matter presented in this thesis has not been submitted elsewhere in part or entirely to any other University or Institute for the award of any degree.

ANURAG TIRUWA

2K13/PHD/DSM/03 Delhi School of Management Delhi Technological University

DELHI SCHOOL OF MANAGEMENT DELHI TECHNOLOGICAL UNIVERSITY SHAHBAD DAULATPUR, MAIN BAWANA ROAD DELHI-110042 (INDIA)



SUPERVISOR'S CERTIFICATE

This is to certify that the thesis titled "Online Consumer Engagement and Purchase Intention: A Study on Indian FMCG Sector", submitted in fulfillment of the requirements for the award of the degree of Doctor of Philosophy is an original research work carried out by Mr. Anurag Tiruwa, under our supervision. The matter presented in this thesis has not been submitted elsewhere in part or fully to any other University or Institute for the award of any degree, to the best of our knowledge.

Prof. (Dr.) Rajan Yadav

Professor Delhi School of Management Delhi Technological University

Prof. (Dr.) P. K. Suri

Professor Delhi School of Management Delhi Technological University

ACKNOWLEDGMENTS

I want to express my deepest gratitude to my guides and mentors Prof. Rajan Yadav and Prof. P. K. Suri for their support and patience for the duration of my research work. This research would not have been in its present form without their active interest and constant guidance. I thank them for not only making a better researcher but also for making me a better person.

I would also like to thank the faculty members of Delhi School of Management who mentored me well during my post-graduation from Delhi School of Management. Also, I thank my fellow research scholars in Delhi School of Management, who have been a constant source of support and learning during the length of this research. I am also grateful to the library committee, admin committee and DSM staff for their undying support.

Most importantly, I would like to thank my family: my father Dr. R. P. Tiruwa, mother Rachna Tiruwa and brother Ankur Tiruwa for their continuous support, patience, help, and encouragement in completing this task which could not have been possible otherwise.

I dedicate this research work to my friend Sonu.

ANURAG TIRUWA

EXECUTIVE SUMMARY

Innovations in web-based technologies have transformed the earlier existing web applications into a new media, leading to an increase in Internet users. It has also been expedited significantly by the advancements in mobile technologies which have changed the way people communicate with each other. Web 2.0 has enabled the sharing of user-generated content on platforms like blogs, wikis, Online Brand Communities, etc. In the last decade, social media has become a phenomenon which has taken over the globe. Social networking sites like Facebook, YouTube, Twitter, Pinterest, etc., have become quite a hit among the worldwide population. This phenomenon has brought about three significant effects. First, the web has become a platform where a large number of activities are now aimed towards consumers. Second, value creation is now shifted from the marketers to the consumers. Third, the consumer has more power than the marketers. These three effects have changed the dynamics between the brands and the consumers from the point of marketing activities. It has put the consumer in a position to hold the power which seated with the brands earlier.

These platforms have made it easy for the marketers to communicate and engage with the consumers more closely as compared to the traditional media. Digital media in the form of websites, blogs, social media platforms and social networking sites have created an ecosystem that makes it possible for the brands to involve and include consumer in interactive communication. Studies have shown that the engagements on social media platforms are beneficial as they provide real-time analytics to understand the sentiment and attitude of the consumers. Brands and marketers are actively pursuing consumers on these platforms to connect and communicate to create a positive perception for keeping the consumer engaged with the brand. Social media platforms have transformed in past years into potential communication channels for the brands to use for marketing and promotion. These Social Networking Sites provide a plethora of opportunities for the user to create and share content as

well as information by creating Online Brand Communities. These Online Brand Communities have become quite active in recent years in a manner that brands and marketers have started to establish brand run communities on Social media platforms. Marketers and academicians have explored the dynamics of Social Networking Sites from the perspective of Online Brand Communities. Due to the dynamic nature of these platforms and the ongoing innovations, there is a need for continual research on the same.

Previous researches have explored the engagements on Facebook Brand Pages from the point of functional engagement elements (likes, share and comments). These studies were in the context of western countries, and therefore their implications might have little or no relevance in the Indian context. Therefore, this study is focused on Indian FMCG brand run Facebook Brand Pages that have the same dynamics as that of an Online Brand Community. It attempts to carry forward the research done in the same area by proposing a framework to understand the consumer's behaviour in a Facebook Brand Page. For this purpose, important factors were identified which affects the consumer's attitude and brand engagement in a brand run Facebook Brand Page.

Further, the study also made attempts to explore, whether attitude and brand engagement has some effect on consumer's purchase intention. The results of the study show that the factors identified have a significant effect on consumer's brand engagement on Facebook along with few factors significantly affecting their attitude towards the brand. The study also shows that there is a significant effect of the Facebook brand engagement on consumers purchase intention. The results of the study were also corroborated with a case study on Patanjali, an emerging Indian FMCG brand in India. The study is beneficial for brands, marketers and small and medium enterprise that can use the input for the better understanding of online consumer engagement on Facebook Brand Pages in the context of the Indian market and its consumers.

vi

TABLE OF CONTENT

Title	Page No.
Copyright Notice	i
Candidate's Declaration	ii
Supervisor's Certificate	iii
Acknowledgements	iv
Executive Summary	V
Table of Content	vii
List of Tables	xii
List of Figures	xiii
List of Abbreviations	xvi
Chapter - 1: Introduction	1
1.1 Introduction	1
1.2 Emergence of Online Brand Communities	2
1.3 Social Networking Site engagement and integration into business	3
1.4 Facebook Brand Pages and Online Consumer Engagement	5
1.5 Research questions	6
1.6 Research objectives	7
1.7 Relevance of the study	8
1.8 Scope of the study	8
19. Outline of the thesis	8
Chapter - 2: Literature Review	11
2.1 Introduction	11
2.2 Social Media	14
2.2.1 Definition	16
2.2.2 Types of Social Media	17
2.2.3 Use in Business	23
2.3 Online Customer Engagement (OCE)	29
2.3.1 Definition	30

2.3.2 Types of Online Consumer Engagement (OCE)	31
2.3.3 Online Consumer Engagement (OCE) strategies	34
2.4 Online Brand Communities (OBCs)	36
2.4.1 Types of Online Brand Communities (OBCs)	39
2.4.2 Consumer Engagement (CE) in Online Brand Communities (OBCs)	40
2.4.3 Drivers for Consumer Engagement (CE)	42
2.4.4 Process of Consumer Engagement (CE) in Online Brand	44
Communities (OBCs)	
2.5 Facebook Brand Pages (FBPs) and Consumer Engagement (CE)	45
2.5.1 Facebook Brand Page (FBP) engagement - benefits to brand/marketers	47
2.6 Research Gaps	50
2.7 Conclusion	51
Chapter - 3: Indian FMCG Sector & Online Consumer Engagement	52
3.1 Introduction	52
3.2 Indian FMCG Sector – an overview	52
3.3 FMCG brands and Online Consumer Engagement on Social Media platforms	55
3.4 Digital Strategies Followed by Indian FMCG Brands	57
3.5 Conclusion	67
Chapter – 4: Research Methodology	68
4.1 Introduction	68
4.2 Hypotheses development	68
4.2.1 Brand Attachment (BA)	69
4.2.2 Communication (COMM)	70
4.2.3 Content (C)	71
4.2.4 Perceived Benefits (B)	72
4.2.5 Credibility (CRED)	73
4.2.6 Critical Mass (CM)	74
4.2.7 Attitude Towards Brand (ATB)	76

4.2.8 Brand Engagement (BE)	77
4.3 Proposed framework	78
4.4 Hypotheses	78
4.5 Research design	79
4.6 Research instrument	80
4.7 Sampling technique	81
4.8 Data collection	82
4.9 Statistical techniques and tools for data analysis	83
4.10 Conclusion	83
Chapter – 5: Data Analysis and Results	84
5.1 Introduction	84
5.2 Demographic profile of respondent	84
5.3 Pilot study	86
5.4 Questionnaire reliability	86
5.5 Testing the assumptions for multivariate analysis	87
5.5.1 Missing Data	87
5.5.2 Outliers	87
5.5.3 Normality	88
5.5.4 Linearity	88
5.5.5 Homoscedasticity	89
5.5.6 Multicollinearity	98
5.6 Confirmatory Factor Analysis (CFA)	98
5.6.1 Convergent and discriminate validity	98
5.6.2 Measurement model	100
5.7 Structural model	106
5.8 Group Differences: Gender, Age, Annual Income, Weekly Internet Usage	110
5.8.1 Gender	111

5.8.2 Age	111
5.8.3 Annual family income	112
5.8.4 Weekly Internet usage	113
5.9 Conclusion	114
Chapter - 6: A Case Study on Patanjali Ayurved Limited	116
6.1 Introduction	116
6.2 Patanjali Ayurved Limited – an overview	117
6.3 Promotional strategies	119
6.4 Social Media strategy for Online Consumer Engagement	123
6.4.1 Patanjali on YouTube	124
6.4.2 Patanjali on Instagram	127
6.4.3 Patanjali on Twitter	128
6.5 PAL's Facebook Brand Page (FBP) engagements	130
6.6 Evaluation of Patanjali's Facebook Strategy	139
6.7 Conclusion	142
Chapter – 7: Findings and Discussions	144
7.1 Introduction	144
7.2 Discussion on SEM Results	145
7.2.1 Relationships between Brand Attachment, Communication, Content,	145
Perceived Benefits, Credibility, Critical Mass and Attitude Towards Brand	
7.2.2 Relationships between Brand Attachment, Communication, Content,	147
Perceived Benefits, Credibility, Critical Mass and Brand Engagement	
7.2.3 Relationships between Attitude Towards Brand and Brand Engagement	149
7.2.4 Relationships between Attitude Towards Brand, Brand Engagement,	150
Brand Attachment, Critical Mass and Purchase Intention	
7.3 Group differences between Gender, Age, Annual Family Income and Weekly Internet Use	152

7.4 Conclusion	159
Chapter – 8: Conclusion and Implications	160
8.1 Introduction	160
8.2 Conclusion of the study	160
8.3 Contribution of the study	162
8.4 Implications of the study	163
8.4.1 To practitioners/marketers/brands/SMEs	163
8.4.2 To academicians/researchers	164
8.5 Limitations of the study	165
8.6 Future scope of study	165
References	166
APPENDIX A	192
APPENDIX B	198
APPENDIX C	201

Title	Page No.
Table 2.1: Drivers of CE	43
Table 3.1: Major FMCG companies in 2018	53
Table 3.2: Major FMCG companies and their Digital Presence and Followers	56
Table 4.1 Constructs and their sources	80
Table 4.2 Selection of sample size based on the number of constructs and items	81
Table 5.1: Demographic Details	85
Table 5.2: Construct reliability	87
Table 5.3: Skewness and Kurtosis Statistics	88
Table 5.4: Collinearity Statistics	98
Table 5.5: Reliability and validity of constructs	99
Table 5.6: Model Fit Indices	101
Table 5.7: Factor loading for the items	104
Table 5.8: Structural results	108
Table 5.9: Group differences within Gender	111
Table 5.10: Group differences within Age	112
Table 5.11: Group differences within Annual Income	113
Table 5.12: Group differences within Weekly Internet Use	114
Table 6.1: Patanjali vs. Other Brands	117
Table 6.2: Patanjali Ayurved Limited Digital and Social Media Presence	123
Table 6.3: Results of the study and the findings of the Case	141

LIST OF TABLES

Title	Page No.
Figure 2.1: Flow of Review of Literature	13
Figure 2.2: Different Dimensions of Social Media	15
Figure 2.3: Social Media Landscape	23
Figure 2.4: Choosing the Right Social Media Platform for Business	28
Figure 2.5: Types of Online Engagements	33
Figure 2.6: Types of OBCs based on initiation	39
Figure 2.7: Types of OBCs based on orientation	40
Figure 2.8: Process of CE	45
Figure 3.1: Amul's Take on the 2020 US Presidential Candidates	59
Figure 3.2: Amul's Take on Amitabh Bachchan's 50 Year in Indian Cinema	60
Figure 3.3: Blog post written by a food influencer to promote Cavin's Milkshake	61
Figure 3.4: Facebook Chat application for users to connect with Colgate India FBP	63
Figure 3.5: Backlinks to Dabur's corporate website on its FBP's about section	64
Figure 3.6: The Twitter post directing the user to Amazon through a Link in the twee	et 65
Figure 3.7: Patanjali's post of its official Facebook brand page for Chennai region	66
Figure 4.1: Proposed Research Framework	78
Figure 5.1: Scatter Plot diagram	89
Figure 5.2: Scatter plot of Perceived Benefits and Attitude Towards Brand	90
Figure 5.3: Scatter plot of Brand Attachment and Attitude Towards Brand	90
Figure 5.4: Scatter plot of Communication and Attitude Towards Brand	91
Figure 5.5: Scatter plot of Content and Attitude Towards Brand	91
Figure 5.6: Scatter plot of Credibility and Attitude Towards Brand	92
Figure 5.7: Scatter plot of Critical Mass and Attitude Towards Brand	92
Figure 5.8: Scatter plot of Perceived Benefits and Brand Engagement	93
Figure 5.9: Scatter plot of Brand Attachment and Brand Engagement	93

LIST OF FIGURES

Figure 5.10: Scatter plot of Communication and Brand Engagement	94
Figure 5.11: Scatter plot of Content and Brand Engagement	94
Figure 5.12: Scatter plot of Credibility and Brand Engagement	95
Figure 5.13: Scatter plot of Critical Mass and Brand Engagement	95
Figure 5.14: Scatter plot of Brand Attachment and Purchase Intention	96
Figure 5.15: Scatter plot of Attitude Towards Brand and Purchase Intention	96
Figure 5.16: Scatter plot of Brand Engagement and Purchase Intention	97
Figure 5.17: Scatter plot of Critical Mass and Purchase Intention	97
Figure 5.18: Measurement model	102
Figure 5.19: Measurement model output	103
Figure 5.20: Structural Model	106
Figure 5.21: Structural Model Output	107
Figure 5.21: Final Model	109
Figure 6.1: Sales and Profit Trends of Patanjali Ayurveda Limited	118
Figure 6.2: PAL's corporate web portal	120
Figure 6.3: PAL's E-Commerce portal	121
Figure 6.4: The blog section of the E-Commerce portal	122
Figure 6.5: Patanjali Ayurved, PAL's YoutTube Channel	124
Figure 6.6: Various Playlists on PAL's YouTube Channel	125
Figure 6.7: About page of Patanjali Ayurved YouTube Channel	126
Figure 6.8: patanjaliproducts, Official Instagram page of PAL	127
Figure 6.9: Patanjali Products, official Twitter profile of PAL	129
Figure 6.10: PAL's official Facebook Brand Page (FBP)	130
Figure 6.11: Link to the e-Commerce portal in the About Section of the FBP	131
Figure 6.12: PAL's organizational profile and all the digital portals	131
Figure 6.13: Facebook posts on Patanjali Facebook Brand Page	132
Figure 6.14: Facebook Campaign posts during the Independence Day week	133

Figure 6.15: FBP posts showcasing the employment drives at various states	134
in North India	
Figure 6.16: FBP posts showcasing the cleanliness drive by Patanjali	135
Yogpeeth at Haridwar	
Figure 6.17: Live streaming of Yog Shivir on Patanjali FBP	136
Figure 6.18: Live Streaming of opening ceremony of Patanjali Research Centre	137
Figure 6.19: Patanjali's local FBP for Chennai	137
Figure 6.20: Chat application on Facebook Brand Page used by PAL	138

LIST OF ABBREVIATIONS

- SNS : Social Networking Site
- OBC : Online Brand Community
- OCE : Online Consumer Engagement
- FBP : Facebook Brand Page
- UGC : User Generated Content
- FMCG : Fast Moving Consumer Goods
- ASV : Average Shared Variance
- MSV : Maximum Shared Variance
- AVE : Average Variance Extracted
- CFA : Confirmatory Factor Analysis
- GFI : Goodness of Fit Index
- CFI : Comparative Fit Index
- TLI : Tucker Lewis Index
- RMSEA : Root Mean Square Error of Approximation
- SPSS : Statistical Package for Social Sciences
- SEM : Structural Equation Modelling
- VIF : Variance Inflation Factor
- TAM : Technology Adoption Model
- TAM2 : Technology Adoption Model 2

UTAUT : Unified Theory of Acceptance and Use of Technology

Chapter -1

Introduction

1.1 Introduction

The Innovations in Internet-enabled services have redefined the act of connecting with the consumer. With the introduction of the Internet and the constant advancements in Internet technologies, marketers are presented with new avenues to explore the opportunities created by the Internet in the field of marketing. Consumers now have access to new-age digital applications which are also known as Web 2.0; that have caught the attention of the marketers as well (Constantinides et al., 2008; Berthon et al., 2012). The literature documents the theoretical foundation of Web 2.0 and further describes its main dimensions and elements. Studies discuss the usage of Social Media by the consumers, the challenges and the possibilities that lie ahead for the marketers and strategies for advertising and promotion through Social Media (Berthon et al., 2012). The evolution of Social Networking Sites (SNSs) has generated, evolving and interactive digital media. Users have the capabilities to share, create, discuss and modify user-generated content as well as defining the nature and the extent of the marketing exchange (Kietzmann et al., 2011; Hanna, 2011; Araujo and Neijens, 2011). With Social Media and Social Networking Sites (SNSs) constantly upgrading themselves to cater to the online population worldwide. Previously known wikis and blogs have transformed themselves to resemble these SNSs. The resultant technologies are now taken the shape of today's online pages into online communities like Facebook, Twitter and other SNSs.

Currently, Social Media sites exist as a coherent and diverse ecology of various platforms, most of which started as a closed group or as a closed campus application but now has expanded their horizon to a worldwide population. The rise in these sites has somewhat made consumers the center of corporate communication and has transferred to the consumers, who now hold power through their engagements and content creation on Social Media (Kietzmann et al., 2011; Dwivedi et al., 2015). Thus, making them critical to the products and brands when it comes to word-of-mouth promotion, where the brand is at the passive end of the promotion. This word of mouth communication has been dubbed as the electronic-Word-of-Mouth (eWOM). With online communities are increasing every day where users are actively communicating about a product/brand on the dedicated themes of the communities, it has attracted the attention of marketers. They need to keep a track on such communities to understand what their consumers want and what is their attitude regarding their product and brand.

1.2 Emergence of Online Brand Communities

SNSs now provide a more interactive platform for the online communities to build a page for users to like and engage. These platforms are used by users for discussing, sharing and exchanging information, media and reviews. This has resulted in the emergence of Online Brand Communities (OBCs) that are a platform for masses to mingle with each other and to engage in discussions and information sharing. The dynamics of these online communities are similar to that of word of mouth communication, where people with similar tastes, liking or similar experiences, engage with each other. These OBCs have the same structure and working like that of an online community, and they indulge in discussing and sharing information about a brand and its products.

These communities are present in the virtual world and are accessible to anyone, anywhere. One can be a participant, or one can choose to be a visitor and gain information and knowledge as per the need. Thus, SNSs and communities have given the user the power to discuss, to submit views, reviews, and criticism. Such an ability of the SNSs has become a smart tool for the marketers to integrate them into their business model. An increasing number of marketers are doing the same. Due to the integration of SNSs users discuss and exchange information about products or services they have used or currently using. The negative or positive nature of the sentiments of the user's experience affects the information exchanged.

Internet usage has increased among people and SNSs are increasingly being accessed more and more. They have become a channel which should be considered by marketers, to engage consumers online. Apart from this; OBCs, independent from the brand are in existence which are moderated by private individuals who are loyal to a brand and users of their community to review products or brands. These communities are a significant influence on the users that are searching the web regarding the information on any product or brand. Whatever the other users have discussed or reviewed; first-time users are influenced by that information. Influence also depends on the credibility of the community and the supporting comments or remarks about the information. It is, therefore, important for marketers to understand and to keep a vigil on such communities to understand the consumer's attitude towards the brands.

With these innovations on SNSs, it has become possible for the marketers to understand the psyche of the consumer. Thus, they are now making official brand communities pages on SNSs to keep their consumers engaged and to open a direct channel of communication. This step enables the brands and marketers to understand the sentiments of consumers towards a product and make necessary improvements to the same. Brands have changed their approach from using the Social Media platform from only disseminating information to consumers, to involving the consumers with interactive audio-visual media communication.

1.3 Social Networking Site engagement and integration into business

Social Media is gaining popularity among businesses as a platform to engage consumers effectively. Marketers are redesigning their business models with a focus on consumer connectivity and interactivity. Brands have started to utilise the SNSs and SNS based OBCs to engage consumers. Most of these OBCs are a web page where the brands engage their consumers by posting information about new products, schemes or any upcoming event/festival with which the brand is associating itself. This engagement is through posts which can be supported by media in the form of images, audio and video. With technologies improving over the years, brands have come up with new and improved ways to integrate these technologies into their businesses. Brands can create their page on SNSs for which they are required to make two necessary decisions. First, the brand should ponder if it would like to have a presence on the SNSs. The second factor is the level of engagement within SNSs, which is decided by the brand, i.e., how it will be interacting with the consumer. The brand should determine the structure of communication and define the limits for the same (Araujo and Neijens, 2011). It is also observed that SNSs adoption is initiated when a brand undergoes Internet innovation itself. It is also noted that the brand which is more consumer intensive are more likely to undergo the change mentioned above.

Nowadays, almost every brand has a page dedicated to connecting it with its followers, consumers or users. The brand can directly connect with the masses and can get direct through polls, or a simple question, supported by audio or by a visual medium. This, in turn, results in comments, likes, and favourites on the post by the followers of the brand, followed by sharing of the post by users that engage with the community. Marketers are focussed on this objective to influence the existing consumers and to capture the potential ones. To achieve the above, Social Media is seamless as a platform where content is conjoined with web technology for publishing informational content (Kietzmann et al., 2011). The content found on the SNSs and other web-based platforms that makes the user capable of creating, sharing and recommending meaningful information. Therefore, helping the brands in influencing the general demographics.

This further leads to analytics generated by the user's likes and shares, allowing the brand to process the data accumulated. It also helps to understand the effect of the posts, which is a part

of engaging consumers. Integrating SNSs into the business strategy provides insights into the attitude that consumers of a product have. It allows the brand to focus on improving its image by following the cues from consumers. Also, it acts as a direct channel to communicate with the consumers and receive feedback directly from the end user of any brand. Facebook, YouTube, Twitter, and Pinterest are some of the most used SNSs which have been developed in recent years. These SNSs provide a platform for brand or marketers to form OBCs and engage consumers using Social Media. Brands have been using the sites mentioned above for engaging consumers through the integration of various dedicated media, i.e., short videos which becomes viral, images that have record shares on the SNS network by users, online games which brand roll out for masses to play and be engaged with, etc. These media, in turn, generates an attitude towards the brand/marketer and is helpful in decision making and the intention to purchase the same in the near future.

1.4 Facebook Brand Pages and Online Consumer Engagement

Since its conception in 2005, Facebook has played an essential part in taking the Social Media industry in a new direction. Its various capabilities in terms of user engagements through a number of media uploads (images, videos, links, messages and status messages) makes it a suitable platform to connect with consumers that are active on the said platform. Facebook Brand Pages (FBPs) are similar to the OBCs, where users engage with each other to share information and co-create content. Brands and marketers are using these capabilities of Facebook to form and moderate FBPs and engage the consumer to make them aware and loyal to them. Majority of the brands are actively using Facebook as a platform to connect and engage with consumers. As the Internet is reaching to more and more people, reach of SNSs like Facebook is proportionally increasing.

Brands are making an effort to create a digital presence in the Social Media ecosystem. Majority of the Social Media platform directs the user to Facebook, making it a vital nodal point for brands to consider while designing their digital marketing strategy. Facebook brand pages are not only capable of reaching out to the consumers that are active users on Facebook, but also to establish interactive engagements with them. FMCG brands in India are also using Social Media platforms to connect with consumers. Majority of the brands are active on Facebook to post content to engage consumers on Social Media platforms. This study attempts to understand the online engagements of the FMCG brands in India with a special focus on Facebook Brand Pages run and moderated by the brands.

Studies have tried to understand consumer engagement on Social Media platforms, with a focus on Facebook. Majority of the studies have attempted to understand the engagement from the perspective of engagement elements (likes, comment and share) to measure the effectiveness of any brand page. Studies have discussed consumer engagement from the perspective of Online Brand Communities (OBCs). However, quite a few have addressed the brand communities from the perspective of Facebook. Also, studies related to consumer engagement have been presented on the functional level of various SNS platforms. These studies measure engagements purely by their respective functional engagements (likes, shares, comments, and referrals). There is a lack of studies where the dimensions of Online Consumer Engagement are taken into account in the context of Facebook Brand Pages and empirically tested. This study attempts to work in these areas and fill the research gaps by proposing a framework that identifies and measures the factors that affect the consumer's engagement with FMCG brand run Facebook Brand Pages (FBPs) and to empirically test its effects on consumer's purchase intentions.

1.5 Research questions

The study aims to address the following questions:

Q 1. What are the different engagements on Online Brand Communities?

Q 2. What are the practices followed by Brand for Online Consumer Engagement?

Q 3. What are the factors/dimensions that influence consumer's brand engagement and attitude towards the brand?

Q 4. Does Online Consumer Engagement influence purchase intention?

1.6 Research objectives

This study aims to present Online Consumer Engagement; specifically, from the perspective of 'Brand Attachment', 'Communication', 'Content', 'Perceived Benefits', 'Credibility', and 'Critical Mass' as the antecedents of 'Attitude Towards Brand' and 'Brand Engagement', and their consequent effects on consumer's 'Purchase Intention'. Thus, this study aims to propose a model of online consumer engagement in the context of Facebook brand communities. For the same, the following research objectives are formulated to investigate the research questions discussed in the previous section:

- To Identify the different practices followed by Indian FMCG Brands for Online Consumer Engagement.
- ii. To analyse the strategies followed by the Indian FMCG Brand for Online Consumer Engagement.
- iii. To analyse the influences of consumer's brand engagement and attitude towards the brand on Purchase Intention.
- iv. To analyse the influence of demographics on the antecedents and consequences of Consumer's brand engagement and Attitude towards the brand.

1.7 Relevance of the study

The relevance of the study is discussed in the points that follow:

- i. Studies that focus on the purchase intention initiated by Online Consumer Engagement through SNSs are quite a few.
- This study attempts to understand the influence of Online Consumer Engagement through SNSs on the consumer's intention to purchase a particular brand.
- iii. This study might help the policymakers, researchers, marketers, online communities and other stakeholders to understand the link between Online Consumer Engagement and Purchase Intention.

1.8 Scope of the study

The study is limited to the OBCs or Brand Pages on Social Networking Site Facebook. The focus is on Indian FMCG brand run Facebook Brand Page (FBPs) which are moderated directly by the brand and are directed towards the consumers.

1.9 Outline of the thesis

This research has been divided into eight chapters which have been structured as follows:

Chapter Two: Chapter Two discusses the theoretical base of the study at hand. Existing literature on the research area was done and reported to find a different aspect of Social Media and online consumer engagement. This chapter gives a detailed introduction with a focus on the following research areas.

 Literature review on various concepts behind Online Consumer Engagement and Social Media; along with definitions essential in understanding the essential concepts. ii. Literature review on the concept of Online Brand Communities and, the way they are being used by brands/marketers to connect and attract consumers. Also, the focus is on the concept behind the Facebook Brand Pages and how they function from the perspective of business strategy.

Chapter Three: This chapter presents an overview of the Indian FMCG sector. It also gives brief about the promotional and marketing scenario from the perspective of Social Media by Major Indian FMCG brands. It talks about how Social Media is being used by brands with integration to their marketing and promotional strategies. The chapter also elaborates on the various widely used strategies by the brands on different Social Media platforms to engage with the consumer.

Chapter Four: Chapter Four discusses the proposed framework through a review of the literature, along with the methodology followed in conducting the study at hand. A detailed overview of the proposed hypotheses based on the developed model is also presented in this chapter. This chapter also explains the research design of the study, along with the sampling technique and quantitative data analyses techniques that are employed in the study.

Chapter Five: Chapter Five presents the analysis of the data collected as per the research design mentioned in Chapter Four. This section explains the demographics of the sample and discusses the reliability analysis of the questionnaire through the data collected in the pilot study. It also includes checking of initial validity (discriminant and convergent) through CFA (Confirmatory Factor Analysis) using IBM® SPSS STATISTICS. CFA is followed by the model fit testing and path analysis using structural equation modeling on IBM® SPSS AMOS. Analysis of group differences among different demographic variables is also reported in this chapter.

Chapter Six: Chapter Six presents a case on the FMCG brand Patanjali's ayurvedic venture, Patanjali Ayurved Limited (PAL). It provides an overview of PAL from the context of the Indian FMCG market. It discusses the digital strategy followed by PAL. The case attempts to interpret the results presented in Chapter Five. It reports the PAL's Facebook Brand Page engagements and elaborates on the implications.

Chapter Seven: Chapter Seven discusses the findings of the results discussed in Chapter Five. It explains the results of the data analysis done on the structural model to test the hypothesised relationships established in Chapter Four.

Chapter Eight: Chapter Eight concludes the thesis, along with the implications of the study for brand/marketers, researcher, and academics. In the end, limitations and future scope of the study is reported.

Chapter - 2

Literature Review

2.1 Introduction

Innovations in web-based technologies have evolved the earlier existing web application into a new media, leading to an increase in internet users. It has also been expedited significantly by the advancements in mobile technologies that have changed the way people communicate with each other. Web 2.0 has enabled the sharing of user-generated content on platforms like blogs, wikis, Online Brand Communities, etc. In the last decade, Social Media has become a phenomenon which has taken over the globe. Social networking sites like Facebook, YouTube, Twitter, Pinterest, etc., have become quite a hit among the worldwide population. This phenomenon has brought about three significant effects. First, the web has become a platform where a large number of activities are now centered. Second, value creation is now shifted from the marketers to the consumers. Third, the consumer has more power than the marketers. These three effects have changed the dynamics between the brands and the consumers from the point of marketing activities.

These platforms have made it easy for the markets to communicate and engage with the consumers more closely, as compared to the traditional media. Digital media in the form of websites, blogs, social media platforms and social networking sites have created an ecosystem that makes it possible for the brands to involve and include consumers in communicative dialogue. Studies have shown that the engagements on social media platforms are beneficial as they provide real-time analytics to understand the sentiment and attitude of the consumers. Brands and marketers are pursuing consumers on these platforms to connect and communicate to create a positive perception for keeping the consumer engaged with the brand.

Social Media platforms have transformed in past years into potential communication channels for the brands to use for marketing and promotion. These Social Networking Sites provide a plethora of opportunities for the user to create and share content as well as information by creating Online Brand Communities. These Online Brand Communities have become quite active in recent years in a manner that brands and marketers have started to establish brand run communities on Social media platforms. Marketers and academicians have and are still exploring the dynamics of Social Networking Sites from the perspective of Online Brand Communities. Due to the dynamic nature of these platforms and the ongoing innovations, there is a need for continual research on the same.

Previous researches have explored the engagements on Facebook Brand Pages from the point of functional engagement elements (likes, share and comments). These studies were in the context of western countries, and therefore, their implications might not be applicable in the Indian context. Therefore, this study is focused on Indian FMCG brand run Facebook Brand Pages that have the same dynamics as that of an Online Brand Community. It attempts to carry forward the research done in the same area by proposing a framework to understand the consumer's behavior in a Facebook Brand Page.

This review of literature is done in two sections. First, a review of studies addressing Social Media, Online Consumer Engagement, Online Brand Communities and consumer engagement in Facebook Brand Pages was conducted in the existing body of literature. The review was conducted to explore the topics and establish a connection among the topics from the perspective of the study at hand. Second, a review of studies to identify the crucial factors and establish and develop hypotheses to find relationships among these factors was conducted. Review for the same is reported in Chapter 4 Section 4.2. This chapter presents the first part of the review to identify the gaps in the research to develop the research questions and objectives stated in Chapter 1, Section 1.5 and 1.6, respectively.

The flow of the literature review is as shown in the flow chart, as seen below:



Figure 2.1: Flow of Review of Literature

2.2 Social Media

In recent years Social Media has become a worldwide phenomenon for social interactions. Their emergence was enabled by the advances in mobile and telecommunication technology along with the accessibility to web-based applications with the said technology. They have become a medium which is an amalgamation of both technology and social interaction. Early studies into the Social Media defined it as

"A set of Internet-based applications build on the technological foundations of Web 2.0, and that enables user-generated content to be created and exchanged" (Kaplan and Haenlein, 2011; wpmu, 2017).

The capabilities of Web 2.0 facilitate sharing, participation and collaboration among the participant users. The social web is one of the important elements of web 2.0, which comprises of online platforms that are used by Internet users for sharing their experiences and perspectives. This is enabled by the ability of Social Media to take on different forms such as, blogs, business networks, enterprise social networks, forums, microblogs, photo sharing, products/services review, social bookmarking, social gaming, social networks, video sharing, and virtual worlds (Harris and Rea, 2009; wpmu, 2017; Aichner and Jacob, 2015). As per the ranking provided by Alexa (2017), the top Social Media websites in 2017 were: (1) Facebook; (2) Twitter; (3) LinkedIn; (4) Pinterest; (5) Google+; (6) Livejournal.com; (7) Flickr.com; (8) Fiverr.com; (9) Twitter; and (10) QQ.com (Alexa, 2017). The current Social Media has evolved from simple platforms such as sixdegrees.com (Kirkpatrick, 2011). The Social Network Revolution has led to an incredible rise in social networking sites. It was noted that the users spent about 18 hours a week on social networking sites in 2015 (Nielsen, 2015). In 2017, it was found that the average amount of time that a person spends on Social Media has increased to 50 minutes each day (Tang, 2017). This increase in usage is due to the penetration of smartphones in the majority of the global population (Sterling, 2016).

Due to the plethora of Social Media platforms, marketers are now actively experimenting on several such social platforms. Major consumer-oriented organisations are actively connecting and conversing with their consumers through a variety of Social Networking Sites. Similarly, blogs have provided marketers with the capability of generating sales leads and create a channel to disseminate information about their product or service. Two-way interaction with consumers has become easy due to the penetration of smartphones in the masses. Therefore, it is easy for marketers and organisations to connect more to the consumers using the help of various Social Media platforms (Crittenden et al., 2010). Social Media is changing how the marketing landscape used to look, by changing the way information was sourced and percolated to the masses. These platforms have opened new avenues for creating a collaborative world. They have empowered the consumers to connect, share, and collaborate; thus, creating an influence that may alter how organisations engage with the consumers to influence them (Singh, 2005; Walmsley, 2010).



Figure 2.2: Different Dimensions of Social Media Source: (Reutemann, 2016)

2.2.1 Definition

Technology is changing at a fast pace, so are the capabilities of the evolving stand-alone and built-in Social Media services (Obar and Wildman, 2015). This poses a challenge in introducing a definition that encompasses all the traits of present social networking sites and defines them collectively. Social Media can be broadly defined as platforms with capabilities to connect and bring people together. However, mobile and messaging technologies can be put under the same definition (Schejter and Tirosh, 2015). Some studies have identified the four communalities in the Social Media ecosystem; such as 1) Social Media are Web 2.0 Internetbased applications (Kaplan and Haenlein, 2010; Obar and Wildman, 2015); 2) User Generated Content (UGC) is the lifeblood of the Social Media organism (Kaplan and Haenlein, 2010; Obar and Wildman, 2015); 3) users create service-specific profiles for the site or app that are designed and maintained by the Social Media organisation (Boyd and Ellison, 2007; Obar and Wildman, 2015); 4) Social Media facilitates the development of the social networks through the networks that any user builds through his/her Social Media profile (Boyd and Ellison, 2007; Obar and Wildman, 2015). Alexa, a Web information company that provides website traffic rankings, offers broad categories for characterizing social networks, social networking, and Social Media.

The most fitting definition which clearly defines the Social Media is provided by Merriam-Webster dictionary as follows:

"Forms of electronic communication (such as websites for social networking and microblogging) through which users create online communities to share information, ideas, personal messages, and other content (such as videos)" (Merriam-Webster, 2017)

16

Such platforms allow strangers to meet and visit the profile of users without prior knowledge about them. They enable the users to decide the content of their profiles, which they want to be visible to the strangers. Therefore, most of the time, meetings on these platforms are between "latent ties" that share some offline connection (Haythornthwaite, 2005). Participants within these systems are primarily communicating with people who are already part of their extended offline social circle. Due to this emphasis on social networking and organizing feature of these platforms, these sites can be labeled as "social network sites." These social networking sites can be defined as the web-based services that have capabilities to allow users to create a profile of public or semi-public nature within the boundaries of the system; articulate a list of other users with whom they share a connection and view the records of connections made by others within the system (Boyd and Ellison, 2007). Even though the SNSs are designed to attract and connect a common population, users within a Social Media platform can be segregated based on age, income, geographic location and other similar factors that segment a population (Hargittai, 2007).

The capabilities that Social Media holds within has led to new and innovative ways to connect. They have become a platform for people to communicate, congregate and collaborate. The following section discusses the various types of social media and their uses based on their respective structure and capabilities.

2.2.2 Types of Social Media

Section 2.2 discussed the definition of the Social Media websites used by users worldwide. These websites are quite popular on a global level. For example, Facebook, Twitter, LinkedIn are popular as the biggest Social Networks. Instagram, YouTube, Snapchat are the popular media sharing sites and Pinterest and Google+ being the next in the list of popularity. However, to group Social Media and Social Networking Sites might be overwhelming and challenging. These platforms are now becoming richer in features, and the line that differentiates them is blurring as they are changing fast in terms of technology and usability. Thus, instead of categorizing these Social Media and Social Networking Sites by their hyper-specific user interests or shifting technology features, it is better to group them into general categories from the capabilities they present to the user. Following are the groups that segregate the said platforms based on what the people want to accomplish with them (Foreman, 2017):

- i. Social networks—Connect with people
- ii. Media sharing networks-Share photos, videos, and other media
- iii. Discussion forums-Share news and ideas
- iv. Bookmarking and content curation networks-Discover, save and share new content
- v. Consumer review networks-Find and review businesses
- vi. Blogging and publishing networks—Publish content online
- vii. Interest-based networks-Share interests and hobbies
- viii. Social shopping networks-Shop online
 - ix. Sharing economy networks-Trade goods and services
 - x. Anonymous social networks—Communicate anonymously

By creating such categories, it becomes easy for brands and marketers to understand why people use them. This might lead to new ideas and channels for engaging with the specific users of these platforms, based in the specialty of the marketer/brand/organisation with greater effectiveness. We further extend this discussion to describe these categories and list the Social Media platforms falling under each category (Takaoka, 2016; Foreman, 2017):

i. Social networks – Social networks comprise of the Social Media or Social Networking online platforms which are used by people to connect with other people with similar interests,

activities, backgrounds or real-life connections (Obar and Wildman, 2015). Users have to create a profile to access the networking services of the website or the mobile application, which is maintained by the social networking website organisation. These users then connect to other profiles and create a virtual network of people who have similar tastes and liking (Ellison, 2007; Obar and Wildman, 2015). Internet-based social networking services are run on the user-generated content, which is posted by users in the form of images, audio or visual media (Kaplan and Haenlein, 2010; Obar and Wildman, 2015).

Examples: Facebook, LinkedIn, Twitter.

ii. Media sharing networks – These are social networks that allow the users to search and share images, videos and other media online. Like the major Social Media networks, these sites can be utilised for lead generation, audience engagement, and most importantly, brand awareness (PcMag, n.d.). Organisations are making use of these networks to create and spread a 'buzz' by posting content/images/videos that would be shared by the users with their peer groups, making the content viral (Affilorama, n.d.).

Examples: Instagram, Snapchat, YouTube.

iii. Discussion forums – Before the emergence of Web 2.0 and Social Media, online forums were the source of information for the people. They were used for discussions, exchange of information, and to connect with like-minded individuals (Blackboard, n.d.). Similar applications have been invented based on social platforms and are now available for people to use for gaining knowledge and sharing the same in a collective group. These websites are the hub of information on current scenarios for users to provide their own opinion or add to the same. Most of the Social Media platforms are working towards creating a safe space online by removing or reducing the user's capability to engage anonymously. However, discussion
forums allow the user to comment and interact anonymously. The anonymity makes the user discuss and share information freely.

Examples: Reddit, Quora, Digg.

iv. Content curation and bookmarking networks – New information is being generated and uploaded on the Internet every second. Moreover, to find a keep track of relevant and useful information has become very important. Content curation through bookmarking helps people to find more about the trending content and media (images, audio, video, etc.), which is being discussed by other users. These bookmarks are usually public and can be viewed by other members of the site where they are stored (Networksolutions, n.d.). The user can collaborate and share their bookmarks to create a larger pool of similar content and help in indexing and curation of content.

Examples: Pinterest, Flipboard.

v. Consumer review networks – It has been documented that 88 % of online shopper seek and consider reviews before making any purchase decision (Abramyk, 2017). The content that makes up the reviews create more value for many online services and websites. These web services build a network around consumer reviews as a part of the value they provide to their users. These networks are sites that review hotel, restaurants, online purchasing, Travelling and so on. Mobile-based applications of such platforms have given them more power as the user can post reviews from anywhere. The social aspect of these platforms allows the reviews to percolate into the relevantly interested masses and generate interest or disinterest for a product or a service.

Examples: Yelp, Zomato, TripAdvisor.

vi. Publishing networks (blogging) – The Late 1990s saw the emergence of the blog which is short for a web-log, where a discussion or information is published in open domain on World

Wide Web (Wortham, 2007). These discussions or information are in a discrete, often informal diary-style text entries called as posts. In the last decade, blogging has become a powerful tool for brands and consumers to publish content online. This particular format lets the user discover, share, and comment on the blog website. A blog can help a brand gain visibility and increase awareness about a business. It can also be instrumental in generating engaging content for the Social Media sites, and in helping brands to establish a niche market for themselves (Foreman, 2017).

Examples: WordPress, Tumblr, Medium.

vii. Social shopping networks - Certain e-commerce websites build their site around a focused integration between the social experience and the shopping experience. Consumers of such websites use linked social networking sites to share their latest purchases, deals, coupons, product reviews, want lists, and other shopping finds (Beal, n.d.). These networks aim to learn more about the consumer through their shopping habits connect like-minded shoppers who can make the purchases and the reviews about the products that they have bought. Social shopping is a personalized form of shopping which solely thrives on the participation of the consumers that like to share information about a product or service to help other shopper/buyers (Nations, 2017).

Examples: Polyvore, Etsy, Fancy

viii. Interest-based networks – There are websites which are focused on interest-based social networking. This is more of a target-based approach where a single subject is focused upon, e.g., books, music, or home design industry (Foreman, 2017). Due to the specific area of interest, these platforms deliver a unique experience for users who have a similar interest. People can share and exchange topic-specific information after getting acquainted with each other, and maybe become friends on the platform (Afetian, 2013; Dayal, 2016). These networks

allow users to come together and collaborate on offline projects after they interact and connect; thus, becoming a platform, which supports collaboration.

Examples: Goodreads, Houzz, Last.fm

ix. 'Sharing economy' networks – These websites are part of the enormous new "sharing economy", where users can share their assets with other users, via the Internet (Platformed, n.d.; Economist, 2013). These platforms are also known as "collaborative economy networks", and they connect people for advertising, trading and selling products and services. Mobile-based networks have made the sharing economy a more useful aspect, and it is forecasted to grow in the near future since it reduces the cost of doing business by removing several channels between the service provider and the consumer (Economist, 2013).

Examples: Airbnb, Uber, TaskRabbit.

x. Anonymous social networks - Majority of the Social Media websites are making an extended effort to make their user hold accountability for their social activities. However, then, there are anonymous social networks that provide a platform for users to be anonymous while using the platform. More popular among teens, these sites are becoming a fun place to blow off steam (Foreman, 2017). People assume virtual identities and aliases which majority of the Social Media websites disagree. However, many mobile-based apps are designed to keep people anonymous and get them to confess their darkest or innermost emotions and thoughts, rather than post thoughtful comments anonymously (Walker, 2016)

Examples: Whisper, Ask.fm, After School.



Figure 2.3: Social Media Landscape Source: (wpmu, 2011)

2.2.3 Use in business

There was a time when the increasing use of Social Media was thought to be a fad and considered more suitable for "kids", which could not be used by businesses as it posed no beneficial use at the time (Copp, 2016). By January 2019, there are over 2.77 billion Social Media users globally. These popular Social Media platforms are becoming marketing channels that collect information about its users and act like a freeway to reach and connect with the user (Kemp, 2017, Statista 2019e). Social Media access through desktop computers provides a variety of opportunities for marketers from the perspective of the business. Mobile-based Social Media on smartphones and tablets allows marketers to utilise the location and time-based awareness of the user. These platforms have potential and capabilities suitable for promotion, marketing research and relationship development (Kaplan, 2012). Marketers have started using Social Media platforms as a tool to monitor and analyse online communication

among the users about the brand or products or related topics of interest (Nowlin, 2013). In 2015, it was reported that the CMOs of organisations have started to pay attention to SNSs. Once these SNSs were called a fad; but that has turned into the most popular way through which brands can connect and reach out to their consumers to increase sales (Samson, 2015).

The following section discusses the various capabilities held by Social Media platforms, which might be beneficial for brands and marketers:

i. Marketing research – Social Media is used worldwide by a significant percentage of the population and therefore, generates a huge amount of data in real-time. Every day there are over 500 million Tweets (Aslam, 2017), 4.5 billion Likes on Facebook (Ho, 2017), and 95 million photos and videos uploaded to Instagram (Parker, 2016). Behind these numbers exists a staggering amount of information about the consumers, which mostly deals with - who they are, what they like, and how they feel about a brand. For conducting a survey, it would take days or maybe weeks to gather enough data for analyses. Due to Social Media, the response can be obtained in an instant through comments and polls that might play an important role in decision making for brands (Samson, 2015).

ii. Valuable consumer insights - One can gather relevant information and data about the consumers to make a smarter decision through daily engagement with them. This can be achieved by "Social Listening"; in other words, monitoring the SNSs to find what all is being discussed by the consumers about the brand, its competition, and other such relevant topics. These insights are crucial for any brand for its business (Newberry, 2017). Certain online-based, as well as standalone tools and applications, are also available which gather real-time information on all the social networks on which a brand or marketer is engaging the consumers. This helps in gauging consumer's perception and attitude towards a brand and generate real-time analytics for further analysis (Copp, 2017). With certain mobile applications, one can access data about the offline movements of consumers. These applications provide information

to the brands and marketers regarding the location and time at which a consumer entered one of its outlets, along with the consumer's engagement about his/her visit to the store (Kaplan, 2012).

iii. Competitive analysis - By carefully monitoring Social Media, one can gain key information about the competition. It is about observing the Social Media accounts of the competition, which also gives insights to understand the industry through the Social Media landscape (Pressault, 2016; Entrepreneur, 2017). The information like the number of posts or messages sent by the competition of the social platform, the number of likes they have got on their profile page/on a post/message, the type and quality of the content they post, their response to the comments and the duration between each post can be helpful in planning a counter strategy for the competition (Jackson, 2017). There are dedicated online and offline applications that are meant for such data analysis. One the basis of the same businesses can upgrade their offerings in terms of products and services, which might be missing (Samson, 2015).

iv. Coupling organic and paid strategies - The capabilities of these platforms allow the marketers to engage with the consumer without any financial spending. However, employing a hybrid paid-and-organic approach has a better and strong synergy attached to them. They can make a huge difference in attracting and engaging the targeted audience (Samson, 2015). A brand needs to be actively involved with the social community comprising of the consumers/audience, listening and learning to what they are saying. It is also important to pay focus on the content that is posted on the social platform as it can play a pivotal role in capturing the attention and attract new people to the social community. If the posts are not able to capture the targeted users on a personal level by the posts, a paid program might be able to create the intended visibility in the said social platform (Gurd, 2016). Incorporating both organic and paid services creates multiple touchpoints for a business to engage with potential consumers. For

example, if a marketer is campaigning through Facebook advertisements, he/she can target consumers based on geographic location, psycho-demographics, their interests, behaviours, and connections (Samson, 2015).

v. Faster content sharing and increase website traffic – A Marketer's biggest challenge is to ensure their content reaches the consumers in the shortest duration of time. Social Media provides a solution to this situation, as content sharing can be done in an organized way on Social Media platforms, along with sorting of the huge data generated from the content (Cisnero, 2014; Copp, 2017). Also, Social Media is capable of Geo-targeting users to reach out to a targeted audience (Copp, 2017). Apart from this, Social Media can be used for effectively increasing the website traffic to the brand's website. It also helps in directing the people to the website but also to increase the ranking as more of the social content is shared among the users and first-time visitors (LePage, 2017)

vi. Sales promotions and discounts – When a brand/marketer is promoting itself or a particular product, they use Social Media for promotional purposes. The promotional discounts which were handed out earlier in the days in the form of printed pamphlets or coupons have been replaced by the digital coupons. These coupons are available on the brands/marketers' Social Media feed, and the followers can easily avail the discounts while making purchases (Payton, 2011; Oliver, 2014). If implemented correctly, the promotions done through coupons can be used for keeping the existing consumers engaged and acquiring new ones. Promotional discounts can be offered as a currency-based coupon (discount-based on the denomination of currency), percentage-based coupon (discount based on a percentage on the price), and a coupon code for store credit (discount based on both Currency and percentage but in the form of store credit) (Oliver, 2014).

vii. Build relationships – Social Media allows marketers to enrich their relationship with their consumers by acting as a two-way channel. Social Media connects with the consumers and

create a dialogue which is very different from the traditional modes of communication that businesses use. Marketers can now interact with consumers and vice versa before, during and after any mutual engagements (Kaplan, 2012; Copp, 2017). Marketers and brands are using Social Media platforms to provide real-time support to the consumers so that consumers do not stop purchasing or associating with the brand due to poor support services (Samson, 2015; Beese, 2016). Social Media provides an economical channel for brands to engage with consumers, which leads to a deeper relationship them. By increasing the longevity of the relationship with consumers, marketers can build a strong, loyal following, where consumers regularly engage with the brands via Social Media platforms to earn discounts and other perks (Kaplan, 2012).

viii. Employee recruitment and advocacy –Organisations can tap the information from the Social Media landscape to select and recruit employees that fall under the interests and skillsets for the required psychographic profile (Parker, 2015; Budzienski, 2017). Passive participants who are currently not looking for jobs but who have the skillset to join the organisation can easily be found through social recruiting. The instant messaging capabilities of all major Social platforms provide a faster way to connect, discuss and recruit the prospective employees (Parker, 2015; Samson, 2015). Large organisations are tapping their employees to multiply their social reach instantly. When the organic reach on the social platform declines, marketers can easily increase the same using the employees to promote their content to their peer network (Parker, 2015; Samson, 2015; Budzienski, 2017).

ix. e-Commerce – Brand and marketers are beginning to use Social Media for increasing the numbers in their audience segment and understand the demographics to deliver more targeted messaging. The Social profile acts as a storefront for quite a lot of the e-commerce websites where there can be an exchange of information, discussion with consumers and handling of queries, new updates about the brand or a product, etc. Users often review a brand or a company

which is beneficial for marketers as they can obtain insights, both positive and negative reviews of the consumers (Dunay, 2013; Zorzini, 2015). The negative reviews are mostly the grievances that the consumers have faced while doing business with marketers. Responding to these negative comments and quickly responding to them reduces the consumer iteration rate. E-Commerce giants like Amazon are using similar strategy to address such inquiries/grievances to create value from the perspective of consumers (Zorzini, 2015).



Figure 2.4: Choosing the Right Social Media Platform for Business Source: (Boatin, 2017)

2.3 Online Consumer Engagement (OCE)

Section 2.2.3 discusses the uses of different Social Media platform that can be used by the businesses for engaging people. Studies related to marketing had discussed the concept of online consumer engagement as an emerging area of interest. The engagement concept has been discussed by quite a few researchers that have put forward definitions to describe the engagement. Engagement has been described as the unidirectional manifestations of the consumer's behaviour towards a brand, which includes the consumer-related online activities such as, review, recommendations, suggestions to other consumers, word-of-mouth, etc. (van Doorn et al., 2010). Another definition suggests and defines engagement as the "consumers' behavioural manifestation toward a brand or firm" (MSI, 2010). Further, engagement has been considered to be a behavioural construct that entices consumers to participate in the brand offerings and other such activities (Vivek et al., 2011).

These studies defined the engagement from a unidirectional point of view. However, now researchers have recognized, that to understand engagement from the perspective of marketing, it is necessary to understand it from a multidimensional perspective. Engagement can be considered as a consumer's cognitive state, which generates a level of attraction or repulsion for the brand (Higgins and Scholer, 2009). By including the effect with cognitive states, engagement can be considered to be behind the formation of consumer's loyalty towards the brand (Bowden, 2009). Studies have also described engagement from a cognitive, emotional, and behavioural perspective, but only a few such studies exist. From this perspective, engagement can be described as the amalgamation of 1) concentration of consumer on a brand, 2) dedication and a sense of belong towards a brand, 3) level of energy needed to interact with a brand, and 4) communication that occurs between the consumer and the brand (Patterson et al., 2006; Hollebeek, 2011). Innovations in Internet technologies have changed the face of the engagement for the marketers. Social Media and other such platforms have become mediated

space where consumers can be engaged and immersed in the brand offerings (Reitz, 2012). Engaging and connecting with the consumers has been made convenient by Social Media platforms.

2.3.1. Definition

Online consumer engagement can be envisioned as the combination of cognitive thoughtfulness and process orientation that are emotionally driven for consumers to participate and engage in active transactions (Reitz, 2012). When consumers respond to the engagements by a brand/marketer through websites, or/and mobile-based applications; they get cognitively immersed into the medium before being engaged. It can be said that experience during interactions with the engaging channel comes before the consumer's attitude and behaviours (Thomson et al., 2005; Mollen and Wilson, 2010). Literature suggests that engagement and involvement seem like the same thing, but they are not. Studies show that engagement goes beyond involvement which is merely readiness to consume the object, decision, or action by the allocation of the cognitive resources (Thomson et al., 2005); But Engagement can be described as actively committing to the brand/marketer both cognitively and effectively via its engagement channels (Mollen and Wilson, 2010). Online engagements are more immersive and they tend to make consumers engaged actively. Therefore, they can be taken as the improvements in engagement techniques as the experiential value of engagement is perceived to be more by the consumers.

A lot of studies have defined engagement from the point of view of connection and bonding with the brand (Wang 2006; Marci, 2006; Rappaport, 2007; Heath, 2007; O'Brien & Toms, 2008, 2010), emotional congruence (Douglas & Hargadon, 2001; Mollen & Wilson, 2010), and pleasure and satisfaction (Mathwick & Rigdon 2004; Fiore et al., 2005). Some of the studies also defined engagement as a complex cognitive process that requires focus, sustained attention, absorption, and thoughtfulness (Guthrie et al., 2004; Marci, 2006; O'Brien & Toms,

2008, 2010; Mollen & Wilson, 2010; O'Brien, 2010), and as an experience that comprises of participating, interacting, and co-creating (Harden & Heyman, 2009; Burns, 2010; Evans & McKee, 2010; Lusch & Vargo, 2010; Solis, 2010; Wang, 2011). From the various themes provided by the academic researchers as well as the industry practitioners, the following definition of online engagement can be considered to be the benchmark as there are selected studies that discuss and define the online engagement holistically:

"Online consumer engagement is the state of being present in a mediated branded space where the consumer is immersed in the brand's offerings meant to deliver purpose and value. Cognitive engagement requires intense levels of focus and concentration in seeking, interpreting, analysing, and summarizing information to a point where consumers may lose themselves in the process and may lose a sense of time and space. Additionally, brand learning (e.g., mission, goals, product offerings, philanthropic efforts, and promotions) occurs while negative responses toward the brand are minimized. Furthermore, online consumer engagement requires affective feelings, which involves emotional bonding and connection with the brand, products, and other users that leads to overall satisfaction. Lastly, consumers must invest themselves within the online vehicle by participating through sharing, conversing, and co-creating with the brand and/or other users (Reitz, 2012)."

2.3.2. Types of Online Consumer Engagement (OCE)

Consumer engagement is not only interacting with the consumer but to actively engage with them. There are four attributes that discuss consumer engagement that can be readily used and implemented for OCE (De Beule, 2013). They can be used to build a strategic OCE plan to increase consumer-brand engagement. The four types of engagements are discussed as follows:

i. Active engagement: As discussed in previous sections, the onset of the Internet revolution, it has made it essential for organisations, to connect and know more about their consumer. The active engagement suggests that brands work to connect and engage consumers on different channels - social, mobile and traditional. Actively engaging with consumers on the major social platform gives better opportunities to receive valuable feedback, which can be used to improve the product /service as per the suggestions. For the same, organisations have to make improvisations to people, processes and technology. A better consumer experience by training and engaging the employees to activate consumers as well as to redesign processes to make them more flexible, accessible and reliable. Acquiring the right technology and using it for encouraging the participants in forums, peer-to-peer support communities will provide better user experiences.

ii. Emotional engagement: Connecting with consumers at an emotional level is important. Dynamic interactions and engagements are not enough to make the consumer an advocate of the brand. It has been noted that consumers that are emotionally engaged with a brand tend to complain less and compliment more, resulting in positive purchase decisions. They are also more likely to defend or suggest a brand on the brand communities. The key to becoming a successful brand is to have emotional associations with owning a product in consumers, and the interactions that happen between consumers and the brand. To emotionally connect, one should try to appeal to consumers on psychological factors that enhance the emotional attachment, such as trust, credibility, and positive brand image. It should garner trust and long-term loyalty.

iii. Rational engagement: Consumers that engage in an online community that is either owned by a brand or by a user, tend to involve purely from the perspective of receiving information regarding a product/service. Consumers gain information about product or services value, quality and details and probably will take part in communities or engage or co-create. The more quality information and knowledge a consumer will have, the stronger their advocacy will be for the brand/service. It is suggested that the brands/marketers should be able to engage the consumer from the point of engaging their intellect apart from entertainment.

iv. Ethical engagement: Engaging the consumer cannot be implied through shallow approaches which might be deemed unethical. In the past ethical engagements were 'optional' but now the scenario is vastly different. Social Media nowadays has billions of users; any unethical behaviour from an organisation cannot be kept hidden. Society has become aware of the ethical principles and standards and is more vocal than their predecessors. Brands/marketers should be focussed on adopting ethical practices while engaging the consumers. While engagement strategies are to be formulated well under the ethical principles, especially the communication and the content of the communication which encompass language, information and other socio-cultural nuances. Apart from this, the organisation has to think from an ethical point of view when it comes to its employees, partners, consumers and community.



Figure 2.5: Types of Online Engagements Source: (De Beule, 2013)

2.3.3 Online Consumer Engagement (OCE) strategies

Technology has made it possible for every person to be connected to web and access information in real-time without delay. Considerably reducing the gap between consumer service and marketing. Earlier the consumer experience used to be about the sales journey. Now it has become more on the lines of getting important information, problems/issues getting solved, getting a response for questions and feeling a connection with the brand. As discussed in the previous sections; consumer engagement, when executed well, can foster brand growth. To provide the consumer with a brilliant end-to-end consumer experience following should be kept in mind (Ference, 2017):

i. Prioritising the consumer experience - Consumer experience allows the brand and marketers to increase satisfaction, loyalty, and advocacy. Organisations/brands should focus more on the different ways through which they can serve consumers. By creating Social Media promotional plan around consumer experience attracts more engagement from the consumers. Creating content with information which helps in building a positive experience for the consumers helps in creating a better brand image as well as garnering more footfall on the Social Media platform. The emphasis should be on the needs of the consumer, and how should the brands fulfill the same. The Social Media platforms enable brands to create a promotional strategy that focuses more on creating a consumer experience. Brands are nowadays obsessing over the consumers and are focused on providing an amazing consumer experience.

ii. Humanising the brand - Social Media platforms provide an environment where brands and consumers can communicate and establish a dialogue; the majority of the brands fail to create an impact on the consumers. They lack at being empathetic and making the consumer feel that the brand understands them. Humanizing the brand suggests, that brands resonate with the consumers, without creating a brand persona which is forced or unrealistic. It makes more sense

to find a personality within the organisation which is passionate about the brand. They should be communicative about the brand and making him/her take the helm as a thought leader to engage the audience/consumers. Another option a brand has is to embrace the vision and the goals and become "the brand" personified. This can be achieved by giving a personal touch to the communication which it does with the consumers. Even the information that is shared with consumers should feel like it is coming from the brand rather than from someone unseen behind the Social Media platform.

iii. Personalising consumer communications – Communication with the consumer is the most important task while using Social Media. Brands have accepted the capabilities of the web and Social Media platforms and have started integrating them in the promotional strategies as a potent and essential channel for communication. It is to be noted that the communication on these platforms can become more personal in terms of brand and consumer. Therefore, brands need to understand the impact of such communication on their businesses as well as on consumers while creating a dialogue. The important thing to remember is to give a personal touch to the dialogue where consumers feel connected. They should see a personified version of the brand with empathy and consideration for consumers. The personalisation of communication means creating an environment where the consumer feels welcomed, and they should not feel like data is being collected from them.

iv. Creating useful content – Content is becoming more and more important as we move ahead in this digital era. Earlier content was used only to educate the consumer about a product or a service, but now, it has become a tool to enhance consumer experience to increment their satisfaction levels. Content that holds high informational, as well as entertainment value, tends to attract more consumers on a Social Media platform when it comes to consumer engagement. The consumer nowadays has become more aware, and they take into consideration the informational, educational as well as the entertainment value of the content. The vibrancy and the quality of the content deeply affect the consumer's engagement with the brand on the web and Social Media platforms. Brands are constantly updating their feeds with posts that have higher content value, i.e., the posts that have more relevant and useful information for the consumer.

v. Listening to the consumer – The web and Social Media makes it easy for consumers to communicate with the brand. They can also communicate their grievances to the brand. This is one of the virtues for which brand are turning up towards the Online Consumer Engagement. The sole purpose of integrating Online Consumer Engagement through Social Media into their promotional strategy is the need to engage and understand the consumer. The brands need to listen to consumers and create an understanding of how they feel about the brand. To find if they have a positive/negative inclination and attitude towards the brand or not. A brand must keep listening to communication. It might hold the key to provide a solution to a consumer's grievance or a chance to provide a better recovery solution for them to become loyal to the brand. The insights from the communication with the consumers can give a more elaborate picture to the brands. It would help them in implementing better solutions to the consumers and provide them with a better experience of consumer support. It would also help in creating a positive brand image for the brand as an organisation that gives attention to its consumers.

2.4 Online Brand Communities (OBCs)

Human communities consist of people with a shared sense of trust and a common value system (Wiegandt, 2009; Martínez-López et al., 2015a). With time, the basic motive evolved, becoming more interest and need centric for every individual, not on geography or family (Wilson, 1990). These communities combined the traditional values of communities with individuals' needs (Von Loewenfeld, 2006). The reach of communities has been expanded by the Information and Communication Technologies (ICT). It has also strengthened the role of

interests and needs of consumers with reduced importance to individuals' geo-location for their integration within a certain community (Uslaner, 2000; Martínez-López et al., 2015b). These virtual communities are linked to a common topic with their own set of rules of conduct within the community, both implicit and explicit (Uribe et al., 2010). These communities promote and encourage the incorporation of user's statements, feelings, thoughts, and actions towards brands, along with offering their shared value to the brand community (Dholakia et al., 2004; Weber, 2007; Tickle et al., 2011).

Internet applications and Social Media have penetrated the lives of the population around the globe, and have created Online Brand Communities (OBCs) on these platforms. These OBCs have evolved in the context of supporting the marketing activities of brands and marketer. This has led OBCs to acquire a strategic position when it comes to consumer relationships and communications. Through various procedures and studies, marketing professionals and academics respectively, have tried to use this mechanism for their success (Lin 2008; Casaló et al. 2013). The literature on OBCs has been focused on two areas. First, studies that shed light on the nature of OBCs and explore the characteristics such as the components of the communities, relationships and their classification and the user behaviour within them. Second, studies exist that have discussed user's engagement and participation in the communities and the consequences that engagement has on brands (Woisetschläger et al. 2008).

The brand community can be defined as a community which is non-geographical and holds a network of set social relations among the users/members of the said community, who admire a common brand (Muñiz and O'Guinn 2001). Other studies have also observed OBCs from different social and management related perspectives (Cova and Pace, 2006; Hajli et al. 2015; Martínez-López et al., 2017). They act as an important platform for the consumers to engage with each other as well as with a brand/marketer (Dholakia et al., 2004; Kane et al., 2009; Brodie et al., 2011b). Now quite a lot of organisations/brands/marketers have started to build

and manage online communities from the perspective of building relationships with consumers. They aim to get their feedback to formulate new strategies to connect with consumers (Wiertz and de Ruyter, 2007).

Section 2.2.3 discussed the use of Social Media for business purposes, and one of these purposes is to serve as a platform for businesses/brands/marketers to create ties and links with their consumers (Thompson and Sinha, 2008). Consumers are known to form their own OBCs on social platforms without involving the brands/marketers. Certain organisations are actively using these OBCs to connect and communicate with their consumers as OBCs are low in cost and developing a community presents a strategic opportunity to the brands for adding value to their existing brand identity (Wirtz et al., 2013). OBCs allows the brand to involve consumers for content co-creation on Social Media platforms; thus, improving their ties with consumers, as well as their competitive position (Tsai et al., 2012). OBCs encourage interactivity among the users within the community, which is beneficial for both brands and consumers. A brand gets to know consumers perception about the brand form the communication and consumer participation within the communities (Wirtz et al., 2013). Marketers now understand the need to connect or create and maintain such communities to increase consumer value (Martínez-López et al. 2017). OBCs have made the consumers an important entity in content co-creation with the brand, which advocates word of communication in the virtual communities; thus, changing their position from being a "passive receiver" (Sashi, 2012; Jahn and Kunz, 2012). Firms nowadays see their consumers as allies and integrating them in new product development, decision making, defending the brand and brand value creation (Hassan and Casaló, 2016; Prahalad and Ramaswamy, 2004). Thus, OBCs can be an excellent platform for marketers for creating real connections and co-creating brand value while engaging consumers (Bagozzi and Dholakia, 2006).

However, there is a gap in research that explores and offer a complete explanation of how brands should maintain and manage the brand community to create higher levels of engagement for consumers through user experience and trust. The research that exists in this respect has not been given due consideration to the effects that OBCs have on the perception of the user towards their purchase behaviour (Porter and Donthu, 2008).

2.4.1 Types of Online Brand Communities (OBCs)

There are different classifications for the OBCs that has been listed in the literature. The most widely used and studied are the OBCs initiated by the user/consumers and the OBCs undertaken by the organisation (Kang, 2004; Jang et al., 2008). OBCs have become quite active in brand discussions, product/ service information and advocating brands and their product/services. Such communities are made by consumers that like to share and collectively create a hub or pool of informative data that would be beneficial for other consumers as well. These OBCs might be in the form of wikis, blogs, SNS fan/unofficial page, etc, and they form the first type of the OBCs. However, innovative organisations that have understood the potential of such OBCs build their own to strike communicative and positive relationships with their consumers.



Figure 2.6: Types of OBCs based on initiation



Figure 2.7: Types of OBCs based on the orientation

This classification of OBC is depicted in Figure 2.6 with further bifurcation into social and professional OBCs under User-initiated OBCs and commercial, non-profit and Government OBCs under Organisation-initiated OBCs (Markus, 2002; cti09, n. d.). OBCs can also be bifurcated depending upon the orientation they have, and the scope they wish to fulfill. Communities that have social, professional and commercial orientation can be further bifurcated into communities that have a concentration on learning, entertainment, relationship building and are focussed on business-to-business and business-to-consumer (Markus, 2002). These subdivisions present a better understanding and segregate the type of OBC a brand/marketer is trying to follow/create based on the needs they have. Figure 2.7 depicts the same.

2.4.2 Consumer Engagement (CE) in Online Brand Communities (OBCs)

The innovations in the Internet technologies and advent of new interactive digital platforms have led to consumer engagement becoming an important aspect of the virtual environment, in creating lasting relationships (Sashi, 2012). Therefore, consumer engagement has become essential in web-based applications such as wikis, blogs, SNSs, etc. (Hollebeek, 2013). Previous studies have tried to understand and explore consumer engagement in OBCs (Algesheimer et al., 2005; Mollen & Wilson, 2010; Van Doorn et al., 2010; Brodie et al., 2011; Gummerus et al., 2012; Wirtz et al., 2013;). These studies defined consumer engagement and were the primary literature which addressed and explored the consumer engagement in an online brand community setup. Keeping communities into context, few studies have attempted to present the concept of consumer engagement in virtual brand communities (Brodie et al., 2013). One of the definitions suggests that consumer engagement refers to the consumer's participation within an OBC, because they can identify the OBC with themselves. This definition explains the engagement behaviour at a basic level. It can be said that it is the intrinsic motivation that a consumer has, to interact and cooperate with community members (Algesheimer et al., 2005). OBCs are based on the member user's interest to help other users and actively participating and engaging within the community. These activities increase the value members have for themselves as well as for the other members of the OBC (Wirtz et al., 2013).

Previous definitions were focused on the user or consumer, engaged with the brand community. However, engagement within a community is more from the perspective of the interaction a consumer experience. The emotional and behavioural factors that follow the engagement might have consequences based on previous long-term engagements with the community (Brodie et al., 2013). Similarly, engagement can be phrased as a commitment a consumer has, due to the close relationship he/she has with a brand, which is represented by a website or OBC where the consumer actively communicates their brand value (Mollen & Wilson, 2010). This perspective suggests that CE involves both emotional and cognitive processing of the relation that a consumer holds with a brand.

Use of technology to engage consumers and to understand their inclinations is one of the critical factors in CE in OBC. (Sashi, 2012). These engagements through the use of internet technologies can be essential in increasing the brand's value for consumers through the

interactive and innovative experience they might have with the brand (Gummerus et al., 2012). The above-mentioned interactive behaviour leads to communications between consumers on a digital platform; such as forums, brand communities on Social Media websites, blogs, etc. Thus, providing the brand with information to increase their brand value, while creating the image of a brand which listens to its consumers.

Therefore, as per the literature, it can be established that CE in OBC has the following qualities:

- i. CE comprises of active, interactive experiences between consumer-brand or between consumer and other users within the OBC.
- ii. CE deals with a psychological state depending on the context, which means the intensity of engagement by a consumer will be highly affected by the dynamism and instructiveness of the CE within the OBC.
- iii. CE is multidimensional, consisting of encompasses cognitive, emotional and behavioural dimensions.
- iv. CE is an important factor in the process of exchange-based relationships, and can be a part of other community-engagement concepts; both as an antecedent or a consequence.

2.4.3 Drivers for Consumer Engagement (CE)

Consumers find it easy to join an OBC instead of directly engaging with a brand or others who engage with the same brand. Engagement with OBC happens only when the consumer is driven by the possible benefits of both hedonic as well as utilitarian. Consumers might join an OBC, but that does not signify that he/she will be engaging actively within the OBC. The need to engage within the OBC requires specific drivers for consumers to participate and become a patron actively. Following are the underlying drivers that prompt a consumer to engage with an OBC (Wirtz et al., 2013). Table 2.1 presents the various drivers of mapped with literature.

S. No.	Driver	Sub-drivers	Remarks	Source
1	Brand- related Drivers	Brand identification	It is an important antecedent to the community participation, which integrates an individual perceived brand identity into self-identity.	Algesheimer et al., 2005; Hollebeek, 2011a; Wirtz et al., 2013
		Symbolic function of a brand	Consumer's passionate emotions for a brand due to strong ties can make a brand community a perfect channel tin creating a strong brand image. These communities provide a platform for consumers to engage in discussion, therefore supporting higher consumer engagement.	Fournier, 1998; Albert et al., 2012; Wirtz et al., 2013
2	Social Drivers -	Social benefits	Online brand communities have a lot of social benefits to offer, which leads to greater interactions within a community and greater identification between community members, improving community engagement.	Wang et al., 2002, Hennig- Thurau et al., 2004; Van Doorn et al., 2010; Brodie et al., 2013; Wirtz et al., 2013
		Social identity	The idea of belonging to a group or being part of a group leads to social identity. The user engages on social platforms and follows or join a community, with which they fee associated. Participants of such communities are Strong associations with a brand among members of a community leads to members reaffirming their social identity and, therefore, their consumer engagement.	Hogg & Abrams, 1988; Hogg & Abrams, 1988; Kim et al., 2011; Wirtz et al., 2013
	- Functional - Drivers	Functional benefits.	Functional benefits (transactional, informational, support, help, entertainment, value, etc.) are one of the needs that make a person participate in an online community. If a participant receives enough of these benefits from other members, they will continue to engage with the community.	Wang et al., 2002; Wirtz et al., 2013
3		Uncertainty avoidance	Online brand communities reduce the level of uncertainties for a member when they are about to make a purchase decision. Their communication and exchange of information with other members leads to reduced levels of uncertainty	Wirtz et al., 2013; Zaglia, 2013
		Information quality	People participate and engage in online communities mainly to have access to quality information. Therefore, a clear and extensive informational repository might help the users within a community and will influence their engagements with the online brand community.	Algesheimer et al., 2005; Dholakia et al., 2009
		Monetary incentives	Brands/marketers tend to use incentive (discounts, sale information, etc.) to promote participation in the community, which might encourage consumers to engage with the online brand community.	Fu ^{°°} ller, 2006; Wirtz et al., 2013

Table 2.1: Drivers of CE

Source: Adapted from (Wirtz, 2013).

2.4.4 Process of Consumer Engagement (CE) in Online Brand Communities (OBCs)

Literature suggests how organisations should manage consumer engagement; which is quite relevant for OBCs. Previous studies have given one of the initial and fitting processes. They proposed that consumer engagement can be encouraged in three stages (Van Doorn et al. 2010):

i. Identification of consumer engagement behaviours – The brand has to identify how a consumer wishes to engage with the brand. Finding a platform which fulfills all the needs to connect and engage the consumer. The content and the type of information that the consumer likes to consume, and that would be able to attract more traction on the OBC should also be looked at carefully. The effects of such engagement on similar platforms need so to be observed and a strategy according to brands should be formed and executed.

ii. Evaluation of the onset of engagement - Before creating an OBC, it is a must to vision the long term and short-term consequences of the OBC. The brand should demark the short-term goals of engaging the consumer (use of season promotion, sale and discount, etc.) and the long-term goals (promptly listening and solving consumer grievances, emotionally engaging consumers that are loyal to the brand, developing a more profound relationship with the consumer). Based on these evaluations, the brand should be able to exploit more from the OBC and create a community which functions as envisioned by the brand.

iii. Action about consumer engagement – Once a brand realizes the above two stages of OBC engagement, they should look for the best suitable resources to develop an OBC. These resources can be a Social Media platform that suits the brand needs, a team of individuals who understands the brand's vision for OBC engagement and create and develop content for the OBC, suitable tools and application that measure the engagement within the OBC statistically. Studies have further suggested how the process of engagement within OBC works. They have conceptualized the process of consumer engagement in the context of virtual brand

communities into five processes (Brodie et al. 2013). Content sharing, learning, co-creating and socializing; these can be considered as the stepping stones to create an engagement process in any OBC. Figure 2.8 depicts and discusses this five-step process.



Figure 2.8: Process of Consumer Engagement

(Adapted from Brodie et al., 2013)

2.5 Facebook Brand Pages (FBPs) and Consumer Engagement (CE)

As discussed in Section 2.2.3, brands have started to use Social Media as a useful marketing channel in their promotional planning to engage the consumer and to create awareness (Muniz & O'Guinn, 2001; Kaplan & Haenlein, 2010; Malhotra et al., 2013; Rohm et al., 2013). Facebook has emerged as an important platform for marketing due to the capability of the platform in establishing direct communication with consumers. Due to its popularity, several companies have started integrating SNSs as marketing channels in their strategic planning

(Kabadayi and Price, 2014). Majority of the brand have Facebook Brand Pages that consumer connects with or "like" to engage with the brand via Social Media. Brands have understood the potentials of Social Media, and they are trying to integrate and accommodate them not only for marketing purposes but to connect, reconnect and create consumer-brand relationships as well.

Facebook allows the user to create a Facebook Brand Page (FBP) which, in other words, is an OBC. Facebook Pages and Facebook groups are the most used formats of the brand communities. Users who wish to join any such community press the "like" button on the brand page, after which the user will start receiving posts from the brand page. There are four types of posts that a brand can choose to use: video, photo, weblinks and status (Olczak & Sobczyk, 2013). Facebook enables a brand to post media on the brand page and receive thousands of comments from consumers that are Facebook users. Thus, making Facebook a platform which provides an environment for consumer-brand dialogues and to understand the sentiments of consumers (Malhotra et al., 2013). With each post, FBP enables the brand with three interactions from the consumer: like share and comment. If a user likes a post, it reflects on the feeds of other users who have liked the brand page apart from the friends in the users "friend list". If the post gains more likes, its value in the context of page marketing value gets higher. If a user shares the post, it is published in their profile or page, which can further be viewed by users not connected with the brand page and might be shared further, thus making the content viral. The user can also share their thoughts by commenting on the brand Posts, which is the most used feature of Facebook (Olczak & Sobczyk, 2013). Though this feature might be considered to be risky, in case a disgruntled or dissatisfied consumer decides to vent out at the company by posting criticising and derogatory comments about the product or brand (Jayasingh and Venkatesh, 2015)

Consumers who join such FBPs, do so to convey sentiments. Both negative and positive, that they feel towards a brand (Wilimzig, 2011). Apart from these reasons, users who are loyal to

the brand choose to join such communities. Consumers might be motivated by the economic benefits such as incentive, discounts and sale that is being provided by the brand on the brand page. Brand, at times, gives exclusive discounts or sale information as an incentive to the consumers that are on the brand page (Vorvoreanu, 2009). Majority of the studies suggest that users join such brand communities to avail the discounts, coupon or the rewards posted by the brand, then making friends or socializing on the brand community (Weman, 2011; Syncapse, 2013). Consumers engage in the brand community by reading the posts updated by the brand, reading user comments, posting comments on the brand's wall or other activities. Studies have shown that more a consumer engages in an online brand community, more it helps the brand in terms of positive outcomes (Burns, 2010).

There has been extant literature that explores consumer engagement from the context and point of view of the Facebook brand community. The empirical research has been focused on the three aspects of posting on Facebook, i.e., Liking sharing and commenting. A holistic approach that identifies and uses generic factors for marketers to understand and focus on is scarce in the literature.

2.5.1 Facebook Brand Page (FBP) engagement - benefits to brand/marketers

Brands and marketers nowadays have become active on Facebook and have started to make Facebook pages which are run and moderated by the Brands/Marketers themselves. As discussed in the previous section the FBPs are quite useful in keeping the consumers engaged and thus making it possible for the brands/marketers to connect and communicate with the consumers from the point of information and promotion. Following are the major benefits that are associated with the FBP engagement (Rhea, 2018):

i. Lowers the marketing expenses – There is no cost for making and starting a Facebook page. The only requirement is that of good graphical and literary content to run any such FBP. Other media needs the brand/marketers to create strategies both from the perspective of marketing as well as finances. FBPs, on the other hand, are free to make and moderate. Brands have converged onto FBPS and similar platforms and have designed their creative marketing strategies. Such platforms enable brands to connect and communicate with their consumers. Even though the costs for running an FBP is zero, the cost to run advertisements, generating likes from users, boosting posts, and running sponsored stories are relatively inexpensive from the context of traditional media. Even though it is less costly, the reach is far greater. The Facebook ads can be used to target a specific audience; thus, it is a great help in targeting the desired consumer base.

ii. Reaching to the targeted audience – The Ad service run by Facebook is very efficient in finding the targeted population on Facebook. Facebook lets the brand to select the audience towards which the FBP is targeted. The selection of demographical variables, along with sociopsychographic variables helps the brands in reaching the specific audience. The targeted audience can see the suggested FBP based on their profile data and the majority of them either engage or follow the brand run FBP, thus making the Facebook advertisement useful. Unlike traditional media, where reach was limited to a geographic region, FBPs have a wider reach which can be expanded to a global level. If the advertisement is targeted correctly, one can easily have the right people to connect with the FBP.

iii. Facebook insights - Facebook provides a dashboard by the name of Facebook Insights, which are provided to Facebook business pages. These insights help the brands in understanding the data generated by the engagements on the FBP by the consumers. The analysis of the data is presented in such a manner that a person from a non-technical background can also understand the statistics shown on the dashboard. These insights majorly deal with the number of likes on the FBP, reach of the post and the FBP, engagements on the FBP, post-performance and many other such important statistics. Brands have the choice to

find a specific post and see how it is performing in the aspects mentioned above. These insights can be downloaded for further analysis and to create better content and post strategies. This is the most important aspect of the FBPs, which lets the moderators of the brand page aware of the measurable effects of the posts and the brand page concerning the engagements of the consumers.

iv. Brand loyalty-building – Brand pages on Facebook can help brands in building loyal brand consumers. Apart from being a place to sell the product, FBPs can be used to post entertaining and vivid content for the consumers to engage with. The content strategy followed by the brand can make the consumers stay loyal to the brand even if it makes a mistake. It presents a positive image of the brand if it is active and responsive on Social Media platforms. Majority of the consumers respond to the personalised communication on the FBPs by the brands and keep continuing their engagements, both online and offline. The content, in this case, should be selected by the brand that showcases the brand's culture and approaches towards the consumers. The content is the first thing to which the consumer reacts. By using simple communication through the content, keeping the existing consumer base, and adding the new consumer to it can be easily done with appropriate digital media strategy.

v. Directing the web traffic - Online engagements on the FBPs by the brand are not the only aspect for which FBPs are used. They can direct the user traffic to the official website of the brand to know more about any product or get more information about the brand. Engaging the consumer with the posts will make any brand an entertainer, but to be a marketer, one needs to direct the consumer to its product on their official websites. Providing backlinks to the official website is a strict necessity for brands to drive the traffic to the brand's official site. Facebook generates a full-width thumbnail image to the link provided on the FBP by the brand. The embedded link draws more attention and makes is more likely to be clicked. Therefore, making the website available to the consumers accessing the FBP. Facebook enables the brands and

marketers to channel the web traffic to their websites and thus increase consumer engagement as well as the user footfall on the official website, thus increasing its ranking.

vi. Accessibility through mobile devices – Facebook is more used as a mobile app than other digital devices. Thus, it has become easier for brands to have a presence on Facebook through FBPs and connect directly with the consumer. The mobility of the platform makes it more efficient in understanding the consumers as well as being accessible to them at all times. The presence of FBPs provides the brand with a clearer understanding of consumer information. Due to the mobility feature, people can engage easily on the FBPs. Majority of the brands hold competitions for the consumers where they are asked to share or comment on a post. They are also asked to post their pictures with certain captions or hashtags. The app makes it possible for them to do so. Also, it helps in generating data which can be accessed to understand consumer sentiments further. The accessibility of Facebook on the mobile app gives an added advantage to engage the consumer without any limitation. People can use the app on their mobile phones, which increases their engagements with any FBP. Brands are focussing on this capability and designing their marketing strategies to engage the consumers on the app.

2.6 Research gaps

The research gaps identified through a review of the literature are as follows:

- Literature has discussed the concept of Online Consumer Engagement on Online Brand Communities from a theoretical point of view. However, the literature on the constructs that are important to Online Brand Communities is inadequate.
- ii. The literature discusses consumer engagement in a Social Media environment and otherwise, but there seems to be a gap concerning the identification of a functional relation between the Online Consumer Engagement and consumer's purchase intention.

- iii. Literature that focuses on Facebook has mostly tested relations between the functional aspects of Facebook on consumer engagement (number of likes, number of comments, number of times shared, number of views, etc.). There is inadequate literature on the driver-based approach to identify and understand the antecedents of Consumer engagement in FBPs and its effect on Purchase Intention.
- iv. There are not many studies that discuss the Online Consumer Engagement in the context of the Indian FMCG sector, which contribute theoretically or empirically to two points discussed above.

2.7 Conclusion

The Chapter has presented a review of the literature in four sections, encompassing Social Media, Online Consumer Engagement, Online Brands communities and Facebook Brand pages in a systematic order. The first section discussed Social Media, its definitions, major Social Media platforms along with their uses for brands and marketers. The second section discussed Online Consumer Engagement from the contextual perspective of brands. This section explored the definition of the same along with various types of Online Consumer Engagements. The Chapter also discussed the strategies for brands and marketers to keep in mind while planning to engage consumers online. The third section discussed the concept of Online Brand Communities, various types of Online Brand Communities. It further explores consumer engagement in Online Brand Communities along with the process for the same. This section also identified the major drivers that drive the consumers to engage in an Online Brand Community. The fourth Section discussed Facebook Brand Pages and the Consumer Engagement that occur on the same along with their benefits to brands and marketers. This chapter ended with the identification of research gaps from the review of the literature. The next chapter presents the Online Consumer Engagement in the context of Indian FMCG Brands.

Chapter - 3

Indian FMCG Sector & Online Consumer Engagement

3.1. Introduction

This chapter attempts to fulfill the first and second objectives of the study (Chapter 1 Section 1.6) in the context of major FMCG brands in India and their Online Consumer Engagement practices by the major. The chapter presents an overview of the Indian FMCG sector and discusses the various strategies followed by the major FMCG brands in the Indian market. This chapter aims to understand and present the online and digital engagements executed by FMCG brands to engage consumers. It starts with an overview of the Indian FMCG Sector and its current scenarios from the perspective of market share, growth and the effects of government regulations. It further explores the major online engagement strategies employed by the major FMCG brands on various Social Media platforms.

3.2 Indian FMCG Sector – an overview

Indian Fast Moving Consumer Goods (FMCG) sector is one of the largest sectors in the country which holds the fourth position in the Indian economy. It accounts for 50 percent of the total FMCG sales in India (ibef, 2018a). The Indian FMCG market is estimated at INR 18,500 crores (US\$ 185 billion) or about. It has grown annually at about 12% per annum over the last decade (equitymaster, 2018). The main drivers for the growth of the FMCG sector can be attributed to consumer's increasing awareness, easier access and changing lifestyles. In terms of revenue, the urban segment of the population is the largest contributor to the Indian FMCG sector. Urban sector contributes 40% of total revenue generated, i.e. around INR 2,940 (US\$ 29.4 billion) in 2016-17 (ibef, 2018a).

The pace of growth for the Indian FMCG market has been faster in rural areas compared to urban areas. There has been a record growth of 50% in the FMCG product segment from the spending done by the rural segment (equitymster, 2018; ibef, 2018b). It is estimated that by 2020, the retail market would reach one lakh crore (US\$ 1.1 trillion) from 67,200 crores (US\$ 672 billion) in 2016. This rapid increase is a result of modern trade, which is expected to grow at a rate of 20– 25 percent per annum, boosting the revenues for the FMCG organisations. If there will be reforms in regulations to create a conducive environment for the direct selling market in India; it is estimated to peak to INR 250 crores (US\$ 2.5 billion) by 2021 (ibef, 2018b). Table 3.1 shows the major players in the FMCG Market along with their yearly turnover. Also, Govt is supporting FMCG Sector by allowing 100 percent Foreign Direct Investment (FDI) in food processing and single-brand retail, along with multi-brand retail where the investment is 51 % (Jain, 2018).

FMCG Company	Turnover* (in 100 Crores)	Location
Hindustan Unilever Limited (HUL)	345.25	Mumbai, Maharashtra
Colgate-Palmolive	43.28	Gurugram, Haryana
ITC Limited	406.27	Kolkata, WB
Nestle	112.92	Vevey, Switzerland
Britannia Industries Limited	93.04	Kolkata, WB
Marico Limited	51.70	Mumbai, Maharashtra
Procter & Gamble	24.55	Mumbai, Maharashtra
Godrej Group	52.61	Mumbai, Maharashtra
Amul	292.20	Anand, Gujrat
Dabur	56.09	Delhi

 Table 3.1: Major FMCG companies in 2018

Source: (Lakhi, 2018; thedailyrecords, 2019; economictimes, 2019a,2019b, 2019c, 2019d, 2019e, 2019f, 2019g, 2019h, 2019i, 2019j)

The government has also taken certain initiatives for the purposive growth of the FMCG sector

in the coming years, which are as follows:

- i. In the Union Budget 2017-18, the Government of India has proposed to spend more on the rural side. The aim is to double the farmer's income in five years; as well as the cut in income tax rate targeting mainly the small taxpayers, focus on affordable housing and infrastructure development will provide multiple growth drivers for the consumer market industry.
- ii. IT is also proposed to allow Foreign Direct Investment (FDI) up to 10 % in online retail about goods and services. Clarity has been provided from the existing e-commerce companies in India.
- iii. The government plans to develop a skilled workforce due to the growing demand among the Industries. The plan is to train 5,00,00 people by the year 2022, with investment from private players and entrepreneurs. The training and education to be imparted to the create the said workforce will be provided by governments, corporate and educational organisations.
- iv. The Goods and Services Tax (GST) has proven to give benefits to the FMCG industry.Quotes several FMCG products such as hair oil, soaps and toothpaste have come under 18% tax bracket, which is lower than the earlier rate, which was nearly 24 %.

There are still avenues where the FMCG sector can make its marks. Potential markets in terms of market share and revenue generation are available which have not been tapped to their full potential. Rural areas of the country are one such domain. This Sector has seen an increase in income as well as aspirations, which have led to the demand generation for branded products in rural areas. Thus, there is an increase in rural consumption of the products. The rural FMCG market in India is expected to grow at a rate of 14.6 percent, and reach 22,000 crores (US\$ 220 billion) by 2025 from 294 lakhs (US\$ 29.4 billion) in 2016 (FinancialExpress, 2018a). The organised sector has seen a rise in previous years, and it is expected that with an increase in the consumer's brand consciousness, there will be further growth in the future. There are other

major factors which affect the growth of the FMCG sector. Growing youth population, primarily in urban areas; easy access to Internet technologies, Introduction of Goods and Services Tax (GST) has provided will lead to mergers and rise of world-class consumer companies in India.

3.3 FMCG brands and Online Consumer Engagement on Social Media platforms

Digital media has changed the marketing landscape, especially in the FMCG sector. Consumers are now having access to information on their mobile devices like smartphones, tablets and laptops. A growing number of consumers prefer to shop or purchase online rather than going to physical stores. Consumers are also making buying decisions by going to the digital space and collecting information through various digital portals and platform. Digital marketing for FMCG Companies means transforming the already existing marketing mix for FMCG Products to bring them in sync with the new digital age. Digital media trends allow the brands/marketers to establish a relationship with the consumers alongside capturing the consumer at the final buying stage.

Social Media provides a better front for FMCG brands to showcase their products. It makes it possible for brands to present a product to the consumers in the same way any traditional marketing channel would do, except extended reach. This reach can be evaluated as well due to the engagements by consumers on the posts/uploads of content by the brand. Majority of the social platforms provide the analytics of an official page owned by a brand. They provide brands with analytics about reach, engagements, no of users that have viewed a post and relevant demographical data. Analytics provided by the Social Media platforms provide a concrete measure of the engagements, which is not possible in traditional media. With growing mobile use and increasing Internet users, tapping the consumers have become easier as brands can connect to them on a more personal environment. People use Social Media platform on a
smartphone, which makes them highly accessible to brands to reach and connect. Due to the reasons discussed above, brands and marketers are integrating these platforms in their marketing strategies.

Digital media offers several options for brands/marketers to promote themselves. These platforms help them to meet the varied objectives about promotion, creating awareness and developing good consumer relationships. FMCG organisations are including innovative digital media strategies to blend these channels with a balanced approach effectively. FMCG brands are spreading their promotional activities on multiple platforms to make the consumer aware and influence their purchase decision. Table 3.2 shows the digital presence of the major FMCG brands in India on popular Social Media platforms.

FMCG Company -	Followers on Social Media Platform*					
Theo company	Facebook ^a	YouTube ^b	Twitter ^c	Instagram ^d		
Hindustan Unilever Limited (HUL)	5,08,033	2,88,371	89.808	4,058		
Colgate-Palmolive	28,84,965	4,80,947	31,000	73		
ITC Limited	10,669	-	5,478	-		
Nestle	1,12,03,215	93,675	2,37,054	4,400		
Parle Agro	1,17,267	1,90,755	1,133	5,739		
Britannia Industries Limited	17,773	12,814	3,833	253		
Marico Limited	77,801	1,218	96	1,763		
Procter & Gamble	4,53,170	73,278	245	-		
Godrej Group	7,68,730	5,495	7,390	2,018		
Amul	15,18,596	1,99,598	1,59,890	1,27,146		
Dabur	17,182	-	5,516	-		

Table 3.2: Major FMCG companies and their Digital Presence and Followers

*As on January 2019; a,b,c,d – Data based on the official brand pages on the Social Media platform.

Table 3.2 depicts the Social Media presence of the major FMCG brands in India. As evident from the table, the majority of the brands have a presence on the popular Social Media. While

most of the brands are active on Social Media platforms, a few are not actively promoting or marketing themselves on these platforms. For most of the brands, Facebook is the most followed platform where consumers engage most with them, followed by YouTube. From the Top FMCG brands Nestle, Colgate-Palmolive and Amul are the most followed by consumers respectively in that order. They are also active on YouTube and Twitter with a decent number of followers. However, a few of these brands are lagging compared to their competitors in terms of their Social Media engagements. ITC Limited, Marico Limited and Britannia Industries Limited are some big names in Indian FMCG market, but as can be seen for Table 3.2, it is evident that they are not using Social Media platforms to their full capabilities. ITC Limited has a low number of followers on Facebook, which is the lowest among the FMCG brands. Also, ITC Limited does not have a YouTube account or an Instagram account. Britannia Industries Limited, though present on all the popular platforms, does not have many followers suggesting a low engagement with consumers. Hindustan Unilever Limited (HUL), Parle Agro, Marico Limited, Procter & Gamble and Godrej Group have a good number of followers on Facebook. However, their presence on other brands does not have a similar following. This might be because they have a strong presence on Facebook (Table 3.2) through their engagements on the Social Media platform. It can be concluded that Facebook is the most preferred choice for the brand for making any Social Media based OBC on any SNS. In the next section, a few of the major strategies that brands should follow are discussed. These strategies could help evolving marketers and existing brands in managing a brand page on any such social platform.

3.4 Digital strategies followed by Indian FMCG brands

FMCG brands are striving to create a perfect marketing mix which combines both the traditional and the digital media. This requires a deep understanding of consumers and their behaviours. The FMCG brands need to create a holistic digital media strategy that focuses on

influencing the consumer with engagements done on a personal level. Brand might be in advantageous position after they integrated digital marketing strategy in their overall marketing mix such as - (1) It helps to distinguish the brand and create a unique brand identity; (2) It is an innovative way to showcase the products; (3) It allows brand to gain consumer insights and increase ROI; (4) mobile-based digital platform helps in reaching more consumers (Rhea, 2018). The digital media platforms offer several channels to FMCG brands to use for digital promotion. The reach and the capability of these platforms allow the FMCG brands to create brand image and awareness among the consumers. Some of the primary strategies that are followed by the Indian FMCG brands/marketers are discussed below.

1. Using Social Media capabilities to have a higher reach to consumer - Social Media has emerged as one of the biggest channels through which one can easily reach out to the consumers. Also, the majority of these Social Media platforms, such as Facebook, YouTube, etc., provide inbuild advanced analytical tools to understand more about the consumers from the analytics generated. Such SNS platforms can help in widening the reach and the target audience. For example, the Facebook Custom Audience Tool (accessible to moderators of any brand page) can help in identifying the target audience through the campaigns run by a brand on the platform. It can easily allow brands to target the users based on the information shared on these platforms by them. Brands can get deeper insights about the consumer and their behaviours through various engagement elements such as likes, share and comments. However, leaning onto only one platform is not wise. Therefore, brands generally connect their digital marketing platforms for using most of the digital ecosystem, leading to a better reach to the masses as well as higher engagements with the consumers.

2. Focus on content-based marketing to build a strong community - As discussed before, there has been a shift in how FMCG brands market their products. It has become more evident with the brands are shifting focus from one-way promotion to create user-generated content

that might bring value to the consumers. Consumers will have strong ties with a virtual community on a Social Media platform only if they see brand sharing or creating informative and intelligent content, which attracts them to further engage with the community. Brands strive to build such content-based communities within various Social Media platforms, in the form of profile or brand pages to create a stronger consumer base. Brands are shifting towards user/consumer-based marketing, which is unique to the digital marketing strategy. This helps the brands to promote their brand more effectively by creating loyal consumers which further generate engagement elements (likes, share and comments) to spread the value, associated with the respective promotion by the brand. For example, Amul has a Facebook brands page where they keep posting content, which is quite liked by the users. Figure 3.1 and 3.2 depicts the said posts where the content and the subject matter of the posts were relevant and struck a chord with the users that follow Amul's Facebook Brand Page.



Figure 3.1: Amul's Take on the 2020 US Presidential Candidates

The images that are posted by Amul are always about a current scenario. Also, videos posted by Amul are related to a current topic from the perspective of special festive days such as Valentine's Day, Women's Day, etc. these posts have a good number of engagements from the consumer which is instrumental in spreading the word-of-mouth about the FBP and connect more users to the brand page.



Figure 3.2: Amul's Take on Amitabh Bachchan's 50 Year in Indian Cinema

3. Identifying and using influencers on respective Social Media platform - Influencer marketing has taken up quite a pace in recent years where people that are active in their respective niche Social Media platform, e.g., YouTube vloggers, Twitter personalities which have a large number of following. These influencers are followed due to their specialties. They can be experts in a particular area, which is of interest to the people that follow them. They can also be celebrities that have a hold on the users; thus, making them an influencer. Through the use of influencers in promoting or marketing themselves or their product, brands can tap their huge fan/follower base to increase consumers awareness. In Case of FMCG brands, consumers are more likely to trust and believe an influencer as they might feel them to be a credible source

of information. Roping in influencer can help in increasing the reach and engagement with the probable and existing consumer's presence of the Social Media platform where the brand is active. In 2017 CavinKare, an Indian FMCG brand introduced its Milkshake in the market. To promote the same, they used influencer marketing by hiring food influencers who would write two blog post of 600 to 700 words across various Social Media channels like Facebook, Twitter, Instagram. They were also asked to post one video on the same platforms talking about Cavin's Milkshake. CavinKare planned the strategy from the perspective that through influencer marketing, they might be able to tap the followers of the food influencers to promote their new product, i.e., Cavin's Milkshake. CavinKare targeted the metropolitan cities in India like Mumbai, Delhi, Bangalore, Chennai, Hyderabad and Kolkata to introduced their product.



Figure 3.3: Blog post written by a food influencer to promote Cavin's Milkshake

4. Constantly engaging with the consumer to create brand loyalty – Social Media has changed the face of promotion and advertising. Now brands can engage with consumers at any time they like, and the reach has become global due to the emergence of Social Media platforms. It has replaced the traditional media and has become a go-to channel for digital promotions. FMCG brands use these platforms to create a constant presence in the consumer's mind. They promote themselves to be relevant to the consumers, which can be done on Social Media platforms. These platforms have an edge over the traditional media when it comes to user engagement. Any FMCG brand can engage consumers by posting shareable content, information which is relevant and meaningful to the consumer. Since the advertising and promotional strategies by FMCG are more from the perspective of creating awareness and reminding the consumer about the brand's products, these platforms serve as an important mode of connecting with their consumers. As discussed previously, content plays an important role in engaging consumers on Social Media. Therefore, it is also essential for connecting with the consumer to evoke a feeling of loyalty towards the brand.

5. Converting the engagements into a conversation - as for any FMCG brand, the key metrics are its brand presence, which can be assessed by the analytics provided by the Social Media platform for the brand page. This is done mostly by measuring the engagement elements (likes, share and comments) which play an important role to understand the sentiment as well the attitude of the consumers towards the brand through the brand posts. The comments, however, provide the brand with a potential way to converse and connect with consumers through the Social Media platform. Brands need to convert the engagements by the consumer into conversations to create a positive brand image as well as create loyalty among those who follow the brand page and for regular engagements. For a successful Social Media strategy to work, establishing communication is always imperative, which can easily be done via Social Media platforms.

Communicating with consumers has become easier because of the diffusion of smartphone technology among the masses. Now brands have more reach to consumers, making it feasible to establish two-way communication with them. However, FMCG brands are not quite active in reciprocating to the engagements of the consumer. Being an active brand on Social Media and being a communicative brand on Social Media are two different things and both have different levels of effect on the consumer. Even though big names like Nestle and Colgate-Palmolive have a strong Social Media presence, they rarely respond to the consumer's engagements, i.e., comments on the brand posts. The reason for this might be the fact that the direct messaging capabilities of all the major platforms might help the brands to communicate securely with any consumer. For example, Facebook has the chat application inbuilt in the profile age, which pops up whenever anyone visits the page. Figure 3.4 depicts the said chat application with a few of the most anticipated questions already put up for the user to select to start a conversation.



Figure 3.4: Facebook Chat application for users to connect with Colgate India FBP

6. Using the Social Media platform to sell products - With the emergence and popularity of e-commerce websites, it has become possible for FMCG brands to sell their products along with in-store selling. Consumers are becoming aware and have started to purchase FMCG products online as well. If an FMCG brand does not have its e-commerce website, it can easily sell through other such platforms that have become virtual market places, e.g., Amazon and Flipkart in India. By connecting a Social Media platform with a market place gives the brand an advantage in the context of reaching beyond the geographical limits. FMCG brands have started to link their engagements on Social Media to the online portals where consumers can see and select the products. Also, Social Media allows brands to promote their products through digital advertisements. These advertisements can be sent to a selected audience based on Social Media capabilities when it comes to advertising. Dabur is one such example which sells its products on Amazon and has linked its corporate as well as a Facebook page to its online marketplace on Amazon where its products are showcased. Dabur promotes itself on Facebook only, and it has no official presence on YouTube and Instagram. Therefore, using

HOURS	
Closed Now 9:00 AM - 6:00 PM -	2012 • Achieves \$1-Billion turnover
CIUSED NOW 9.00 AM - 0.00 PM +	2011 Acquires 30-Plus brand
BUSINESS INFO	2010 Forays into professional skin care market Dabur Acquires Namasté Group of US
Started in 1884	 Acquires Hobi Kozmetik Group of Turkey
ADDITIONAL CONTACT INFO	See All
🖂 corpcomm@dabur.com	
http://www.dabur.com	
MORE INFO	
About	
Dabur India Limited is the fourth largest FMCG Company in India with Revenue of over Rs 7,800 Crore & Market Capitalisation of over Rs 35,000 Cr.	
Ø Founding date	
1884	
2 Awards	
Dabur has been ranked as the Most Trusted Leader in the Healthcare category in the Brand Trust Report 2012 Dabur India Ltd Chairman Dr. Anand Burman named E&Y E See More	
Products	
http://www.dabur.com/Products-Health%20Care http://www.dabur.com/Products-Personal%20Care http://www.dabur.com/Products-Foods See More	
Corporate Office	

Figure 3.5: Backlinks to Dabur's corporate website on its FBP's about section

Facebook to direct consumers to its corporate website where all its products are showcased along with the prices and reviews.



Figure 3.6: The Twitter post directing the user to Amazon through a link in the tweet

It uses Facebook as a platform where it does most of its engagements. Also, Dabur uses the same strategy on its Twitter brand page by attaching a shortened link in each message; it tweets about any of its products, as shown in Figure 3.6.

7. Creating campaigns in regional languages - FMCG brands are sold not only in urban cities but in semi-urban and rural sectors as well. At the same time, the Internet has penetrated the masses, and most of the villages are now digitally connected. It was reported that by the end of September 2018, 56 crores (560 million) mobile Internet users were active in India (Ganguly, 2018). This has been made possible due to a surge in rural Internet usage in the past few years. Due to this phenomenon, it is has become easier for brands to connect with consumers who can now access Social Media platforms with ease. The capabilities of digital media allow the

brand to create brand pages in regional languages. Indian FMCG brands mostly use Social Media platforms like Facebook is due to its capability of providing language options while making a brand page. Brands should make their digital media campaign taking in consideration of connecting with a regional population through Social Media platforms.

PATANJA	Chikitsalaya	REPART	Patanja Like This	li Page · February 3, 20'	•••
	1 வியாதிகளுக்கும் ஆயுர்வேத ப்படி சிகிச்சை அளிக்கப்படும்	Ayurv	edic treatn	nent available for all	diseases
Diabetes Blood pressure Psoriasis Asthma Wheezing Obesity Arthritis Cervical spondilities Rheumatoid arthritis	 சர்க்கரை வியாதி ரக்த கொதிப்பு தோல்வியாதி மூச்சிதின்றல் விஸிங் உடல் பருமன் முட்டு வலி கழுத்து எழும்பு தேய்மானம் ஆமவாதம் 	O 19 D) Like	2 Con	nments 10 Shares
 ♦ Infertility ♦ All types of skin diseases 	 சுழந்தைஇன்மை அணைத்து விதமான சரும வியாதிகள் 	Most R	leievant * Nataraj N	latu Soriyasis treatme	ent?
 ◆ Bladder stones ◆ Thyroid ◆ Sinusitis ◆ Hairfail ◆ Dandruff 	 + சிருநீரக கல் > தைராய்டு > சைனஸ் > முடி உதிர்வு > பொடுகு 		Like - Rep Relevant is s Itered out.	ly 1y selected, so some cor	nments may have
Menstruation problems Pcod I Fibroid I Cyst Piles I Haemorrhoids I Fistula Insomnia Loss of Appetite Ulder I Acidity	் மாதவிடாய்பிரச்சனைகள் ◇ கருப்பைகட்டி ◇ மூலம் ◇ தூக்கமின்மை ◇ பசிஇன்மை ◇ அல்சர்வுசிடிடி		Write a or	omment	0 6 9
 ♦ Indigestion ♦ Fever ♦ Ansemia ♦ Leucorrhoea 	∻ அஜீர்னம் ∻ ஜூரம் ∻ ரக்தசோகை ∻ வெட்டைதடு				
AYURVEDIC TREATMENT FOR ALL T	PES OF DISEASES IS AVAILABLE HERE				

Figure 3.7: Patanjali's post of its official Facebook brand page for Chennai region

Since most of the population might not be comfortable with English as a language preference on Social Media; brands should try and make brand pages in regional languages. This serves two purposes; first, the consumer feels connected with the brand, and this generates brand loyalty. Second, due to the language preferences brand page will be visible to more users, thus increasing the reach for the brand.

3.5 Conclusion

The chapter has presented an overview of the Indian FMCG market and the major FMCG companies. It also emphasised on the role of the government on supporting the FMCG sector in the Indian context. Furthermore, the involvement of Indian FMCG companies on Social Media platforms was explored; along with the major digital media strategies followed by the same on said platforms. Major strategies have been discussed that are followed by the Indian FMCG brands to engage consumers that are active on the Social Media platforms discussed in section 3.1. As it is evident that the majority of the brands have a presence on popular Social Media platform, but they fail to use these platforms to their full capacity in the context of engaging and capturing more followers. Due to the competitive environment, these brands need to come up with fresh strategies to connect with the consumers and to keep engaging with them. There is a need for these brands to focus more on the content and the information quality of the post to keep the consumers engaging, as well as for spreading the word-of-mouth from the engaging consumers to garner more followers on the respective Social Media brand page. The next chapter presents the research methodology followed in the study.

Chapter – 4

Research Methodology

4.1 Introduction

As discussed in the previous chapters, this study attempts to understand the influence of Online Consumer Engagement on consumer's Purchase Intention, through the identification of various influencing factors that are critical to a brand community. This chapter is presented in two parts. The first part identifies and reports the important factors through review of the existing literature, to propose a framework. It starts with the development of hypotheses through a review of the literature. It proposes 17 hypotheses in total based on the relationships observed in other studies. The second part of the chapter provides a detailed methodology used in this study to test the hypotheses proposed in the first part. The study follows an exploratory-cumdescriptive research design for which it uses a questionnaire as the research instrument. A standardised questionnaire is used based on the proposed framework. The chapter also presents the sampling techniques used for the study. Due to the non-availability of a sampling frame, convenience sampling was used to identify the respondents of the study. Data collection was done by both offline and online methods. Based on the type of data collected, the study has used Structural Equation Modelling as the statistical technique for the analysis of the collected data. This Chapter presents the framework and the methodology for implementing statistical analysis on the data collected to fulfill the third and fourth objectives, discussed in Chapter 1 Section 1.6.

4.2 Hypotheses development

Following factors were identified after a thorough literature review that affects consumer engagement in the Social Media environment and further consumer's intention to purchase:

4.2.1 Brand Attachment (BA)

'Brand Attachment' as a concept has been explored by the researchers in the recent years and it has played an important role in understanding the factors affecting the relationship between consumer and brand (Belaid and Behi, 2011; Louis and Lombart, 2010). Studies have discussed that the attachment experienced by consumers towards a brand is affected by their perception of the brand; and, it can act as a predictor for their attitude towards the brand (Shimp, 2010). Consumer's strong emotional attachments result in them being loyal and attached for a longer duration with the brand (Thomson et al., 2005). Previous studies have addressed and discussed the effects of attachment and attitude from consumers perspective (Park and MacInnis, 2006; Fedorikhin et al., 2008); and many have supported that, a consumer's attitude towards a brand is affected by his/her attachment with the same (Louis and Lombart, 2010). This leads to H_{1a}:

HA1a: Brand Attachment has a significant influence on Attitude Towards Brand

Literature talks about 'Brand Attachment' and connects it with OBCs and brand commitments (Thomson et al., 2005; Carroll and Ahuvia, 2006; Zhou et al., 2012). However, quite a few studies have been conducted that have studied the functional relation of BA on 'Brand Engagement'. Previous studies define BA as a construct, based on relationship, that shows that an emotional tie exists between a consumer and a brand (Park et al., 2010). Studies in the field of marketing suggest that consumers behaviours are strongly influenced by attachment (Thomson et al., 2005; Zhou et al., 2012). The longevity of the relationship between a consumer and brand depends on the strength of the consumer's emotional bond with the brand (Carroll and Ahuvia, 2006). In a brand community, consumers create strong attachments and engage with the brand in long term brand-relationship. This engagement might converge to brand equity-relevant behaviours within the community (Ahluwalia et al., 2000; Zhou et al., 2012). This leads to H_{1b} :

HA1b: Brand Attachment has a significant influence on Brand Engagement

In recent years, 'Brand Attachment' has played a major role as an important factor that affects the relationships between the consumer and brand (Louis and Lombart, 2010; Belaid and Behi, 2011). Attachment theory defines BA as an emotional link or bond shared between a brand and an individual (Thomson et al., 2005; Louis and Lombart, 2010). This bond allows the consumers to identify themselves with the brand and establish a close relationship. Studies suggest that the bond or connection that a consumer feels with a certain brand can be a predictor to consumer's purchase behaviour (McAlexander et al., 2003; Thomson et al., 2005; Esch et al., 2006; Assiouras et al., 2015). Therefore, suggesting that BA is a strong determinant of 'Purchase Intention'. This leads to H_{1c} :

HA1c: Brand Attachment has a significant influence on Purchase Intention

4.2.2 Communication (COMM)

Consumers have easy access to the Internet, which leads them away from the traditional media but closer to Social Media channels. They are actively using these platforms, searching for information and opinion, which might be useful (Mangold and Faulds 2009; Bambauer-Sachse and Mangold 2011; Schivinski and Dabrowski, 2016). Social Media has transformed itself into an essential communication channel that has a significantly influential role to play in the marketing environment. It facilitates communication between consumers and brands (Gretzel and Fesenmaier, 2012; Gretzel and Dinhopl, 2014), and among the consumers, too (Xiang and Gretzel, 2010). Brands are using Social Media platforms not only for promotion and advertisement; but also, to manage consumer queries and issues and to get deeply involved with the consumers (Solis 2010). Findings of the previous studies, on the impact of Word-of-Mouth, User Generated Content (UGC) and brands communication to create awareness among the consumers (Godes and Mayzlin 2009; Bruhn et al., 2012), suggest that Social Media 'Communication' has a positive effect on consumer's 'Attitude Towards Brand' (Schivinski and Dabrowski, 2016). This leads to H_{2a}:

HA2a: Communication has a significant influence on Attitude Towards Brand

As discussed above, communication with consumers is important for brands/marketers. Nowadays, organisations have started to include creative strategies with messages having good content as well as communication with a notion of producing desired effects in the consumer population (Laskey et al., 1989; Logan, 2014). Social Media-based campaigns targeted at consumers provide additional touchpoints to encourage the already existing interaction between the consumer and the brand. Such communications, where the brand listens to consumers, help the brand/marketer to understand the common undertones. Thus, plan and create a better strategy to engage the consumers through online content (Murdough, 2009; Ashley and Tuten, 2015). The more interactive this communication between the consumers and the brands is; more are its effects on consumers involvement (Coyle and Thorson, 2001; Stewart and Pavlou, 2002; Fortin and Dholakia, 2005). Therefore, suggesting that any information that is presented in a highly interactive format leads to a higher level of engagement (Luarn et al., 2015). This leads to H_{2b} :

HA2b: Communication has a significant influence on Brand Engagement

4.2.3 Content (C)

In today's digital scenario, OBCs and SNSs must keep the users engaged. Majority of the OBCs on SNSs are moderated to oversee the exchange of communication in the context of the content within the community. 'Content' of the posts from the brand or any of the users hold high importance from the point of online engagement. Moreover, the informational value of the content was found to be more relevant for consumer's motivation to engage with the brand on online platforms. (Peng et al., 2004; Cvijikj and Michahelles, 2013). Content posted online can

easily bind the users by conditioning their feelings, and making them feel connected positively. This may result in having an optimistic overview about the engagement and which may lead to a positive 'Attitude Towards Brand', with which the engagement was associated (Chen and Wells, 1999; Chen et al., 2002; Peng et al., 2004; Coursaris et al., 2016). This Leads to H_{3a} :

HA3a: Content has a significant influence on Attitude Towards Brand

As discussed above, branded social 'Content' can be used for promotional purposes to influence the consumer's 'Attitude Towards Brand' and also share the same content within their peer groups/networks (Ashley and Tuten, 2015). Previously conducted studies suggest that the content posted through the various SNSs such as Facebook, affect how users engage in an OBC (Cvijikj and Michahelles, 2013). Also, the attributes of the content posted (informativeness, vividness, enjoyment, etc.) can be an important factor of consumer's engagement on a brand page (Taylor et al., 2011; De Vries et al., 2012; Chauhan and Pillai, 2013; Tafesse, 2015). This leads to H_{3b}:

HA3b: Content has a significant influence on Brand Engagement

4.2.4 Perceived Benefits (B)

While engaging with brand communities, consumers derive certain benefits from the brand page in the form of online coupons, discounts, information about the sale, etc. Brands/marketers often use SNSs to offer such benefits to consumers to encourage them to interact more with the brand (Gummerus et al., 2012). Studies suggest that the consumer tend to hold a positive image about such brands/marketers and stay loyal to those whom they think to be caring towards them (McKnight et al., 2002; Shi et al., 2016). Consumers that receive information about discounts and other benefits, monetary or otherwise, tend to hold a positive attitude towards the brand/marketer and thus leading to more interactions as well. Consumers with a high level of interactions with the brand page tend to receive relevant information from

the brand and thus develop a positive 'Attitude Towards Brand' (Dholakia & Durham, 2010, Shi et al., 2016). This leads to H_{4a}:

HA4a: Perceived Benefits has a significant influence on Attitude Towards Brand

Consumer's connection or engagement with a brand is driven by their utilitarian motives, which varies from getting the latest information about a brand and benefits it is offering which might be of use to the consumer (Zaglia, 2013). SNSs nowadays provide the environment where this need for useful information and connection can be observed (Raacke and Bonds-Raacke 2008; Bonds-Raacke and Raacke 2010). Brand post information about the latest products/services and other benefits that can be availed by the consumers. Majority of the users are that follow any brand Facebook or Twitter seeking sales page on information/discounts/coupons posted by the brand on the brand page/profile (Logan, 2014). The practical benefits of engaging with a brand page are the prime motivator for consumers to engage more with a brand on Social Media platforms. This leads to H_{4b:}

HA4b: Perceived Benefits has a significant influence on Brand Engagement

4.2.5 Credibility (CRED)

Consumers that are active on an SNS's brand page are seeking credible information and updates that might find relevant and important to keep them motivated to engage with the brand. (Dholakia et al., 2009). Brand pages that provide the consumers with relevant and high-quality information in the form of promotion, sale, product/service information are regarded as a valuable source (Shi et al., 2016). The quality of the information, i.e., credible and accurate, makes the consumer feel that their engagement with the brand are worthwhile due to the useful information they are getting from the online community (Gummerus et al., 2012). Quality of information along with credibility tends to make the consumer feel confident about the brand

while engaging within in a brand page, as it makes them feel that the brand is providing them with authentic and genuine information (Cvijikj & Michahelles, 2011). This leads to H_{5a} :

HA5a: Credibility has a significant influence on Attitude Towards Brand

Brand pages have become popular and gain traction in the number of followers based on the posts uploaded on the page. If these posts provide information about the new arrival of products, product description, helping in choosing and deciding on a specific product, then consumers find the brand page to be useful (De Vries et al., 2012). Previous studies have suggested that consumers are more likely to engage with a brand page for searching for useful information (Park et al., 2009; Lin and Lu, 2011). This useful information is generally about new products or services offered by the brand, marketing activities, including sales and promotions (Muntinga et al., 2011; De Vries et al., 2012). The informational content posted on an FBP lets the users identify between the product alternatives and make better choices. Therefore, they engage more with brand pages because of the informational posts. (Muntinga et al., 2011; Luarn, et al., 2015). The usefulness of the information and the credibility comes from the fact that the official FBP has posted the information for the users to seeks and utilise leading to higher intent to visit the brand page (Raney et al., 2003). This leads to H_{3b}:

HA5b: Credibility has a significant influence on Brand Engagement

4.2.6 Critical Mass (CM)

Individuals that hold importance in the social network of a user make up the 'Critical Mass'. It holds peers in the close-knit network with whom the user engages the most on SNSs, or for whom he/she engages in the SNSs (Sweeney et al., 2008; Chu and Kim, 2011; Rauniar et al., 2014; Coulter and Roggeveen, 2012). Studies have shown that consumers tend to adhere to the expectation of the people that are considered to be in their critical mass while making decisions. Their actions and decisions are from the point of seeking approval from the same individuals

by acquiring a product or service that their critical mass deems fit (Venkatesh and Davis, 2000; Nysveen, 2005; Amin, 2007). Therefore, users constantly seek opinions from the people they feel are important and influence their decisions. Consumers while using Social Media platforms exhibit behaviour which can be associated with electronic word-of-mouth. They tend to turn towards their contacts on the platform for information regarding any product or services (Chu and Kim, 2011). Thus, it can be said that a user's close network affects his/her judgement regarding decision making or influencing the user's attitude. This leads to H_{6a} :

HA6a: Critical Mass has a significant influence on Attitude Towards Brand

The social circle of an individual and its influence has played a crucial part in disseminating market information (Bearden et al., 1989; Goldenberg et al., 2001). As discussed in the section above, any individual's behaviour is influenced by their network, types of friends, the tie strength and group cohesion (Lewis et al., 2008; Cao et al., 2009; Chu and Kim, 2011; Wilcox and Stephen, 2013). There is a dearth of literature where the effects of the peer groups or the critical mass, which influences an individual's decisions have been studied. When it comes to SNSs and FBPs, it can be assumed that when information about a brand page is received by a person belonging to the user's 'Critical Mass' (as described in the previous hypothesis H_{6a} ,); the same has a strong influence on their intention to engage with the brand page. Previous studies suggest that consumer's 'Critical Mass' exerts a greater influence for the consumers to join an FBP and indulge in brand engagement (Brown and Reingen, 1987; Bansal and Voyer, 2000; Goldenberg et al., 2001; De Bruyn and Lilien, 2008; Palazon et al., 2015). This leads to H_{6b} :

H_{A6b}: Critical Mass has a significant influence on Brand Engagement

Studies on user's adoption of technology have suggested 'Critical Mass' to be an important factor (Glass and Li, 2010; Lou et al., 2000; Cheng et al., 2012). As defined previously it is the

close-knit peer group of a user for which or with whom user engages on a Social Media platform (Sweeney et al., 2008; Chu and Kim, 2011; Rauniar et al., 2014; Coulter and Roggeveen, 2012). For a user active on a Social Media platform, his/her Social Media network holds people he/she feels to be important. OBCs on Social Media platforms deals in user-generated content and its sharing; thus, users tend to rely on his/her peer network within the community for their suggestions/opinions. Users lean towards the expectations of their critical mass while making purchase decision to acquire a product or service (Venkatesh et al., 2003; Madan and Yadav, 2016). Since this study attempts to explores consumers behavioural intentions; 'Purchase Intention' to be precise, CM shows significant relevance to be considered to affect the same. (Hsu and Lu, 2004; Yu et al., 2005; Cheng et al., 2012). This leads to H_{6c} :

HA6c: Critical Mass has a significant influence on Purchase Intention

4.2.7 Attitude Towards Brand (ATB)

Literature suggests that there is a functional relation between 'Attitude Towards Brand' and the 'Brand Engagement' on the brand's Social Media brand page. Literature talks about the attitude, but a clear relational focus on the relation between ATB and BE has not been explored thoroughly. Very few studies exist that suggests that the engaging content posted by a brand results in a positive attitude towards the brand. They further strengthen the possibility that the consumers will continue to visit the brand page; and that, further brand engagement from the consumers part is to be anticipated (Logan, 2014; Coursaris et al., 2016). This leads to H_{7a} :

HA7a: Attitude Towards Brand has a significant influence on Brand Engagement

Studies on consumer behaviour have suggested that the consumer's intention to purchase is strongly influenced by the attitude consumers hold for a product/service (Ajzen and Fishbein, 1980). It has been found that the same holds in the context of brands; i.e., consumers attitude towards a brand affects their behavioural intent towards purchase of the brand (Fishbein and Ajzen, 2005; Jin and Suh, 2005; Leonidou et al., 2010; Limbu et al., 2012; Punyatoya, 2015). This leads to H_{7b}:

HA7b: Attitude Towards Brand has a significant influence on Purchase Intention

4.2.8 Brand Engagement (BE)

Brand engagement is a crucial component for a brand to consider while building a constructive relationship with consumers (Keller, 2001). It can be described as the emotional tie that binds the consumer to the brand, which makes the consumer return back for more engagements (Goldsmith, 2012). It has been reported that users exhibit purchase behaviour after engaging in online brand communities on Social Media and other digital platforms (Vision Critical, 2013). Consumer's engagement with a brand, the brand shows their readiness to be involved, to gather information, to talk about it and create awareness to other individuals through word-of-mouth (Keller, 2001). Social Media and SNSs differ from the traditional media and provide the consumers with the capabilities to generate and share user content and interact with others 2008). Marketing professionals identified brand-related (Chung, have that advertisement/information/content shared among friends/critical mass shows more significant effects on brand awareness and purchase intentions (Lee, 2010; Khang et al., 2012). Previous studies have corroborated that purchase intention is affected by brand engagement (Appelbaum, 2001; Hollebeek et al. 2014; Song and Yoo, 2016). This leads to H₈:

HA8: Brand Engagement has a significant influence on Purchase Intention

4.3 Proposed framework

Based on the different factors identified; following the proposed framework is conceptualized,



as shown in Figure 4.1.

Figure 4.1: Proposed Research Framework

4.4 Hypotheses

Alternate hypotheses are as follows:

H_{A1a}: Brand Attachment has a significant influence on Attitude Towards Brand
H_{A1b}: Brand Attachment has a significant influence on Brand Engagement
H_{A1c}: Brand Attachment has a significant influence on Purchase Intention
H_{A2a}: Communication has a significant influence on Attitude Towards Brand

H_{A2b}: Communication has a significant influence on Brand Engagement

H_{A3a}: Content has a significant influence on Attitude Towards Brand

H_{A3b}: Content has a significant influence on Brand Engagement

HA4a: Perceived Benefits has a significant influence on Attitude Towards Brand

HA4b: Perceived Benefits has a significant influence on Brand Engagement

HA5a: Credibility has a significant influence on Attitude Towards Brand

H_{A5b}: Credibility has a significant influence on Brand Engagement
H_{A6a}: Critical Mass has a significant influence on Attitude Towards Brand
H_{A6b}: Critical Mass has a significant influence on Brand Engagement
H_{A6c}: Critical Mass has a significant influence on Purchase Intention
H_{A7a}: Attitude Towards Brand has a significant influence on Brand Engagement
H_{A7b}: Attitude Towards Brand has a significant influence on Purchase Intention
H_{A7b}: Attitude Towards Brand has a significant influence on Purchase Intention

4.5 Research design

This study attempts to understand the influence of Online Consumer Engagement on consumer's purchase intention, through the identification of various influencing factors which are critical to a Facebook Brand Page. Exploratory-Cum-Descriptive research design has been followed to meet the research objectives discussed in Section 4.2.

The first stage of the study involved an exploratory research design that developed a theoretical understanding of the consumer engagement construct and its dimensions. A systematic literature review was conducted to sequentially document and discuss the various antecedents and consequences of the OCE; thus, fulfilling the first objective of the study. Special attention was given to the strategies that are followed by the brand in the FMCG sector for online consumer engagement, in line with the second objective.

The second stage employed a descriptive research design for examining the dimensions identified in the first phase. The scale to measure the dimensions was developed and using various quantitative techniques, the relationships between the dimensions were examined, including testing the hypotheses for the said relationships, fulfilling the third and fourth objective of the research.

The third stage of the research discussed the findings and results of the empirical analysis of descriptive research conducted in the second phase. This section also includes detailed

discussion based on the results of the study. Further, one case on Patanjali Ayurved Limited (PAL), one of the emerging players in the Indian FMCG sector, was developed to test and corroborate the findings of the empirical study done in the second phase. The case study focused on the online strategies followed by PAL to engage the consumers on the FBP moderated by PAL.

4.6 Research instrument

A structured questionnaire is used as a research instrument which has been designed using standardised questionnaires from similar. Table 4.1 shows the constructs and the studies from where the standardized questions were adopted.

S. No.	Construct	Sources
1	Brand Attachment	Louis and Lombart, 2010
2	Communication	Schivinski and Dabrowski, 2015
3	Content	Peng et al., 2004
4	Benefits	Rauniar et al. 2014
5	Credibility	Shi et al., 2016
6	Critical Mass	Van Slyke et al., 2007; Rauniar et al., R., 2014
7	Attitude Towards Brand	Spears and Singh, 2004
8	Brand Engagement	Kim and Johnson, 2016
9	Purchase Intention	Md Husin et al., 2016

Table 4.1 Constructs and their sources

In total, nine constructs were identified as discussed in Section 4.2. Eight constructs were measured using a seven-point Likert scale where, 1-stands for strongly disagree, 2 disagree, 3-somewhat disagree, 4-neutral, 5-somewhat agree, 6-agree and 7-strongly agree. One of the constructs was measured on a seven-point semantic differential scale. Likert scale was used to increase the precision in the degree of expression. The questionnaire was pre-tested for its structure on a sample of 48 respondents to test the reliability of the questionnaire. These

respondents were those people who were active on the SNS, Facebook. The data collected for the pilot study were not included in the principal analysis.

4.7 Sampling technique

This study adopted a non-probability convenience sampling technique, which is considered appropriate in such studies (Hair et al., 2010). For probability sampling, one needs to have a sampling frame where the population is well defined concerning the characteristics under study. Since the sampling frame for the population is lacking, a non-probabilistic convenience sampling was used in this study; the respondents were selected keeping in view the criteria that they are actively using Facebook and are familiar with the concept of brand owned FBPs. Some studies shed light on behavioural intents of consumers and have employed and used convenience sampling to collect the primary data (Featherman and Pavlou, 2003; Wu and Wang, 2005; Lewis et al., 2010).

Literature does not suggest a predefined set of rules for the selection of sample size since it depends primarily on the number of the variable and the level of complexity of the model. The sample size should be adequate to represent the population on which the study is being conducted. For conducting quantitative analysis, studies have suggested a sample size of 200 to 400 (Netemeyer et al., 2004; Hair et al., 2010). Following table categorise the minimum sample size as per the complexity of the model (Hair et al., 2010):

(adapted from fran et al., 2010)			
Minimum sample size	Number of constructs	Number of items	
100	Five or fewer constructs	More than three items	
150-300	Seven or less constructs	Not important	
500	Large number of constructs	less than three observed measures	

 Table 4.2 Selection of sample size based on the number of constructs and items (adapted from Hair et al., 2010)

Based on the above suggestions, this study has incorporated a sample of initial 533 responses, out of which 501 were deemed fit for analysis (Hair et al., 2010). The sample size comprised of the Facebook users who follow any Facebook based brand pages and actively engage in such brand communities.

4.8 Data collection

Data collection was done in two phases:

- i. An offline Pilot study was conducted based on a self-administered structured questionnaire to get insights of the respondents as well as to check the reliability of the questionnaire. Respondents were asked to answer the questions specified on a seven-point Likert scale.
- ii. The final survey questionnaire for data collection using both offline and online method.

The offline method was used to collect responses from the respondents during the pilot study. The offline method was employed in this phase of data collection only as the respondents could clarify any query regarding any of the survey questions. The online survey was created using Google Forms®, an online service through which a link was generated, which can be used by any of the respondents to access the online survey questionnaire. The online survey was sent to the respondents as a link through the message delivery application on Facebook®, emails, and other electronic media. This method was found to be useful and convenient since the respondents were already actively using Facebook and were expected to know the FBPs. Use of online survey questionnaire addressed the problems of the incomplete survey. The survey questionnaire was also filled by the respondents. The survey questionnaire was circulated to the respondents in December 2017 – January 2018, where a total of 533 respondents filled the survey questionnaire.

4.9 Statistical techniques and tools for data analysis

The study used the statistical software package IBM® SPSS STATISTICS version 20.0 for data analysis along with Microsoft® Excel 2013, primarily for data screening and for checking the reliability of the questionnaire during the pilot testing. Also, IBM® SPSS AMOS Version 20.0 was used for further data screening. It was primarily used for Confirmatory Factor Analysis (CFA) and Structural Equation Modelling (SEM) for confirming the validity of the constructs, hypotheses testing, for validating the proposed model and to analyse the group differences through critical ratios within different demographic groups.

4.10 Conclusion

The Chapter presented the research methodology employed to test the proposed hypothesised framework for the study. The study employed an exploratory-cum-descriptive research design. For the exploratory research design, the study identified important variables that conform to consumer's online engagements were identified along with the dependent. Based on the literature on consumers engagement in an online setting, major nine variables were identified. A proposed conceptual framework with relevant literature to support the development of the hypotheses was presented. Further, the proposed framework depicting the formative relationships has been presented. For the descriptive research design of the study, variables identified as discussed above were utilised to develop a questionnaire for collecting data. The methodology behind the sampling technique and selection of the sample presented, along with methods of data collection and the statistical tools and techniques used for data analysis to streamline the descriptive research. The next chapter presents the data analysis of the data collected as per the methodology discusses in this chapter.

Chapter – 5

Data Analysis and Results

5.1 Introduction

This chapter attempts to fulfill the third and fourth objectives of the study (Chapter 1 Section 1.6). The chapter starts with discussing the analysis of the demographic profiles of the sample. To check the reliability of the questionnaire, a pilot study was undertaken. The results of the study showed that the questionnaire shows high reliability; thus, the questionnaire survey was deemed fit for collection from the sample population. Also, the same was vetted by a few experts to remove any discrepancies in the questionnaire before its final circulation. For the final study, 533 responses were collected from the sample population. After the process of data cleaning and identification of outliers, 501 responses were deemed fit for the analysis. Further, the data was tested for assumptions of multivariate analysis has been implemented on the dataset to test the divergent and convergent validity of the variables; and to test the fitness of the proposed model. Path analysis was implemented on the collected data to test the hypotheses developed in Chapter 4 Section 4.2. Out of the 17 hypotheses, 12 were accepted. The chapter further analyses and presents the demographic variables for group differences for the proposed framework. The following section elaborates the analysis of data and the results obtained.

5.2 Demographic profile of respondents

The following section discusses a few characteristics of the respondents; namely, gender, age, annual income, occupation, Internet usage. The final study comprised of 501 respondents that were active on various FMCG FBPs. Their demographic profile is reflected in Table 5.1. Out of 501 respondents, 40.5% were females, and 59.5% were male. In age, the biggest segment of

respondents belonged to the age groups of 21-25 years (32.7%), followed by the age groups 26-30 years (24%), and 31-35 years (14.2%). As shown in table, the biggest income group was below INR 5,00,000 per annum (28.3%) followed by the income groups above INR10,00,000 per annum (26.5%) and INR 5,00,000-7,00,000 per annum (25.9%). Majority of respondents were Government/Private sector employees (45.5%) followed by Self Employed Professionals (27.5%) and Entrepreneurs/Traders (15.2%). Details about their Internet usage every week were also collected from respondents to understand their Internet usage patterns. Majority of the respondents used the Internet for more than 15 hours/week (50.7%) followed by about 10 hours/week (22.0%) and about 15 hours/week (19.8%).

Characteristics	Frequency	Percentage
Gender		
Female	203	40.5
Male	298	59.5
Age		
Below 20	38	7.6
21-25	164	32.7
26-30	120	24.0
31-35	71	14.2
36-40	52	10.4
Above 40	56	11.2
Annual Family Income (INR per annum)		
Below 5,00,000	142	28.3
5, 00,000 - 7, 50,000	130	25.9
7, 50,000 – 10, 00,000	96	19.2
Above 10,00,000	133	26.5
Occupation		
Student	51	10.8
Govt. / Private sector employee	228	45.5
Self-employed professional	138	27.5
Entrepreneur/Trader	79	15.2
Other	5	1.0
Internet Usage		
Less than 5 hours/week	38	7.6
About 10 hours/week	110	22.0
About 15 hours/week	99	19.8
More than 15 hours/week	254	50.7

Table 5.1: Demographic Details

5.3 Pilot study

Once the survey questionnaire was completed. It was tested to check its reliability. Offline mode was used for testing of the questionnaire as it gives better chances for the respondents to understand the questionnaire and respond. The interaction provided valuable feedback to refine the questionnaire. The pilot study was conducted within Delhi NCR. Participants were given a printed copy of the questionnaire. 15 minutes were given to each respondent to complete the survey, but most of them were able to complete the survey within 6-8 minutes. Participants were encouraged to ask questions regarding any unclear survey question or the underlying connotations. The desired number of participants were between 40-60. Total of fifty-eight thorough responses were collected during the pilot study, which was deemed fit to be analysed further for checking the reliability of the questionnaire.

5.4 Questionnaire reliability

The study employed a standardised survey questionnaire for which the reliability of the same was checked. To assess the same, Cronbach alpha (α) was computed, which measures the internal consistency of the items (Cronbach, 1951). The internal consistency reliability is "the degree to which responses are consistent across the items within a measure" (Netemeyer et al., 2003). IBM SPSS STATISTICS® version 20.0 was used to assess the reliability of the survey questionnaire. Previous studies suggest that the Cronbach alpha should be > 0.7 for internal consistency to be acceptable (Nunnally, 1978; Hair et al., 2010). Table 5.2 depicts the Cronbach alpha for all the items within a construct where the internal consistency of all the constructs was found to be within acceptable range, i.e., > 0.7. Apart from the mentioned measures for testing the reliability, there are reliability tests, including composite reliability and the average variance extracted, which are derived from the confirmatory factor analysis (CFA) discussed in section 5.6.

S. No.	Construct	No. of Items	Cronbach alpha (α)	
1	Brand Attachment	3	0.759	
2	Communication	4	0.839	
3	Content	3	0.828	
4	Benefits	4	0.772	
5	Credibility	3	0.725	
6	Critical Mass	5	0.762	
7	Attitude Towards Brand	5	0.815	
8	Brand Engagement	3	0.849	
9	Purchase Intention	3	0.863	

 Table 5.2: Construct reliability

5.5 Testing the assumptions for multivariate analysis

The underlying assumptions of multivariate analysis, i.e., whether the dataset can be used for multivariate analysis or not, were also tested. These assumptions are about missing data, outliers, normality, linearity, homoscedasticity and multicollinearity, which are discussed as follows.

5.5.1 Missing Data

The data was checked for missing values as it might pose problems in analysis. Missing values were found using descriptive analysis in IBM® SPSS Statistics, and no missing values were found in the dataset.

5.5.2 Outliers

Outliers are the cases present in a dataset which are different from the other values in terms of extreme difference. Squared Mahalanobis Distance (D^2) was computed to detect the outliers in the dataset. Outliers have a significantly different value for D^2 than the other values of D^2 in the dataset (Byrne, 2010). Appendix B shows the table of Squared Mahalanobis Distance, which suggests that there exists minimal evidence of any extreme outlier in the dataset.

5.5.3 Normality

The most critical assumption for implementing SEM analysis on a dataset is that the data needs to be normal. Normality of a dataset is checked for the presence of skewness and kurtosis in the normal distribution curve of the dataset at hand; Skewness of the curve represents the balance of distribution (if distribution is on the left side or the right side); whereas, kurtosis represents the flatness or peakedness in the distribution (Hair et al., 2010). Table 5.3 shows that values of measure of skewness and kurtosis are well under the acceptable values (skewness $< \pm 3$ and kurtosis $< \pm 10$) (Hair et al., 2010).

	Mean	SD	Skewness	SE for Skewness	Kurtosis	SE for Kurtosis
GEN	1.59	.491	387	.109	-1.857	.218
INCOME	2.44	1.161	.114	.109	-1.441	.218
FREQUENCY	3.14	1.007	723	.109	820	.218
AGE	3.01	1.280	.782	.109	278	.218
OCCUPATIONS	2.32	.752	.390	.109	046	.218
В	5.43	1.070	981	.109	1.431	.218
COMM	4.90	1.175	644	.109	.469	.218
С	5.01	1.086	598	.109	.644	.218
CRED	4.86	1.057	538	.109	.355	.218
СМ	4.26	1.164	285	.109	310	.218
BE	4.91	1.095	672	.109	.892	.218
BA	4.44	1.293	486	.109	.027	.218
ATB	5.22	.877	196	.109	.470	.218
PI	4.88	.973	334	.109	.242	.218

Table 5.3: Skewness and Kurtosis Statistics

5.5.4 Linearity

Another most important assumption is the linearity of the dataset, which is required for multivariate analysis. To check the linearity of the dataset, scatter plot was used where the data points were found to be distributed on a linear fashion along the fit line; thus, reflecting on the linearity of the dataset. Figure 5.1 shows the scatter plot, which shows the data scattered along the fit line indicating the linearity of the dataset.



Figure 5.1: Scatter Plot diagram

5.5.5 Homoscedasticity

Homoscedasticity is an essential assumption for applying multivariate analysis on a dataset. A set of data can be termed as homoscedastic if there is an equal level of variance demonstrated by the dependent variable with respect to the predictor variable (Hair et al., 2010). Data needs to be homoscedastic to run a regression analysis, to check the same, scatter plot of standard predicted and residual values were used to establish the homoscedasticity of the dataset. The scatter plots suggested that the dataset was Homoscedastic and thus suitable for regression analysis. Figure 5.2 to Figure 5.17 shows the scatter plot depicting homoscedasticity in the dataset.



Figure 5.2: Scatter plot of Perceived Benefits and Attitude Towards Brand



Figure 5.3: Scatter plot of Brand Attachment and Attitude Towards Brand



Figure 5.4: Scatter plot of Communication and Attitude Towards Brand



Figure 5.5: Scatter plot of Content and Attitude Towards Brand


Figure 5.6: Scatter plot of Credibility and Attitude Towards Brand



Figure 5.7: Scatter plot of Critical Mass and Attitude Towards Brand



Regression standardized Residual

Figure 5.8: Scatter plot of Perceived Benefits and Brand Engagement



Figure 5.9: Scatter plot of Brand Attachment and Brand Engagement



Regression Standardized Residual

Figure 5.10: Scatter plot of Communication and Brand Engagement



Figure 5.11: Scatter plot of Content and Brand Engagement



Figure 5.12: Scatter plot of Credibility and Brand Engagement



Figure 5.13: Scatter plot of Critical Mass and Brand Engagement



Figure 5.14: Scatter plot of Brand Attachment and Purchase Intention



Figure 5.15: Scatter plot of Attitude Towards Brand and Purchase Intention



Regression Standardized Residual

Figure 5.16: Scatter plot of Brand Engagement and Purchase Intention



Regression Standardized Residual

Figure 5.17: Scatter plot of Critical Mass and Purchase Intention

As seen in the scatter plots, all the plots have a similar scatter. This similar scatter is due to the same random disturbance among the relationships between the dependent variable and independent variables. This depicts the homogeneity of variance among the relationships, thus, suggesting the homoscedasticity of the dataset.

5.5.6 Multicollinearity

Multi-collinearity refers to the presence of correlations among the dependent variables, which causes variability in results. Multicollinearity issues were examined through the collinearity statistic, under which two statistics are taken into consideration, i.e., tolerance and variance inflation factor (VIF) of the independents in the proposed model. Tolerance is defined as the amount of variability of the selected independent variable, value for which should be > 0.1(Hair et al., 2010). VIF is the inverse of the tolerance value, value for which should be < 3 (Hair et al., 2010). Table 5.4 shows the Tolerance and VIF values for the independent variables showing that they are well within the acceptable range.

	v	
Constructs	Tolerance	VIF
В	.638	1.568
COMM	.756	1.323
С	.629	1.590
CRED	.605	1.653
CM	.637	1.570
BE	.539	1.856
BA	.591	1.691
ATB	.504	1.697

Table 5.4: Collinearity Statistics

5.6 Confirmatory Factor Analysis (CFA)

5.6.1 Convergent and discriminate validity

For testing the convergent and discriminant validities, data was analysed using IBM SPSS AMOS® version 20.0. The measurement model was made and ran on the statistical software to find both model validities as well as the model fit, which is discussed in section 5.6.2.

The convergent validity refers to the high proportion of variance shared by all measured variables of a specific latent variable in common (Hair et al., 2010). Convergent validity is measured by Composite reliability (CR) along with the average variance extracted (AVE). Composite reliability is also used to check the internal consistency, where AVE reflects 'the mean-variance extracted for the items loading on a construct' and is a summary indicator of convergence. Literature suggests that to confirm the convergent validity (correlation among the items within a factor), CR > 0.70, AVE > 0.50, where CR > AVE (Hair et al., 2010; Liébana-Cabanillas et al., 2014; Mohamed et al., 2014; Xu et al., 2014). The values obtained after the calculations were found to be well within the acceptable ranges, which are represented in Table 5.6.

	CR	AVE	MSV	СОММ	BA	С	В	ATB	СМ	BE	CRED	PI
СОММ	0.838	0.571	0.248	0.755								
BA	0.827	0.617	0.396	0.306	0.785							
С	0.734	0.508	0.430	0.447	0.524	0.693						
В	0.819	0.531	0.430	0.483	0.406	0.656	0.728					
ATB	0.867	0.566	0.292	0.325	0.407	0.499	0.452	0.752				
СМ	0.850	0.539	0.352	0.264	0.593	0.425	0.295	0.366	0.734			
BE	0.773	0.534	0.402	0.474	0.629	0.634	0.581	0.540	0.525	0.731		
CRED	0.772	0.533	0.349	0.498	0.491	0.565	0.589	0.508	0.469	0.591	0.730	
PI	0.775	0.535	0.378	0.304	0.615	0.455	0.463	0.517	0.504	0.559	0.456	0.732

 Table 5.5: Reliability and validity of constructs

Note: CR: composite reliability; AVE: average variance extracted; MSV: maximum shared squared variance

The discriminant validity assesses whether a construct is distinctively non-similar from other constructs (Hair et al., 2010). By testing discriminant validity, it is verified that the construct is unique from other constructs. Discriminant validity was measured by calculating the values for AVE; maximum shared squared variance (MSV). The values for MSV and AVE were well

within the acceptable ranges as shown in Table 5.3, i.e., MSV<AVE; therefore, there are no discriminant validity issues (Hair et al., 2010; Wong et al., 2014; Zhou, 2014). The square of the Average Variance Extracted (AVE) for each construct is greater than the inter-construct correlations as seen in Table 5.5, suggesting that there are no discriminant validity issues (Fornell and Larcker 1981). Thus, no convergent and discriminant validity issues exist within and among the constructs.

5.6.2 Measurement model

The measurement model, as shown in Figure 5.18, was used to assess the measurement model, which was followed by an examination of the hypotheses by the use of the structural model. Six indices were taken into consideration as suggested by previous studies, i.e., χ^2/df (≤ 3.00), comparative fit index (CFI ≥ 0.90), goodness-of-fit index (GFI ≥ 0.90), Standardized Root Mean Square Residual (SRMR ≤ 0.08), root mean square error of approximation (RMSEA ≤ 0.08) and p-Value of close fit (PClose ≥ 0.05) (Gaskin, 2012; Leong et al., 2011; Hair et al., 2010; Arbuckle and Wothke, 1999; Browne and Cudeck, 1993).

Table 5.6 shows the model fit indices. The final measurement model consisted of nine constructs and 33 reflective items. Table 5.7 shows the items with factor loadings being well above the acceptable cut-off level of 0.05, which are considered significant (Hair et al., 2010). These items were considered for further analysis and used in the structural model discussed in section 5.6. Figure 5.19 depicts the output for the same.

The measurement model was further tested for Common Method Bias (CMB) using a common latent factor. CMB was checked through the Common Latent Factor method in AMOS®. It was found that there exists a CMV of about 31 % in the data set, which is under an acceptable level of 50% (Podsakoff et al., 2003).

		Table 5.6: Mo	odel Fit Indices	5					
Measurement Model									
χ²/df	CFI	GFI	SRMR	RMSEA	PClose				
1.899	0.949	0.905	.0494	0.042	0.998				
Structural Mod	lel								
χ^2/df	CFI	GFI	SRMR	RMSEA	PClose				
1.895	0.948	0.905	.0502	0.042	0.999				
Acceptable Rang	es of Values								
χ^2/df	CFI	GFI	SRMR	RMSEA	PClose				
Between 1 and 3	>0.95	>85	<0.08	< 0.06	>0.05				

Note: CFI = comparative fit index; GFI = goodness-of-fit index; SRMR = Standardized Root Mean Squared Residual; RMSEA = root mean square error of approximation; PClose = p-value of Close Fit.



Figure 5.18: Measurement model



Figure 5.19: Measurement model output

	Table 5.7: Factor loading for the items							
S. No.	Name of Dimension	Factor Loadings ¹						
	Brand Attachment							
1	I have a lot of affection for this brand.	0.893						
2	I am attracted to this brand.	0.881						
3	I remain involved with a brand on Facebook due to the affection I have for the brand.	0.791						
	Communication							
4	I think brand pages provide a convenient platform for the consumers to voice their complaints and suggestions to the brand.	0.838						
5	I think it is possible to communicate instantly with brands through brand pages without any time and space boundaries.	0.876						
6	I think brand pages make it easy to connect and communicate with the brand.	0.864						
7	I feel that the brands do take actions on the complaints and suggestions posted in comments.	0.671						
	Content							
8	I like the content posted (Images, Videos, Articles) on the brand page.	0.821						
9	Games and/or videos created by brands are fun and keeps me engaged.	0.795						
10	I think the entertaining content (images, videos, articles) on brand page positively influences the brand's image.	0.81						
	Benefits							
11	Facebook brand page enables me to get connected and stay in touch with the brand.	0.802						
12	Facebook brand page makes it easier for me to stay informed about the brand and its product.	0.863						
13	Latest information about the discounts and promotions can be obtained from the brand page without visiting any stores and/or shops.	0.785						
14	Following brand pages on Facebook keeps me updated about new offerings (product, discounts, etc.).	0.841						
	Credibility							
15	I think that the brand page provides reliable information.	0.851						
16	I think brand pages are a reliable information source that enables a transparent channel between brands and consumers.	0.857						

a for the it Table 5 7. F . . adir

17	I think that the information offered by my brand page is sincere and honest.	0.763
	Critical Mass	
18	People who influence my behaviour think that I should Engage with the brand page.	0.808
19	People who are important to me think that I should Engage with the brand page.	0.828
20	People in my environment (friends and colleagues) suggests that I should Engage with the brand page.	0.859
21	Many people I communicate with, are active on brand pages.	0.752
22	The people I communicate with will continue to use brand pages in the future.	0.693
	Attitude Towards Brand	
23	Bad Good	0.828
24	Unfavourable Favourable	0.787
25	Unlikable Likable	0.834
26	UnappealingAppealing	0.759
27	UnpleasantPleasant	0.811
	Brand Engagement	
28	I am interested in learning more about this brand.	0.849
29	I would be interested in other products offered by this brand.	0.855
30	I like to visit the website for this brand.	0.773
	Purchase Intention	
31	I will definitely purchase the product in the future, for a brand that I follow on Facebook.	0.829
32	I will continue to buy the brand that I follow on Facebook.	0.856
33	I will refer the brand to my friends/family/colleagues, that I follow on Facebook. or loadings greater than 0.5 is acceptable (Hair et al. 2010)	0.803

5.7 Structural model

The structural model analysis assesses the hypothesised relationships as proposed in Chapter 4 Section 4.2. The proposed model (Chapter 4 Section 4.3) is used to build and analyse the structural model using IBM SPSS AMOS®. As the model depicted in Figure 4.1 in Chapter 4 Section 4.2 shows; 'Brand Attachment', 'Communication', 'Content', 'Benefits', 'Credibility' and 'Brand Engagement' act as independent constructs with paths leading to 'Attitude Towards Brand' and 'Brand Engagement'. Further, 'Attitude Towards Brand' and 'Brand Engagement' in turn, act as an independent construct and have paths leading to 'Purchase intention', an endogenous construct. Figure 5.20 Shows the Structural model used for analysis.



Figure 5.20: Structural Model

This section discusses the results of the structural model. Model fit indices similar to the ones used for assessing the measurement model (section 5.6.2), were used for assessing the structural model as well. Namely, χ^2/df , comparative fit index (CFI) goodness-of-fit index (GFI), Standardized Root Mean Square Residual (SRMR), root mean square error of approximation (RMSEA) and p-Value of close fit (PClose) were the model fit indices which

were taken into consideration. The structural model analysis was conducted on the model without making any changes to the model. The analysis tested the 17 hypotheses discusses in Chapter 4, section 4.4. Out of 17 hypotheses, 5 hypotheses were not supported. The Structural model output is depicted in Figure 5.21. moreover, Table 5.8 shows the results of the hypothesised relationships



Figure 5.21: Structural Model Output

Majority of the paths hypothesised during the study were found to be significant; namely, Brand Attachment \rightarrow Attitude Toward Brand, Communication \rightarrow Attitude Towards Brand, Perceived Benefits \rightarrow Attitude Towards Brand, Credibility \rightarrow Brand Engagement. Therefore, H_{A1a}, H_{A2a}, HA4a, and H_{A5b} were not supported by the empirical results; since, t-values < 1.96 (H0 Accepted) and p < 0.05 (H0 Accepted) as reported in Table 5.8 (Greenland et al., 2016; Mashele and Chuchu, 2018). The results supported the significant influence of 'Content' (β = 0.204, t-value = 2.445) and 'Credibility' (β = 0.241, t-value = 3.164) on 'Attitude Towards Brand', supporting hypotheses HA3a and HA5a. Also, 'Brand Attachment' (β = 0.265, t-value = 4.258), 'Communication' (β = 0.123, t-value = 2.343), 'Content' (β = 0.126, t-value = 2.302) and 'Attitude Toward Brand' ($\beta = 0.152$, t-value = 2.888) has significant influence on Brand Engagement, supporting hypotheses H_{A1b}, H_{A2b}, H_{A3b}, H_{A4b}, H_{A6b} and H_{A7b}. 'Attitude Toward Brand' ($\beta = 0.251$, t-value = 4.357), 'Brand Engagement' ($\beta = 0.145$, t-value = 1.988), 'Brand Attachment' ($\beta = 0.345$, t-value = 4.699) and 'Critical Mass' ($\beta = 0.13$, t-value = 2.226) also show significant influence on Purchase Intention, supporting hypotheses H_{A7b}, H_{A8}, H_{A1c} and H_{A6c}.

Hypotheses	Relation	Estimate	S.E.	C.R.	P-Values	Remarks
H _{A1a}	$BA \rightarrow ATB$	0.084	0.05	1.279	0.201	Not Supported
H _{A2a}	$COMM \rightarrow ATB$	0.011	0.055	0.206	0.837	Not Supported
H _{A3a}	$C \rightarrow ATB$	0.204	0.075	2.445	0.014	Supported
H _{A4a}	$B \rightarrow ATB$	0.115	0.067	1.461	0.144	Not Supported
HA5a	$CRED \rightarrow ATB$	0.241	0.073	3.164	0.002	Supported
HA6a	$CM \rightarrow ATB$	0.08	0.056	1.357	0.175	Not Supported
H _{A1b}	$BA \rightarrow BE$	0.265	0.054	4.258	***	Supported
H _{A2b}	$\text{COMM} \rightarrow \text{BE}$	0.123	0.059	2.343	0.019	Supported
H _{A3b}	$C \rightarrow BE$	0.167	0.08	2.159	0.031	Supported
H _{A4b}	$B \rightarrow BE$	0.152	0.071	2.093	0.036	Supported
H _{A5b}	$CRED \rightarrow BE$	0.08	0.077	1.143	0.253	Not Supported
H _{A6b}	$CM \rightarrow BE$	0.126	0.06	2.302	0.021	Supported
H _{A7a}	$ATB \rightarrow BE$	0.152	0.061	2.888	0.004	Supported
H _{A7b}	$ATB \rightarrow PI$	0.251	0.054	4.357	***	Supported
H _{A8}	$BE \rightarrow PI$	0.145	0.08	1.988	0.048	Supported
H _{A1c}	$BA \rightarrow PI$	0.345	0.052	4.699	***	Supported
H _{A6c}	$CM \rightarrow PI$	0.13	0.052	2.226	0.026	Supported

 Table 5.8: Structural results

Note: P-Values Significant at p<0.001(***), p<0.01 and p<0.05. (Poorrezaei, 2016)

The results showed that the hypothesised relationships H_{A1b} , H_{A7b} and H_{A1c} were supported at a significance level of 0.001 (t-values > 3.291) (Poorrezaei, 2016). Relationships hypothesised by H_{A5a} and H_{A7a} were supported at a significance level of 0.01 (t-values > 2.576) (Poorrezaei, 2016). Also, H_{A3a} , H_{A2b} , H_{A3b} , H_{A4b} , H_{A6b} , H_{A8} , and H_{A6c} were supported at a significance level of 0.05 (t- values > 1.96) (Poorrezaei, 2016). Figure 5.22 shows the final model after removing the non-significant relationship from the proposed model.





The explanatory power of dependent variables was measured by the value of R^2 , which estimates the percentage of variance within a dependent construct explained by its predicator constructs. Therefore, 'Attitude Toward Brand' (R^2 = 0.344) suggests that 34 percent of the variance in 'Attitude Towards Brand' is being explained by 'Content' and 'Credibility'. Similarly, 'Brand Engagement' (R^2 = 0.615) suggests that 62 percent of the variance in BE is being explained by 'Communication', 'Brand Attachment', 'Content', 'Perceived Benefits', 'Critical Mass', and 'Attitude Toward Brand'. Also, 'Purchase Intention' (R^2 = 0.490) suggests that 49 percent of the variance in Purchase Intention is being explained by 'Attitude Toward Brand', 'Brand Engagement', 'Brand Attachment' and 'Critical Mass' in the context of the study.

5.8 Group differences: Gender, Age, Annual Income, Weekly Internet Usage

Section 5.7 discussed the results of the tested hypotheses, developed and proposed in Chapter 4 Section 4.2. This section advances further to test the effects of the moderating variables Gender, Age, Annual Income, Weekly Internet Usage on the path relationships discussed in section 5.7 Table 5.8. The multi-group capabilities of the IBM SPSS AMOS® for employing SEM technique is used to see the effects of distinct groups for demographic variables Gender, Age, Annual Income, Weekly Internet Usage. For this purpose, the data was divided into smaller subgroups, namely - Gender, Age sub-groups, Annual Income sub-groups and Weekly Internet Usage sub-group. Once these subgroups were made in AMOS® structural model was run for each sub-group within a variable. The pairwise comparison of path coefficients across sub-groups, considering the critical ratios for the differences among the groups for establishing reliability and validity. Tables 5.9 to 5.12 represent the results of the multi-group analysis to identify group differences within the proposed model for the sub-groups of the demographic variables.

5.8.1 Gender

The sample data was segregated into two groups, one group consisted of females, and the other group consisted of the male within the sample. The results showed that out of the 17 relationships, only one showed a significant difference in males and females. That, there is a significant difference between Brand Attachment with Brand Engagement ($z = 2.18^{**}$, p = 0.01) suggesting group differences for different gender groups. Table 5.9 shows the estimates and p-values for structural models for both gender groups, as well as the z-scores.

Gender	Fema	le	Male	9		
Genuer	Estimate	Р	Estimate	Р	z-score	Remarks
$BA \rightarrow ATB$	0.068	0.318	0.056	0.448	-0.114	Not Supported
$COMM \rightarrow ATB$	0.02	0.817	-0.001	0.989	-0.188	Not Supported
$C \rightarrow ATB$	0.245	0.036	0.175	0.076	-0.462	Not Supported
$B \rightarrow ATB$	0.091	0.232	0.063	0.569	-0.207	Not Supported
$CRED \rightarrow ATB$	0.182	0.048	0.281	0.021	0.648	Not Supported
$CM \rightarrow ATB$	0.069	0.507	0.079	0.305	0.074	Not Supported
$BA \rightarrow BE$	0.09	0.223	0.327	***	2.18**	Supported
$\text{COMM} \rightarrow \text{BE}$	0.228	0.025	0.106	0.145	-0.975	Not Supported
$C \rightarrow BE$	0.163	0.207	0.171	0.09	0.052	Not Supported
$B \rightarrow BE$	0.183	0.032	0.124	0.273	-0.421	Not Supported
$CRED \rightarrow BE$	0.031	0.753	0.126	0.305	0.598	Not Supported
$CM \rightarrow BE$	0.165	0.156	0.153	0.055	-0.082	Not Supported
$ATB \rightarrow BE$	0.075	0.461	0.245	***	1.364	Not Supported
$ATB \rightarrow PI$	0.268	0.003	0.199	0.002	-0.614	Not Supported
$BE \rightarrow PI$	0.087	0.377	0.145	0.056	0.468	Not Supported
$BA \rightarrow PI$	0.263	***	0.222	0.002	-0.399	Not Supported
$CM \rightarrow PI$	0.154	0.127	0.106	0.097	-0.4	Not Supported

Table 5.9: Group differences within Gender

Notes: z-score: *p<0.10; **p<0.05; ***p<0.01 (Tarhini et al. 2014)

5.8.2 Age

The sample data was divided into two age groups, one being respondent with age less than 30 years and other belonging to an age greater than 30 years. The results show that the age moderates two of the relationships out of the seventeen, i.e., there is a significant difference between the relations existing between Brand Attachment with Brand Engagement ($z = -2.446^{**}$, p = 0.05) and Critical Mass with Purchase Intention (z = -2.208, p = 0.05) suggesting

group differences for different age groups. Table 5.10 shows the estimates and p-values for structural models for both the Age groups, as well as the z-scores.

A G 0	Below	30	Above	30		
Age	Estimate	Р	Estimate	Р	z-score	Remarks
$BA \rightarrow ATB$	0.021	0.746	0.139	0.062	1.198	Not Supported
$COMM \rightarrow ATB$	-0.034	0.635	0.079	0.378	0.985	Not Supported
$C \rightarrow ATB$	0.161	0.06	0.303	0.083	0.728	Not Supported
$B \rightarrow ATB$	0.069	0.382	0.059	0.708	-0.058	Not Supported
$CRED \rightarrow ATB$	0.273	0.006	0.186	0.089	-0.584	Not Supported
$CM \rightarrow ATB$	0.073	0.33	-0.001	0.987	-0.703	Not Supported
$BA \rightarrow BE$	0.332	***	0.06	0.489	-2.446**	Supported
$\text{COMM} \rightarrow \text{BE}$	0.131	0.07	0.128	0.222	-0.021	Not Supported
$C \rightarrow BE$	0.179	0.036	0.117	0.566	-0.279	Not Supported
$B \rightarrow BE$	0.139	0.08	0.216	0.239	0.386	Not Supported
$CRED \rightarrow BE$	0.056	0.56	0.082	0.517	0.165	Not Supported
$CM \rightarrow BE$	0.138	0.066	0.182	0.047	0.369	Not Supported
$ATB \rightarrow BE$	0.168	0.018	0.246	0.028	0.586	Not Supported
$ATB \rightarrow PI$	0.174	0.008	0.324	***	1.329	Not Supported
$BE \rightarrow PI$	0.15	0.096	0.101	0.228	-0.399	Not Supported
$BA \rightarrow PI$	0.166	0.019	0.317	***	1.451	Not Supported
$CM \rightarrow PI$	0.201	0.005	-0.019	0.782	-2.208**	Supported

Table 5.10: Group differences within Age

Notes: z-score: *p<0.10; **p<0.05; ***p<0.01 (Tarhini et al. 2014)

5.8.3 Annual family income

After evaluating the demographic data collected from the survey, the respondents were divided into two groups based on annual family income. One group comprised of respondents that have a family income less than INR 7,50,000 per annum, and the other where the family income was higher than INR 7,50,000 per annum. The results of the analysis indicated that the relations existing between Brand Attachment with Attitude Towards Brand ($z = -2.225^{**}$, p = 0.05), Perceived Benefits with Attitude Towards Brand ($z = -2.51^{**}$, p = 0.05), Critical Mass with Attitude Towards Brand ($z = 1.789^{*}$, p = 0.1), Communication with Brand Engagement (z =2.563, p = 0.05) and Attitude Towards Brand with Purchase Intention ($z = -1.657^{*}$, p = 0.1) has a significant difference suggesting group differences for different annual income groups. Table 5.11 shows the estimates and p-values for structural models for both annual income groups, as well as the z-scores.

Annual Income	Below 7.5	LPA	Above 7.5	5 LPA		
Annual Income	Estimate	Р	Estimate	Р	z-score	Remarks
$BA \rightarrow ATB$	0.176	0.007	-0.049	0.518	-2.255**	Supported
$COMM \rightarrow ATB$	-0.075	0.296	0.095	0.237	1.577	Not Supported
$C \rightarrow ATB$	0.107	0.206	0.295	0.022	1.219	Not Supported
$B \rightarrow ATB$	0.249	0.01	-0.071	0.392	-2.51**	Supported
$CRED \rightarrow ATB$	0.283	0.002	0.188	0.103	-0.649	Not Supported
$CM \rightarrow ATB$	-0.017	0.79	0.192	0.048	1.789*	Supported
$BA \rightarrow BE$	0.226	0.002	0.241	0.003	0.136	Not Supported
$COMM \rightarrow BE$	-0.009	0.909	0.299	0.001	2.563**	Supported
$C \rightarrow BE$	0.13	0.166	0.263	0.055	0.803	Not Supported
$B \rightarrow BE$	0.24	0.027	0.036	0.672	-1.476	Not Supported
$CRED \rightarrow BE$	0.022	0.83	0.118	0.322	0.616	Not Supported
$CM \rightarrow BE$	0.235	0.003	0.07	0.473	-1.312	Not Supported
$ATB \rightarrow BE$	0.233	0.017	0.103	0.216	-1.013	Not Supported
$ATB \rightarrow PI$	0.336	***	0.148	0.024	-1.657*	Supported
$BE \rightarrow PI$	0.128	0.217	0.129	0.076	0.007	Not Supported
$BA \rightarrow PI$	0.234	0.003	0.232	***	-0.019	Not Supported
$CM \rightarrow PI$	0.121	0.089	0.132	0.101	0.108	Not Supported

Table 5.11: Group differences within Annual Income

Notes: z-score: *p<0.10; **p<0.05; ***p<0.01 (Tarhini et al. 2014)

5.8.4 Weekly internet usage

The sample was segregated in two different groups. One where the respondents use the Internet for less than ten hours a week and the other, where the respondents use the Internet for more than ten hours a week. The results showed that there is a significant difference between the relations existing between Brand Attachment with Attitude Towards Brand ($z = -2.306^{**}$, p =0.05), Communication with Attitude Towards Brand ($z = -2.194^{**}$, p = 0.05), Content with Attitude Towards Brand ($z = 2.383^{**}$, p = 0.05) and Perceived Benefits with Attitude Towards Brand ($z = -1.821^{*}$, p = 0.1) suggesting group differences for different users in terms of Internet usage per week for users. Table 5.12 shows the estimates and p-values for structural models for both Weekly Internet usage groups, as well as the z-scores.

Weekly Internet	<10 hr/v	veek	>10 hr/v	veek		
Usage	Estimate	Р	Estimate	Р	z-score	Remarks
$BA \rightarrow ATB$	0.222	0.005	-0.011	0.857	-2.306**	Supported
$\text{COMM} \rightarrow \text{ATB}$	0.221	0.035	-0.055	0.427	-2.194**	Supported
$C \rightarrow ATB$	-0.085	0.475	0.28	0.004	2.383**	Supported
$B \rightarrow ATB$	0.321	0.022	0.03	0.689	-1.821*	Supported
$CRED \rightarrow ATB$	0.181	0.256	0.211	0.01	0.165	Not Supported
$CM \rightarrow ATB$	-0.015	0.814	0.156	0.062	1.628	Not Supported
$BA \rightarrow BE$	0.126	0.143	0.281	***	1.393	Not Supported
$COMM \rightarrow BE$	0.113	0.303	0.111	0.14	-0.018	Not Supported
$C \rightarrow BE$	0.349	0.007	0.081	0.437	-1.612	Not Supported
$B \rightarrow BE$	0.005	0.974	0.191	0.022	1.096	Not Supported
$CRED \rightarrow BE$	0.147	0.379	0.089	0.313	-0.31	Not Supported
$CM \rightarrow BE$	0.114	0.104	0.138	0.122	0.212	Not Supported
$ATB \rightarrow BE$	0.231	0.037	0.186	0.017	-0.329	Not Supported
$ATB \rightarrow PI$	0.306	0.001	0.195	0.003	-0.95	Not Supported
$BE \rightarrow PI$	0.122	0.216	0.114	0.134	-0.069	Not Supported
$BA \rightarrow PI$	0.198	0.012	0.268	***	0.67	Not Supported
$CM \rightarrow PI$	0.095	0.137	0.113	0.13	0.186	Not Supported

Table 5.12: Group differences within Weekly Internet Use

Notes: z-score: *p<0.10; **p<0.05; ***p<0.01 (Tarhini et al. 2014)

5.9 Conclusion

This chapter has presented the data analysis of the collected data in two phases. In the first phase, a pilot study was conducted based on the questionnaire developed for the study. The data collected for the same was used to find the reliability of the questionnaire through Cronbach alpha values. Once established that the questionnaire has high reliability, the questionnaire was circulated for the data collection for the second phase. The collected dataset was tested for the assumptions of multivariate analysis (missing data, outliers, normality, linearity, homoscedasticity and multicollinearity), which were found to be true. Confirmatory Factor Analysis was implemented on the dataset to test the discriminant and convergent validities. The model was tested for its fitness through measurement model using six indices ($\chi 2$ /df, Comparative Fit Index, Goodness-of-Fit Index, Standardized Root Mean Square Residual, Root Mean Square Error of Approximation and p-Value of close fit). These indices

were found to be under acceptable values showing the fitness of the model. The structural model tests the hypotheses developed in Chapter 4, Section 4.4 and its results analysed and interpreted. Groups differences for different demographical variables for the same model are also tested and interpreted. The next chapter presents a case on Patanjali, an emerging FMCG brand in India. The Case attempts to corroborate the results of the hypotheses tested in this chapter.

Chapter - 6

A Case Study on Patanjali Ayurved Limited

6.1 Introduction

FMCG Sector in India is one of the largest sectors to cater to consumers. There is a massive competition in the market due to similar products in the same categories by the major brands and well as by the local marketers. It becomes essential for the brands and marketers to continually keep the consumer engaged through advertisements and other marketing and promotional channels. Brands are using traditional channels for the same, which includes both print and media.

Last decade has seen a change in the digital innovation in the context of internet and mobile technologies, due to which web platforms have become quite popular globally. Social Media has become a phenomenon which is used by almost everyone. Brands and marketers have turned to Social Media platforms for engaging the consumers and use their capabilities for marketing and promotional purposes. The Indian FMCG brands have also made an effort to create their presence on the popular Social Media platforms.

In recent years, Patanjali Ayurved Limited has emerged as another important brand in the Indian FMCG market. It started as an Indian brand under the flagship of Patanjali and Baba Ramdev and has given fierce competition to some of the major brands under certain herbal product category. Patanjali; though a late starter in utilizing Social Media platforms for marketing and connecting with consumers have become quite active digitally. This study aims to explore the online consumer engagements undertaken by Patanjali on digital platforms, with special focus on its official Facebook Brand Page. This chapter attempts to fulfill the first and second objectives of the study, as discussed in chapter 1, section 1.6.

6.2 Patanjali Ayurved Limited – an overview

Yoga Guru Baba Ramdev and Acharya Balkrishna are the founders of the Patanjali Ayurved Limited (PAL) as an Indian consumer goods company in the year 2006. The conception of this organisation was from the perspective of providing a quality product to create a better quality of life for the people of India. PAL produces healthcare products, personal care products, and ayurvedic products. (Bloomberg, 2018).

Product	Brands of Patanjali	Some of the Competitive Brands		
Sharbat/ Squash	Amla, Bel, Brahmi, Gulab, Keshar	Kissan Squash, Rooh-afza, Minute		
	Badam, Orange, Nimbu, Mango	Maid, Rasna		
Juice	AloeVera, Anar, Amla, orange	Real, Priyagold, Godrej and Kissan		
		Juices		
Jam	Apple, Pineapple and Mix Fruit Jam	Kissan, SIL, Tops, Druk		
Salt	Saindha Namak	Tata, Annapurna		
Chyawanaprash	Sada, Special Chyawanaprash	Dabur, Baidyanath, Zandu kesari		
		Jiwan		
Flour	Arogya Aata	Aashirwad, Annapurna		
Candy	Amla, Anardana	Dabur Hazmola		
Washing Powder	Ujjwal	Arial, Surf-excel, Tide		

Table 6.1: Patanjali	vs.	Other	Brands
----------------------	-----	-------	--------

With the current production capacity of INR 510 crores (US\$5.1 billion) PAL is all set to expand to a capacity of INR 870 crores (US\$8.7 billion) by opening new manufacturing units at several places within India. PAL provides the farmers with information and technical aid for cultivation and efficient farming. All the consumables manufactured by PAL are available to the consumers only through authorized Patanjali Stores.

The industrial area in Haridwar acts as the PAL's main Manufacturing units and headquarters while Delhi marks the location for its registered office (Aradhak, 2018). PAL imports a large number of herbs from Nepal under the trademark Nepal Gram Udyog and has set up manufacturing units for production in the neighbouring country too (Simplus Information Services, 2015). There are further plans to establish more manufacturing units in India as well as in Nepal (Webdunia, 2015). Patanjali is being accorded as the fastest growing Indian FMCG company as per CLSA and HSBC. With an annual turnover of INR 10,216 crore (US\$ 1.4 billion) for FY 2017, Patanjali is valued at INR 30 crore (US\$ 440 million) (Dutta, 2017).



Figure 6.1: Sales and Profit Trends of Patanjali Ayurveda Limited Source: (Patanjali.org, 2018)

Recently Patanjali was placed seventh in the list of Most Influential Brand (MIB), where Microsoft and Apple were placed at 8th and 9th place (FinancialExpress, 2018b).

After Patanjali's rose to Rs 10,500-crore organisation within a decade, the MNC's have shifted their focus on the Ayurveda sector (Bhushan and Chakravarty, 2018). Patanjali is aiming to outclass the rivals in the FMCG domestic market. PAL has made total revenue of 10,561 crores in the year 2016-17, making it the second-largest FMCG player after Hindustan Unilever. What

is even more impressive is that Patanjali's revenue puts the shade on the combined sales of India's MNC giants. Ghee is the bestseller for Patanjali, contributing 14% to total revenues, followed by toothpaste (9%) and hair oil (8%). Its Divya pharmacy store contributed 8% Sales (Somvanshi, 2017). Patanjali's hair oil is recorded to be higher than that of Bajaj Corp, with revenue at Rs 825 crore. Most of Patanjali's revenues are generated by its toothpaste sales at Rs 940 crore, a segment where the home-bred player has captured 14% market share.

6.3 Promotional strategies

The philosophy behind Patanjali is providing the people of India with products that are made in India. Baba Ramdev as a spokesperson for PAL has focused on the demerits of the products sold by the MNC's in India and has emphasised on purchasing "swadeshi" (made within the country). Baba Ramdev has continuously pitched Patanjali as a "Swadeshi" alternative to all the MNC products which help source Indian money abroad to "Videshi" (foreign) FMCG companies like HUL, P&G and Nestle. Patanjali is one of the important strategies to connect on an emotional level with the consumer and then drive them towards their brand.

PAL's association with yoga guru (yoga teacher) Baba Ramdev was the major advantage regarding branding itself. This helped in establishing Patanjali as a brand. Baba Ramdev popularized majority of Patanjali's products while teaching yoga in his numerous yoga camps that are still held both on national and international levels. During these camps, Baba Ramdev used to directly address the masses and talk about a specific Patanjali product. The direct communication established by Baba Ramdev promoted the brand on an international level.

The focus of Patanjali as a brand was more on content marketing where they attracted the consumers to their product and never bombarded them with ads. The personalized approach by Baba Ramdev on yoga camps by directly engaging with at the consumers. He never harped about how good Patanjali's products are. Instead, he focused more on the evil of the other

similar products from various MNC's, and the benefits of the Indian made products. He concentrates on promoting the reasons to the consumer and leave them on their own to explore the wide variety of the products that Patanjali offers.

Majority of the promotion and marketing was done by word of mouth communication by the followers of Baba Ramdev. They suggested the products that they used to their social masses. The critical mass effect then took place, leading to sales of Patanjali products. To make Patanjali a brand, traditional marketing strategies were also followed. Patanjali has a continuous presence on religious television channels like, 'Astha' and 'Sanskar' along with various News channels. They have a presence on print media as well, with advertisements in newspapers and billboards at their local outlets being the most frequently followed marketing practices by Patanjali. In 2016, during the last quarter for the year, Patanjali was the third most advertised brand on television (CIIM, 2016).



Figure 6.2: PAL's corporate web portal

Alongside traditional advertising, PAL also incorporated Digital marketing into their marketing and promotional strategy. Currently, Patanjali has two online websites; www.patanjaliayurved.org. and www.patanajaliayurved.net, as shown in Figures 6.2 and 6.3.

www.patanjaliayurved.org is the corporate website for PAL and showcases the organisation's overview along with its vision and mission. The corporate portal act as a medium to invite dealerships from retailers that are interested in selling Patanjali products. Also, this web portal has a link to patanajaliayurved.net as well as lists of megastores, Patanjali Chikitsalaya (medical center) and Patanjali Arogyakendra (medical advisory center) located in all the states in India. The portal also contains a download section where forms for different dealerships are available which can be filled and submitted online. The portal is easy to navigate and uses high-quality image and simple content for everyone to read.



Figure 6.3: PAL's E-Commerce portal

Even though Patanjali products were exclusively available at physical stores, PAL decided to introduce an online e-commerce portal where consumers can browse through various Patanjali

products and order online, too (Figure 6.3). The portal is easy to navigate, and the access to multiple tabs and information about the products is easy to access. Having an e-commerce portal gives them access to people who shop prominently online as well as those who do not have time to identify a local Patanjali store and physically visit and buy any Patanjali product. The website also has a section by the name Knowledgebase which is similar to a blog and contains useful information about health care, personal wellbeing,' Ayurveda' (herbal science) and tips on health issues. The Content of the blog is informational and provides knowledge in 4 fronts, as seen in Figure 6.4. PAL has its entire range of products showcased on this e-commerce portal with prices in INR. It is observed that none of the bigger FMCG players in the same market sell their product on such online despite their comments on media on how they are integrating Digital media into their business strategy.



Figure 6.4: The blog section of the E-Commerce portal

6.4 Social Media strategy for Online Consumer Engagement

Apart from these two online portals, Patanjali also has a digital presence in all major Social Media platforms. In February 2017, Patanjali started its advertising campaigns through digital media and begun collaborating with Social Media and tech giants, Facebook, and Google (ABPLive, 2017). However, this was not the picture in 2014, as PAL was only focusing on the traditional means of marketing and promotion. Even though its competitors were using Social Media platforms for engaging and connecting with its consumers, PAL was quite reluctant about trusting the potential of Social Media to attract the masses. Post 2014 it actively started using different Social Media platforms which are considered essential for any Brand/marketer. Different Digital and Social Media platforms where Patanjali is actively engaging the consumers are as follows:

S. No.	Digital Media Platform	Title	No. of Followers/Likes/Reviews*
1	Facebook	Patanjali Products - पतंजलि उत्पाद	760,883
2	YouTube	Patanjali Ayurved	329,342
3	Instagram	Patanjali Ayurved @PypAyurved	276,710
4	Twitter	patanjaliproducts	11,167

Table 6.2: Patanjali Ayurved Limited Digital and Social Media Presence

*as on 20 February 2019

As many brands do, Patanjali too was able to understand the value of the various Social Media platforms, as efficient channels to promote and connect with the consumers. They have invested in these platforms in recent years and integrated platforms into their marketing and promoting strategies.

6.4.1 Patanjali on YouTube

Youtube is one of the prominent Social Media platforms where PAL is quite active. Youtube is a streaming service where users can make their channels for uploading self-made videos for other users to view and engage with. The engagements on the video uploads are in the form of like, share and comments. The user follows a channel which they like. The success of a YouTube channel is majorly assessed by the number of followers it has. PAL made its debut on the Social Media platform on July 2014. Figure 6.5 shows the Interface of the Patanjali's YouTube channel, Patanjali Ayurved, which has 329,342 followers (till 20 February 2019). It also has 351, 037, 341 total views on the channels. One can see that most of the video seen on the channel page are of the products that PAL sell. The link to the main video that is showcased on the channel is shown in Figure 6.5. This video is constantly updated, and each video shows Baba Ramdev showcasing a new product and discussing its benefits to the audience. In Figure 6.5, Baba Ramdev is showcasing Patanjali's Dant Kanti Gel Toothpaste.



Figure 6.5: Patanjali Ayurved, PAL's YoutTube Channel

PAL has been using YouTube as a platform to engage people with good video content on its channel. Majority of the channel content has been put under various playlists to make it easy

for the users on YouTube to access is shown in Figure 6.6. The video content primarily holds television commercials and Patanjali Advertisements. There are playlists which hold the informational details about Patanjali products. In most of these videos, Baba Ramdev is featured talking about the details of the products and the benefits for which they should be

consumed by the users. Also, there are several playlists (as can be seen from Figure 6.6) which contain instructional videos featuring Baba Ramdev and Acharya Balkrishna on the topics of yoga and health respectively. The channel viewership is not that strong, but the videos that were uploaded in January 2019 have garnered 220 to 6,208 views (till 20 February 2019). For an organisation that was mostly focussing on the traditional channels to promote and market themselves, PAL is quite active on YouTube; it frequently uploads on its channel with relevant and meaningful content to engage the consumers.



Figure 6.6: Various Playlists on PAL's YouTube Channel

This channel was initially used to upload the videos about various yoga-asanas (yoga postures) featuring Baba Ramdev and Acharya Balkrishna, which were also released on VCDs and DVDs in the year 2014. Also, a few of the videos featured the launching of some of the prime products. However, in recent years, Patanjali has become more vigilant in using the capabilities

and engaging the consumers on YouTube. Later, as PAL became more active on digital platforms, new dimensions in the form of videos showcasing specific content were uploaded.

Baba Ramdev is known to showcase Patanjali products in any public event, i.e., 'yog shivirs' (yoga camps) in various parts of India. Recordings of these events are done and from the recorded footage, the instances when Baba Ramdev is promoting a Patanjali Product are edited out. These short videos have a duration varying from 15 seconds to 2 minutes and have informational content about the product. Along with these personalised promotional contents, television advertisements are also showcased on the channels. These videos are linked to various other Social Media platforms, such as Facebook, Twitter and Instagram. The footfall on the majority of the videos discussed above ranges from 2,000 to 2,60,000 views.

The about section of the Channel gives a description of PAL, as shown in Figure 6.7. As can be seen from Figure 6.7, PAL has given backlinks to its e-commerce portal as well as other Social Media platforms. Giving backlinks helps in directing the traffic from YouTube channel to the respective Social Media (Facebook, Twitter and Instagram) and other digital pages owned and managed by PAL (www.patanjali.org and www.patanjali.net).

= ** ***	lant.	181	• = • • •
A	Patanjal Apared a International and west function constant constant	-	
 image <	Description	illen onen interne matterielen p	En efficience en
ang nan-man ang n	Lee Constant Const Constant Const Constant Const		

Figure 6.7: About page of Patanjali Ayurved YouTube Channel

6.4.2 Patanjali on Instagram

Instagram is another popular Social Media platform which is garnering quite a lot of following of late. It is a photo and video sharing online service owned by Facebook Inc., which allows the user to upload photos and videos which can be edited using digital filters. The accounts made by the user can be shared publicly and people can follow users. The content can be tagged for another user, along with the location. Users can also view the trending content on both the mobile application and website of the service. In recent years, Instagram has become quite popular among the masses and has become a platform which cannot be ignored when it comes to strategizing the digital engagements on Social Media platforms. PAL too, has been using the Social Media platform to attract and engage more consumers.

Figure 6.8 shows the main account page of the official Instagram page of PAL on Instagram. As seen in the figure, the info section of the page displays a toll-free number for any queries. It also displays the web address of the e-commerce portal of PAL for directing the users to the web page. PAL is utilising its official page for promoting its products to the users that are more active on this photo and video sharing platform.



Figure 6.8: patanjaliproducts, Official Instagram page of PAL
The posts are similar to what they share on their official Facebook brand page. The official page on Instagram has garnered 272,860 followers. So far, patanjaliproducts has posted a total of 1989 posts which comprises of both images and videos (till February 20th, 2019).

PAL uses both images of the products as well as the video as content for the posts. These images are mostly of products sold by Patanjali on its e-commerce portal and its retail shops. Instagram has 75 million Indian users as on January 2019 (Statista, 2019a). Tapping such a huge potential user demographic is an opportunity which PAL is making use of by actively participating and engaging the consumer with its regular posts. Each post has hashtags which are referring to the name of products that are being showcased on the image or video along with #patanjali as the common hashtag in all the posts. Each post has received more than 1000 likes suggesting that users are engaging with the PAL's official page on Instagram. All the content that is shared in the form of posts on Instagram has relevant information about the products that are being sold by PAL. The images are same that are shared on the Facebook brand page of PAL making the content similar on all the platforms to create a similar visual content for the users to engage with.

6.4.3 Patanjali on Twitter

Twitter is another social networking platform which is quite popular among Internet users. It is an online messaging service which allows people to send out messages called "tweets". These tweets are visible to anyone accessing the user's profile and those who are following the user. The message should have a maximum of 280 characters to be eligible for sending out as a tweet. In other words, Tweeting is a form of micro-blogging. There are 3.44 crore (34.4. million) Twitter users in India, which makes it one of the essential platforms for PAL to promote and showcase its products (Statista, 2019b).

PAL has an official Twitter profile by the name 'Patanjali Products' with its handle as @PypProducts. Twitter works more on word of mouth and the informativeness of the tweet.



Figure 6.9: Patanjali Products, the official Twitter profile of PAL

Also, how much the tweet entertains the users on Twitter also makes a difference in making the user engage with the brand page. Figure 6.9 shows the official Twitter profile page of Patanjali Products. Majority of the tweets that are tweeted by PAL are about their products with relevant hashtags which contain the products name and #Patanjali as the common hashtags for most of the tweets. The tweets majorly contain an external link about which PAL is tweeting. Mostly these links are directing the user to the Facebook Brand Page of PAL where there is no restriction on the number of characters, unlike twitter. Most of the tweets are short and entices the reader to click on the link given in the tweet to know more. The links are mostly directed to the Facebook brand page moderated by PAL. Since Twitter is more about microblogging, it is used by PAL to direct its traffic to Facebook Brand Page; therefore, making the users engage with more content and information on the brand page. Tweets have fewer likes, but most of the tweets are shared by users/followers.

6.5 PAL's Facebook Brand Page (FBP) engagements

PAL has been quite active on its digital media promoting through Social Media platform as discussed in previous sections. Facebook is one Social Media platform that has been used for promotional purposes by PAL. Approximately 3 crores (300 million) Internet users in India are active on Facebook as of January 2019 (Statista, 2019c). As discussed above, PAL uses other Social Media platforms to provide a backlink to its FBP "Patanjali Products - पतंजलि उत्पाद". It directs the user traffic from these platforms to FBP, as it does the most of its promotional activity on Facebook. The FBP is moderated well and it frequently posts updates to engage consumers. PAL is quite active on its FBP and it updates new posts frequently.

Patanjali's one of the important strategies is, to connect on an emotional level with the consumer and then drive them towards their brand. Facebook is the most-liked and accessed Social Media platform Patanjali has used over the years. With more than 7 Lakh (7 million) followers, this platform has brought in the most engagement. Since its introduction of the Social platform, Patanjali has been well received by the youth in three Indian states, viz, Karnataka, Orissa, and Punjab (states in India) that collated to 80 percent of the response (ABPLive, 2017).



Figure 6.10: PAL's official Facebook Brand Page (FBP)

PAL has been using Facebook and its capabilities for various advertising and promotional activities which are reported as follows:

i. Linking the e-commerce portal with Facebook Brand Page - Patanjali uses the Facebook page to promotes its E-commerce portal patanajaliayurved.net; Thus, directing the consumer to its products. Brands must provide a backlink to their website/corporate



Figure 6.11: Link to the e-Commerce portal in the About Section of the FBP

portal. Along with the link to the e-commerce portal, Patanjali also provides its contact details and an overview of the organisation in the "About" Section of the brad page.



Figure 6.12: PAL's organisational profile and all the digital portals

ii. Content with a clear message - Patanjali uses vivid images while posting and the posts are quite frequent, as to show regularly in the feed section of the followers of Patanjali. Since PAL advocates for "Swadeshi" and is against the big foreign MNCs in the FMCG sector, most of the post focus on the theme of nationalism. The messages within the images are written in Hindi as well as the status about the post is written in Hindi with appropriate hashtags.



Figure 6.11: Facebook posts on Patanjali Facebook Brand Page

Patanjali also utilised the capability of Facebook, where they can show other related products below the post, which leads to the e-commerce portal, patanjaliayurved.net. This increases the footfall on their e-commerce portal as well as the chances of purchase by the Facebook user. Also, on Each image which is posted, there is a link to the e-commerce portal. The content of the images is simple, and it uses soothing colour schemes. The focus is on the theme of nationalism. Posts use a catchy one-liner to appeal to the masses.

iii. Unique campaigns around Independence Day - For the past many years, just before Independence Day (15th of August), Patanjali post images with messages that invoke the feeling of nationalism. Figure 6.14 shows such posts posted by Patanjali. The messages are from a prominent historical person who played a strong role in reforming India. The images

contain a quote from the person on the "Swadeshi" theme, discouraging the use of "videshi" products, as shown in Figure 6.14.

Patanjali created a special hashtag for this campaign, #SwadeshiSankalp which can be used in other Social Media platforms too. These posts contain a similar structure as the product posts regarding the status and the e-commerce portal link in the images. The major difference is the image of a prominent personality from our rich Indian history along with a quote from them in support of the use of "Swadeshi" Indian product against the use of the "Videshi" product. The colour scheme used emulates a tricolour which gives a patriotic feel to the post. These posts are targeted towards the emotions of the consumers and address their patriotic and sense of being an Indian. The campaign garnered mostly positive sentiment form the users.



Figure 6.14: Facebook Campaign posts during the Independence Day week

iv. Patanjali through showcasing social contributions - Facebook is a very influential media, and it can be used to improve or create a positive image for a brand or an organisation. Patanjali too uses Facebook for creating awareness about social work or any such occasion/event/news. E.g., Patanjali, under the flagship of "Swadeshi Abhiyan,"

organises "Patanjali Swawlamban Shivir" (Patanjali Self-independent Camp) which is targeted towards the unemployed youth of our country and provide them job opportunities within the organisational structure of various ventures undertaken by a PAL. Recently such "Shivirs" (camps) were organised by Patanjali in the Indian states of Rohtak, Panchkula, Chandigarh, Delhi, Jaipur and Ahmedabad. Events were covered as one image that shows pictures from the events. The Status message of the post is in Hindi and has a positive message and sentiment. The images are clear and present a successful picture of the "Shivirs". Again, the web link of the Patanjali e-commerce portal is present in the image. Such posts help in creating a positive image of the Organisation as well as creating good word of mouth both electronically and in the real world. This post garnered mostly positive sentiments from the users and was shared by the users as well.



Figure 6.15: FBP posts showcasing the employment drives at various states in North India

v. Promoting other initiatives of Patanjali - Another such event that was posted on the Patanjali brand page was the cleanliness drive conducted after the "Kanwar Yatra" by Patanjali

Yogpeeth, Haridwar. The drive was called "Nagar Swachhata Abhiyaan" (Town Cleanliness Drive), and it was conducted in the Haridwar city. After the "Kanwar Yatra" ended, the local youth along with the volunteers form the Patanjali Yogpeeth came together and joined hands to clean the city. Patanjali posted several images which were taken during the cleanliness drive and showed youth volunteers cleaning the cityscape.

The image above shows, Patanjali promoting the Facebook brand page of "Patanjali Yogpeeth," Creating a brand is easy, but to keep up the brand image is difficult. By Showcasing the social work or any such philanthropic activity taken by the brand can make it more likable by presenting a positive picture of the brand. The cleanliness drive took up by Patanjali as a brand affect the PAL and its brand image. Posts like these strengthen the philosophy behind PAL and garner positive attitude of the consumer towards the brand.



Figure 6.16: FBP posts showcasing the cleanliness drive by Patanjali Yogpeeth at Haridwar

Patanjali has used a branded house strategy where all the products and services come under one umbrella brand, i.e., Patanjali. Therefore, it is a must for them to ensure that Patanjali as a brand should have a positive brand image.

vi. Going live with public events hosted by Patanjali - Facebook has the capability of live streaming. Users can comment on the live stream and a direct conversation or dialogue can be established between the user who is streaming and the ones who are viewing. Patanjali utilises this capability of Facebook and keeps the consumers engaged with the brand. Quite a number of their events have been live-streamed on their Facebook brand page. The live streams are viewed by the users, and they comment throughout the streaming. Users may also share the like on their Facebook feed or tag people, which makes it more accessible for other interested user or those users who might engage out of curiosity. Comments of users on these live streams are generally held positive sentiments.



Figure 6.17: Live streaming of Yog Shivir on Patanjali FBP



Figure 6.18: Live Streaming of opening ceremony of Patanjali Research Centre vii. Promotion through localized Facebook brand pages - Apart from previous strategies, Patanjali also runs localised advertising campaigns that targeted at regional levels. Andhra Pradesh, Kerala, Tamil Nadu, Karnataka, and Telangana were some of the states that were targeted by these campaigns. Also, they have separate Facebook pages for the localised Chikitsalayas (hospitals) which have the same strategy of posts as the "Patanjali Products -पतंजलि उत्पाद". By engaging people at the local level, Patanjali expects to extend its reach outside the northern region of the country. They have fierce competition with the local ayurvedic products in the southern region, which makes it more important to have a reach with the consumers.



Figure 6.19: Patanjali's local FBP for Chennai

viii. Communicating with users through Facebook Chat - PAL uses the chat capabilities of Facebook to connect directly with the consumers. The chat is meant to cover consumers that wish to know more about Patanjali products, who have queries regarding other ventures of Patanjali. Other important stakeholders for which Patanjali is utilizing the Facebook chat are potential retailers. As a Brand, PAL must connect and communicate with the consumers (both Business to business and Business to consumer) which helps in creating



Figure 6.20: Chat application on Facebook Brand Page used by PAL

engagements which might lead to a loyal consumer base. By communicating with the users, Patanjali is creating value from the perspective of the consumer and providing them with an experience where the consumer feels that the brand is listening to them. This communication is important to the consumer as it makes them connect more with the brand.

6.6 Evaluation of Patanjali's Facebook strategy

PAL has been actively using Facebook for promotional purposes. Their digital media activity has increased since the year 2014. Their Social Media strategy using different Social Media platforms, as discussed in previous sections, has garnered quite a lot of attention from the consumers. PAL has used the public stature and following to brand itself. Baba Ramdev is a brand himself when it comes to yoga, and Acharya Bal Krishna being an "Aushadi Vigyani," (herbal scientist) Patanjali was already a brand. Patanjali as a brand was a big help to PAL to

market itself under these two well-established umbrella brands, Baba Ramdev and Patanjali. The population was already felt attached to the above two brands which helped in creating a connection with PAL as a new venture by Patanjali. Majority of the marketing was done by word-of-mouth among the consumers of Patanjali and the masses. Baba Ramdev is instrumental in directly putting various Patanjali products to the people through traditional media and his "Yog Shivir(s)." The same information is disseminated by the very same attendee to his/her peers. The critical mass played its role, and the sales of the products can be attributed majorly to the word-of-mouth among people who use the products and the probable consumer. Masses were already involved and invested in Patanjali due to Baba Ramdev and various "Aushadhalayas" (medical dispensaries) and "Chikitsalayas" (Hospitals) under Patanjali. Brand Attachment was already high in the masses towards Patanjali. PAL rode on the same to establish a connection with the consumers. Majority of the promotion and sales in the initial years can be contributed to the collective attachment of people who are already feeling a sense of attachment with Patanjali as a brand.

The real win for Patanjali came when they invested in digital media campaigns in the year 2014. Initially, Patanjali was not focusing on digital modes of advertising and marketing and was making considerable investments in traditional media. However, in the last few years, Patanjali has realized the power of Social Media. "Patanjali Products - पतंजलि उत्पाद" the official Facebook Brand Page came into existence in the year 2014 as per the date on the Facebook brand page. The content that is posted on the official FBP is unique to the philosophy followed by Patanjali. Content is attractive and vibrant in colours; but, not too colourful to make people dislike the content. The information is short and precise and hits the right sentiment of anyone who is engaging with the brand page. Majority of the information that is shard on FBP is considered to be credible since it always has the brand effect of Baba Ramdev as well as Patanjali. The content focusses majorly on three things – 1) information and benefits of the

brand that is being promoted through the port; 2) Information about other Patanjali ventures, and 3) messages which are having sentiments directed towards "swadeshi". Patanjali also connects with the consumer through the chat capabilities of Facebook to communicate. Patanjali representative generally answers any query about Patanjali within one hour. They are trying to create a connect with the consumer. They can be asked about the nearest Patanjali store or any product sold at the Patanjali stores. This has helped Patanjali in connecting more with its consumers.

The engagement and general attitude of consumers towards PAL and its products are positive. The comments that are posted on the posts on FBP have positive sentiments majorly with a few negative comments as well. People also prefer to engage and like the posts regularly. Some even ask for better usage of the product shown on the post and others reply to the query as well. Overall, the FBP is garnering quite a lot of attention and positive sentiment and attitude of the consumer for Patanjali. Also, PAL has linked the majority of its Social Media platforms to Facebook, where they are most active and where most of the Social Media users have engaged with PAL as a brand.

On the financial front, revenue generated by Patanjali was around INR 2000 Crore (20 Billion) for the years 2014-2015; Which increased to INR 10,216 crore (10.2 Billion) in 2016-2017 (Patanjali, 2018). A small part of rapid increase within a short period can be credited to the Social Media campaigns especially the Facebook brand page that has an active role to play when it came to advertising and marketing PAL in the masses. Patanjali has been running television have been running TV Campaigns since the conception of PAL in 2006. The traditional Media was able to capture the masses, but the reach was not penetrative and specific. After 2014 Patanjali's digital presence has become more prominent. Facebook and YouTube are the primary Social Media platform where maximum consumer engagement regarding subscription/follow/like can be measured. Facebook has the highest number of these measures,

becoming a robust platform for Patanjali to connect with its consumers by engaging them with relevant posts on health and well-being. A recent report suggests that Patanjali is collaborating with both Google and Facebook in coming up with effective digital strategies for Patanjali (BusinessStandard, 2017). According to recent reports, it is forecasted that Patanjali is aiming to cross INR 20,000-25,000 crore (200-250 Billion) by the end of March 2019 (Neogy, 2018). They have started putting serious efforts in digital promotion and may come up with better strategies to capture more consumers in the future. Table 6.4 corroborates the results of the study and the evaluation of PAL and its Facebook Brand Page engagements

Hypotheses	Relation	Findings of the Study
H _{A3a}	$C \rightarrow ATB$	Users have liked and shared the content posted by PAL. Also, the comments posted have positive sentiment, thus showing a positive attitude towards the content and the brand.
H _{A5a}	$CRED \rightarrow ATB$	Users that engage with PAL FBP are mostly engaging due to the credible brand Baba Ramdev and Patanjali. The comments by the users on PAL's posts at times show the same, as users praise Baba Ramdev and Patanjali for their efforts.
Наіь	$BA \rightarrow BE$	The user that follow PAL on Facebook are aware and show attachment to the brand. The increase in followers is supporting the same. They also prefer to engage with the brand and profess their attachments to the brand in the form of engagement elements (likes, share and comments).
Назь	$COMM \rightarrow BE$	PAL uses the chat facility provided by the Facebook platform and it has been noted that They typically reply to a query within an hour, which makes the communication prompt. Users prefer to communicate and connect with the brand they feel that the communication has some value and the brand feels that the consumer is important enough to revert.
Назь	$C \rightarrow BE$	PAL has been using quite engaging content on its bran page, thus making the user engage and connect more with the brand page. The quality, vividness and information of the content makes the user engage with the content and create user-generated content in the form of comments and sharing of the same.
H _{A4b}	$B \rightarrow BE$	PAL showcases the benefits of the products through different media it posts on tits brand page (images and videos), which makes users aware of a product. They also direct the consumer to PAL's eCommerce website where the users are provided with more information regarding other products of PAL, thus increasing the digital engagement.

 Table 6.4: Results of the study and the findings of the Case

Набь	$CM \rightarrow BE$	Word of mouth had an important role in PAL's sales in the initial years. Consumers have been using Patanjali since its conception, based upon he suggestions made by their peers. The same network influences them to join and engage with the brand page. Users actively get involved with the PAL's FBP due to their peer groups following the same.
H _{A7a}	$ATB \rightarrow BE$	The general attitude of the consumers who use PAL products is positive and in favour of PAL. Word of mouth has played and important role in creating a positive image of its products. Users prefer to engage with an FBP if their attitude is positive about the same. This is evident from the number of followers on of the FBP.
Нать	$ATB \rightarrow PI$	PAL has been associated with Patanjali and Baba Ramdev which have created a niche for themselves. They have a huge following and people admire Baba Ramdev as a 'yoga guru' (yoga teacher). As discussed in the case, Baba Ramdev introduces the products during his Yog Camps. The general attitude that people have towards due to the 'Critical Mass' and 'Brand Attachment', it influences their Purchase Intention.
Нав	$BE \rightarrow PI$	Since 2014, PAL made its presence on social media and Facebook is one platform where it has the most engagements with consumers. Since 2014, PAL's revenue has increased exponentially, which, to some extent, can be attributed to the digital platforms where it engages consumers on its official brand pages.
HA1c	$BA \rightarrow PI$	Patanjali and Baba Ramdev are brands in themselves. Masses are already engaged with them and they hold positive sentiments towards these two. Their association with PAL makes the masses more attached to it as a brand associated under a branded house. Their loyalty to Patanjali and Baba Ramdev transcends to PAL as well leading to their intention to purchase
Набе	$CM \rightarrow PI$	The vast majority of the consumers were influenced by people in their peer circle or people on which they rely for decision making in day to day life. As previously discussed, word of mouth has helped PAL in increasing sales. Word of mouth is only effective if an influencer can affect the decision of people in his/her peer group.

6.7 Conclusion

This chapter presented a case on Patanjali Ayurved Limited (PAL) and the online consumer engagement strategies followed by PAL to engage the consumers on various Social Media platforms. This case focuses mainly on the PAL run Facebook Brand Page, which has the maximum number of followers compared to other Social Media platforms used by PAL. The Study explored the strategies followed by PAL on the FBP to engage the consumers on the brand page. This case is the extension of Chapter 3 and it also fulfills the first and second objectives of the study (Chapter 1 Section 1.6). This case also corroborates the results obtained by the empirical study to achieve the third and fourth objective of the study (Chapter 1 Section 1.6). Next chapter presents the discussions on the findings of the study.

Chapter - 7

Findings and Discussions

7.1. Introduction

The recent years have seen a rise in the use of digital media by various FMCG brands, as discussed in Chapter 2. Brands are using multiple Social Media platforms to establish an interactive dialogue with consumers. Social Networking Sites like Facebook, Twitter, Pinterest and YouTube have increasingly influenced the masses, especially the young generation that is more active on these platforms. The Online Brand Communities have become a recent phenomenon driven by the capabilities of these Social Networking Sites. Previous studies have discussed the importance of integrating Online Brand Communities into the marketing strategy. Also, there are benefits that brands/markets can have such as analysing consumer's attitude, their inclinations towards the brand and most importantly, their behavioural intentions. FMCG brands have started to include digital platforms, for marketing and promotional purposes; making it a vital part of marketing, consumer engagement and relations. They have realized the potential, and the possibilities Online Brand Communities on such platform provides in the form of both normative and subjective information. This information can be easily collected from the peer-to-peer interaction among the consumer as well as the interactions that occur between the brand and the consumer and vice versa. As discussed previously, as a Social Networking Site, Facebook has gained wide popularity on a global level since its conception. Due to its media, networking and interactive capabilities, it is one of the premier Social Media platforms used by the FMCG brands. This study is focused on the FMCG run Facebook Brand Pages, and it tries to establish relations between the factors which are essential for engaging consumer/user to the brand page and their intention to purchase the brand in future.

Previous studies have worked on the acceptance of Social Networking Sites and Online Brand Communities as a medium of communication by the consumers. They have employed models like TAM, TAM2, UTAUT to see the intention of adoption of an Online Brand Communities and Facebook Brand Pages. These models encompass significant factors that are pivotal in engagement and thus influence behavioural intentions. Some studies specifically address the engagement in Facebook Brand Page, but they are focused on the specific functionalities of engagement elements (likes, shares, comments, etc.). This study identifies the functional factors that play a vital role to create a stronger influence on consumers, leading to a behavioural intent towards purchasing. It presents the relationships that exist between independent factors.

7.2 Discussion on SEM Results

The study presents the relationships that exist between the independent factors; namely, 'Brand Attachment' (BA), 'Communication' (COMM), 'Content' (C), 'Perceived Benefits' (B), 'Credibility' (CRED) and 'Critical Mass' (CM) with the dependent factors 'Attitude Toward Brand' (ATB) and 'Brand Engagement' (BE). Further, the relationship between 'Attitude Towards Brand' and 'Brand Engagement' as dependent factors on 'Purchase Intention' (PI) have also been explored. As discussed in Chapter 4, section 4.4, following relationships have been studied:

7.2.1 Relationships between Brand Attachment, Communication, Content, Perceived Benefits, Credibility, Critical Mass and Attitude Towards Brand

Majority of the relations proposed in Chapter 4, Section 4.2 were found to be having no significant effect on Attitude Towards Brand. The SEM results showed that Content and Credibility were found to have a significant effect on Attitude Towards Brand when discussed in the context of an FMCG Brand run Facebook Brand Page.

'Credibility' was found to be the most important factor to have a significant effect on 'Attitude Towards Brand' (Chapter 5 Section 5.7). A brand creates its positive image among its consumers from the informational value of brand promotion and advertising. The information which reaches the consumer should hold credibility. If the brand chooses to provide the consumers with useful information about its products, promotions and discounts, consumers always question the credibility of the information from the perspective of truthfulness and quality. Trust is an essential part of credibility and the amount of trust that the brand holds for the consumers; and it majorly affects the consumer's intention to further engage with the brand on Social Networking Sites or otherwise. The informational, truthfulness and trust values of the posts by the brand on a Facebook Brand Page leads to a credible image of the brand. The consumers prefer to engage with a brand that holds values in itself. The credibility of the information, along with the perception of credibility about the brand, extends to creating a better attitude towards the brand.

'Content' was the other important factor that suggested a significant influence on 'Brand Engagement' (Chapter 5 Section 5.7). Content posted on Facebook Brand Pages is essential to be assessed before being published as they represent what the brand wants to communicate with the consumers. The content is mostly in the forms of the status message, images and videos. These contents should be relevant and up to date with the needs of the consumers. Majority of the FMCG brand focuses more on the festivities or any such occasions to promote themselves through these three types of posts. The content needs to be positive, entertaining, vivid in design and style and most importantly catering to consumer's need. These factors affect the consumer's engagement on the Facebook Brand Page as they find the content interesting, entertaining; therefore, leading to further sharing of the said content thus generating a word of mouth promotion on their own. This can only be possible if the brands decide to post quality content, which is rich in information, is vivid and entertaining and makes the consumers feel

good or pleasurable about the brand. A carefully organized content strategy could quickly change the perception of the consumer about the brand or create a new re-imagined attitude towards the brand. Thus, content plays a vital role when it comes to consumers attitude towards the brand.

7.2.2 Relationships between Brand Attachment, Communication, Content, Perceived Benefits, Credibility, Critical Mass and Brand Engagement

The structural results show that 'Brand Attachment', 'Communication', 'Content', 'Perceived Benefits' and 'Critical Mass' have a significant effect on 'Brand Engagement' when discussed in the context of an FMCG Brand run Facebook Brand Page. Brand Attachment being the most important and influential factor followed by 'Content', 'Perceived Benefits', 'Critical Mass' and 'Communication' that effect 'Brand Engagement' significantly.

'Brand Attachment' was found to be the most important factor to effect 'Brand Engagement' significantly (Chapter 5 Section 5.7). Attachment to a brand might lead the consumer to engage more with a brand. When a consumer is attached to a brand, he/she might be willing to advise others to purchase or refer to the brand. Also, they might engage more with the Online Brand Communities or the Facebook Brand Pages set up by the brand. A consumer is more likely to join and be part of a Facebook Brand Page when they are emotionally invested in a brand. This makes them participate within the brand page actively and involve with the peers and brand itself. The higher a consumer's attachment with the brand, the higher is their engagement with the brand on Social Media platforms.

'Content' was found to be the second most important factor to influence 'Brand Engagement' (Chapter 5 Section 5.7). Most of the time, consumers are interested in joining any brand community or brand page to affiliate with the brand or get information about the brand. For any Facebook Brand Page, it is a must to have a quality content which is having a positive influence on the brand. The quality of media, i.e., images, audio and video, play an important role in engaging a user within any Online Brand Community. This content should be entertaining, vivid and with high quality in terms of content value. Consumers are attracted to a brand run Facebook Brand Page or keep up within the Facebook Brand Page only if its posts are informational or have content that is seen as valuable by a consumer. Without Brand Attachment into play, a brand needs to capture the attention of the consumer through the content it posts on the Facebook Brand Page. The quality and the content value help in creating a positive brand image among the Facebook Brand Page member.

'Perceived Benefits' was found to be another important factor that significantly affects Brand Engagement (Chapter 5 Section 5.7). Any brand community based on a Social Media platform is joined/liked/engaged with by consumers if they see some benefits. These benefits can be in the form of information (about the product/service, about sales and discounts coupon, usefulness and benefits of the products, etc. If a brand chooses to post such information in the form of media on a Facebook Brand Page, it tends to increase the consumer's engagement on the brand page. The main reason for any Facebook user to follow and engage with any Facebook Brand Page is to feel connected. Consumers who are loyal to a brand prefer to show their loyalty by associating themselves with any such brand community. Brand reciprocate these loyal consumers by informing them about discounts sale and other such information which is exclusive to the consumers. If a consumer perceives the Facebook Brand Page to be beneficial from the informational as well as the financial point of view, he/she will start engaging/continue to engage with the brand community.

'Critical Mass' also showed significant influence on the 'Brand Engagement' on the Facebook Brand Page (Chapter 5 Section 5.7). As discussed in the previous section, consumers are affected by the peer groups, family and friends when making decisions. They tend to go with the most favourable decisions that the peer would approve. When it comes to following and engaging with a Facebook Brand Page, users prefer to rely on their peers whom they look up to for decision making. Majority of the 'Critical Mas's comprises of the close-knit network including family and close friends. If any user from 'Critical Mass' is engaged with the Facebook Brand Page and share the content on his/her timeline, people tend to engage with the post (like share and comment). Following the Facebook Brand Page also is influenced by who from the users 'Critical Mass' is already following the Facebook Brand Page and dictates the level of engagement by users on/with the Facebook Brand Page.

'Communication' also affect 'Brand Engagement' significantly (Chapter 5 Section 5.7). Communication plays a significant role for brands while managing any Facebook Brand Page (Chapter 5 Section 5.6). Earlier the communication was one way, from brands/marketers to the consumers. Now due to easy accessibility of information through the Internet, the consumer is more aware and needs more from the brand in terms of value. Having an interactive and loyal relationship with consumers is what brands/marketers strive for. Facebook Brand Page allows the brand to establish a dialogue which can be easily monitored by them along with real-time interaction. The capability of these platforms gives the brands a direct line to the consumers, and the response and quality of the interactions create values for the consumers. Consumers that follow any Facebook Brand Page continue to do so since these interactions from the brand create value for them, thus, generating brand loyal consumers. User prefers to engage more with a brand on Facebook Brand Page that communicates regularly and shows cordiality and promptness in the interactions.

7.2.3 Relationships between Attitude Towards Brand and Brand Engagement

The structural results show that 'Attitude Towards Brand' has a significant influence on 'Brand Engagement' (Chapter 5 Section 5.7). When consumers have a positive attitude towards a

brand, or they perceive a brand in a positive sense, they tend to engage more with that brand on a Facebook Brand Page. Thus, suggesting that the attitude; which is influenced by the content posted by the FMCG brand and the credibility of the information of the said posts, is an important factor to consider while making Facebook Brand Page strategy. The consumer needs to have a positive/favourable attitude towards the brand, which can be impressed or improved by the FMCG brand by the use of quality content such as images, video and information. The attitude developed by the posts shared on Facebook Brand Page does affect the engagement of the consumer within the Facebook Brand Page and with the brand as well.

7.2.4 Relationships between Attitude Towards Brand, Brand Engagement, Brand Attachment, Critical Mass and Purchase Intention

The structural results show that 'Attitude Towards Brand', 'Brand Engagement', 'Brand Attachment' and 'Critical Mass' show significant influence on 'Purchase Intention'. 'Brand Attachment' being the most important and influential factor followed by 'Attitude Towards Brand', 'Brand Engagement' and 'Critical Mass' that effect 'Purchase Intention' significantly.

'Brand Attachment' was found to be the most influential factor to influence consumers 'Purchase Intention' that consumer's exhibit (Chapter 5 Section 5.7). It is mostly emotional where a consumer identifies different constructive emotions such as joy, happiness, etc. when associated with a brand. Consumers follow any Facebook Brand Page only if they feel connected and attached to the said brand. Previous studies suggest that 'Brand Attachment' leads to a commitment on the part of the consumer, which further conforms to the buying behaviour with repetition. Thus, brands need to connect with the consumer on an emotional level by employing content, marketing and promotional campaigns designed to address the emotional connect with the consumer. 'Attitude Towards Brand' was found to be the second most important factor to influence the 'Purchase Intention' (Chapter 5 Section 5.7). Consumers tend to develop an attitude towards the brand they either use or admire. This attitude can be either positive or negative based on the perceptions and experiences of the consumer with that brand. Most of the times, the attitude towards the brand is developed based on the brand images and the brand influence due to the marketing and promotions conducted by the said brand. The Online Brand Communities and Facebook Brand Pages help in developing consumer's attitude towards any brand by relevant, informative posts. The promotional information and the user-generated content (which is a part of word-of-mouth communication) help in building a positive attitude; thus, helping FMCG brands in creating a positive brand image using Facebook Brand Pages. The consumers' intent to purchase or the decision to purchase is affected by the attitude that they hold towards the brand.

'Brand Engagement' was found to be the significantly influencing 'Purchase Intention' showing that the engagement consumer exhibit on a Facebook Brand Page (Chapter 5 Section 5.7). It leads to their behavioural intent. Brand Engagement is influenced by 'Communication', 'Content', 'Perceived Benefits', 'Credibility' which are important factors to consider by the FMCG brands while creating a Facebook Brand Page. Their constructive influence on 'Brand Engagement' can affect the consumer's purchase intention. If a consumer feels positive, connected and attached to an FMCG Facebook Brand Page, it might influence its engagement with the band within the Facebook Brand Page as well as outside of the Social Media sphere. The consumer might become an advocate or avid fan of the brand leading to referrals and further purchase behaviour. If consumers felt compelled to engage with the brand or the Facebook Brand Page, that means the brand was successful in capturing the attention of the consumer. Consumers engagement comprise of frequenting the website/brand page of the brand, making others know that they use the brand, referring the brand to their peers, following

the news and other information about the brand on Social Media and otherwise. These behaviours act as precursors to their intent for purchase behaviour.

'Critical Mass' also has a significant effect on 'Purchase Intention'. A consumer is influenced by the social circle that surrounds him/her and with whom he/she regularly interacts (Chapter 5 Section 5.7). Facebook Brand Pages are mostly joined by users when they find their peers in the same community, or they are referred to the said Online Brand Community by someone belonging to the 'Critical Mass'. Since the Online Brand Communities are built on a social networking platform, therefore, the critical mass led engagement may further affect the purchase intentions of consumers.

7.3 Group differences between Gender, Age, Annual Family Income and Weekly Internet Use

Apart from hypotheses testing and validating the proposed model, this study also analysed the moderating effects of Gender, Age, Annual Family Income and Weekly internet usage to see if there are any significant group differences present in the SEM results, within the subgroups of these demographics. Following were established:

Gender

Only one significant group difference was identified in Gender between 'Male' and 'Female' subgroups. Where the strength of a significant relation between 'Brand Attachment' and 'Brand Engagement' were more in the case of Males. Whereas, Females show no significant relationship between 'Brand Attachment' and 'Brand Engagement' (Chapter 5 Section 5.8).

In the year 2018, there were 19.3 crore Male Facebook user as compared to 5.8 crore Female Facebook users in India (Statista, 2019d). As per the statistics, the reason for males showing

the significant relationship might be due to their higher activeness on Facebook. They might be engaging more with the brand they feel attached to. Also, their engagements with any Facebook Brand Page might be with different brands to find about any FMCG product on behalf of their peer group.

As the statistics above show, in the case of females, the same might not be true. The study is conducted in the Indian context. Therefore there might be psychographic reasons behind no significant relationship between Brand Attachment and Brand Engagement. Females have to manage a higher work-life balance due to which they might not be engaging with the FBPs of the brand towarss which they feel attached. The lesser number of females active on Facebook might be linked with a certain age (discussed in the next section).

Age

Two significant group differences were identified in Age between 'Below 30' and 'Above 30' subgroups.

i. The strength of significant relation between 'Brand Attachment' and 'Brand Engagement' was more in the case of respondents below age 30 years. Whereas, respondents above age 30 years showed no significant relation between 'Brand Attachment' and 'Brand Engagement' (Chapter 5 Section 5.8). This difference can be attributed to the facts that the younger generation is more active on Social Networking Sites like Facebook and they prefer to be connected with the brand they like. They are also more technology savvy as compared to the older age group (above 30 years.) which might be the reason for the younger age group (below 30 years.) to show a significant relationship between 'Brand Attachment' and 'Brand Engagement'. The younger generation tends to incline more towards different brands that are

very popular and visible in the cyber-sphere. Thus, engaging more with the Facebook Brand Pages, which they follow out of love for a brand.

ii. The strength of significant relation between 'Brand Attachment' and 'Brand Engagement' was more in the case of respondents below age 30 years. Whereas, respondents above age 30 years showed no significant relationship between 'Critical Mass' and 'Purchase Intention' (Chapter 5 Section 5.8). 'Critical mass' is a group of peers/Family/acquaintances, which affects an individual's decision. The younger age group showed that this group of people influence their decision or intention to purchase, which holds as the younger generation (below 30 years) tends to follow set norms or social requirements which can make them acceptable in a social group. The older age group (above 30 years.) on the other hand, are not dependant on social approvals from 'Critical Mass' and they make their purchase decision based on informational value, Quality of the product and other technical or functional specification. Therefore, having no influence of 'Critical Mass' on 'Purchase Intention' for the subgroup.

Annual family income

Five significant group differences were identified in family income between below INR 750000 per annum and above INR 750000 per annum subgroups.

i. The strength of significant relation between 'Brand Attachment' and 'Attitude Towards Brand' was more in the case of respondents that have a family income below INR 750000 per annum. Whereas, respondents that have family income above INR 750000 per annum showed no significant relation between 'Brand Attachment' and 'Attitude Towards Brand' (Chapter 5 Section 5.8). Individuals belonging to lower-income group (below INR 750000 per annum), are more attracted towards a brand which might be economical in price and has benefits perceived by them. They might focus more on how the brand makes them feel. The engagements by the brand might play a great role in creating an attitude. After which they

might make the purchase. However, Individuals belonging to higher income group (above INR 750000 per annum), does not make any inference about the brands from the engagements on Facebook brand pages. If they have an attachment to a brand, they might prefer to purchase it right away. Also, they might not be focused on developing an attitude and perception about a brand first.

ii. The strength of significant relation between 'Perceived Benefits' and 'Attitude Towards Brand' was more in the case of respondents that have a family income below INR 750000 per annum. Whereas, respondents that have family income above INR 750000 per annum showed no significant relation between 'Brand Attachment' and 'Attitude Towards Brand' (Chapter 5 Section 5.8). Individuals belonging to a lower-income group (below INR 750000 per annum), are more concerned with monetary incentives or the benefits (discounts, sales information, coupons, etc.) provided by the brands on the Facebook Brand Page. They are more likely to interact and draw perception and attitude towards a brand based on the regular posting of such information. However, individuals belonging to higher income group (above INR 750000 per annum), may not be concerned with the benefits posted by the brands and their perception or the attitude might be influenced by other factors like quality and commitment from the brand rather than the discounts and sales coupon provided by the brand.

iii. The strength of significant relation between 'Critical Mass' and 'Attitude Towards Brand' was more in the case of respondents that have family income above INR 750000 per annum. Whereas, respondents that have a family income below INR 750000 per annum showed no significant relation between 'Critical Mass' and 'Attitude Towards Brand' (Chapter 5 Section 5.8). As regards individuals belonging to higher Income group (above INR 750000 per annum), they might be adhering particular lifestyle where the societal acceptance in terms of the decision to purchase was more affluential on the individual's perception of a brand. The peer might influence the individuals' attitude towards a brand. Whereas the individuals belonging

to a lower-income group (below INR 750000 per annum), are not affected by the societal or the peer group they interact with, and their perception and attitude towards the brand are purely based on individual experiences or interactions with the brand.

The strength of significant relation between 'Communication' and 'Brand Engagement' iv. was more in the case of respondents that have family income above INR 750000 per annum. Whereas, respondents that have a family income below INR 750000 per annum showed no significant relation between 'Communication' and 'Brand Engagement' (Chapter 5 Section 5.8). Individuals belonging to the higher income group (above INR 750000 per annum) seek a more communicative engagement with the brand on Facebook Brand Page, thus leading to further engagement with the brand. They might be interested in the brand's capability to connect and engage the consumers on a communicative plane rather than a unidirectional interaction only from the brand. Due to their higher income, they might not be interested in the information regarding discounts and sale, but expecting a better consumer experience from the brand. Whereas, the individuals belonging to a lower-income group (below INR 750000 per annum), might be more interested in the informational communication in the form of images regarding brand information, discounts, or sale. They might be more interested in the price of the product and the benefits of the products in the said price range. The price-sensitive nature of the segment might be the reason there is no significant relationship between 'Communication' and 'Brand Engagement'.

v. The relation between 'Attitude Towards Brand' and 'Purchase Intention' were significant in both subgroups. However, the strength of the significant relationship was more in case of respondents that have family income above INR 750000 per annum. Whereas, respondents that have a family income below INR 750000 per annum showed no significant relation between 'Attitude Towards Brand' and 'Purchase Intention' (Chapter 5 Section 5.8). This might be due to the reason that a positive behavioural attitude towards the brand leads to behaviour intention to purchase, irrespective of the two subgroups.

Weekly internet use

Four significant group differences were identified in Weekly Internet Use between <10hr/week and >10hr/week subgroups.

i. The strength of significant relation between 'Brand Attachment' and 'Attitude Towards Brand' was more in the case of respondents that used the Internet less than 10 hours per weeks. Whereas, respondents that used the Internet for more than 10 hours per week showed no significant relation between 'Brand Attachment' and 'Attitude Towards Brand' (Chapter 5 Section 5.8). Individuals belonging to higher Internet usage group (above INR 10 hours per week) are more active on the Internet they have access to more information about any brand, and thus, their attitude towards a brand is more from the perspective of the information they have collected over the Internet. The information gathered by them might affect their attitude towards the brand more than the information provided by the brand itself. Thus, they rely more on the Internet for information irrespective of their attachments with the brand to form an opinion or attitude about the brand. Individuals belonging to lower Internet usage group (below INR 10 hours per week) are comparatively less active on the Internet than the higher Internet usage group. They might be readily influenced by the engagements done by the brand on digital platforms, including Social Media. Individuals from this group might not be searching more about a product or a brand to gather more information regarding its benefits, price, discount, etc., and solely adhere to what is being presented to them by the respective brand. This behaviour might be existing because they are more attached to the brand; they follow on a Social Media platform. Therefore, for their limited internet use, they rely on the brand due to their attachment with it, which has a significant effect on their attitude towards the brand.

ii. The strength of significant relation between 'Communication' and 'Attitude Towards Brand' was more in the case of respondents that used the Internet less than 10 hrs per week. Whereas, respondents that used the Internet for more than 10 hrs per week showed no significant relation between 'Communication' and 'Attitude Towards Brand' (Chapter 5 Section 5.8). Individuals that show lower Internet activity (< 10hrs/week), mostly engage with specific websites Social Networking Site platforms where they like to connect or collaborate with their peers or the brand. The communication they have with a brand or about a brand on these platforms results in developing their attitude towards the said brand. However, Individuals that show higher Internet activity (> 10hrs/week), are not limited to only a few platforms, but they draw their attitude towards the brand from experiences and exposures from various interactions that occur on different and varied platforms. Their focus is not only on the Facebook Brand Pages but also on other blogs, wikis and discussion boards where they can communicate with the brand as well as with their peers.

iii. The strength of significant relation between 'Content' and 'Attitude Towards Brand' was more in the case of respondents that used the Internet for more than 10 hours per week. Whereas, respondents that used the Internet less than 10 hrs per week showed no significant relation between 'Content' and 'Attitude Towards Brand' (Chapter 5 Section 5.8). Individuals that show higher Internet activity (> 10hrs/week) have more exposure to the information in various forms. They are more aware of the quality and the type of content which shows promise or represent a better image of a brand, thus, developing an attitude towards the said brand. Whereas, individuals that show lower Internet usage (< 10hrs/week), they have lesser exposure and they might not be able to infer or perceive any such attitude towards the brand.

iv. The strength of significant relation between 'Perceived Benefits' and 'Attitude Towards Brand' was more in the case of respondents that used the Internet less than 10 hrs per week. Whereas, respondents that used the Internet for more than 10 hrs per week showed no

significant relation between 'Perceived Benefits' and 'Attitude Towards Brand' (Chapter 5 Section 5.8). Individuals that show higher Internet activity (> 10hrs/week) can source information about sales, discounts and such other monetary incentives or benefits provided by the brand using the Internet and other Social Media platforms apart from brand run Facebook Brand Pages. On the other hand, Individuals that show lower Internet activity (< 10hrs/week) are relying more on the known and popular platforms such as Facebook thus creating an opportunity for the brand to tap these individuals as potential or recurring consumers of their product. Thus, leading to these individuals having a significant influence on the attitude they hold towards the brand due to benefits offered on the Facebook Brand Page.

7.4 Conclusion

Through this chapter, the first and second objectives of the study (Chapter 1 Section 1.6) were achieved with the help of secondary research, and the case on PAL helped in understanding the practices that are being followed by the FMCG brands for Online Consumer Engagement. Apart from the case, various other brands were also investigated from the perspective of understanding their Facebook/Social Networking Site strategy to engage with the consumer. The third and fourth objectives of the study were achieved by empirically testing the proposed model and the relationships/influences of the different factors identified during the study. The model was tested and the results from the analysis presented with the results which address the third objective of this study. Group differences between demographic groups were analysed, and the results suggested that there are few group differences within the proposed model for certain demographic groups, as stated in objective four. The next chapter concludes the study with a conclusion, implications and future scope of research.

Chapter - 8

Conclusion and Implications

8.1 Introduction

The previous Chapter discussed the findings of the study based on the primary data collected as per the methodology discussed in Chapter 4. This chapter presents the conclusion of the study, along with its implications for various stakeholders. The chapter is in four sections. The first section presents the conclusion of the study. The second section discusses the contribution of the study. The third section presents the implications of the study for the stakeholders. The fourth and the last section presents the limitations of the study along with suggestions to future research to extend the research on the relationship between Online Brand Communities and consumer's Purchase Intention.

8.2 Conclusion of the study

With increasing innovations in Internet technologies, masses are becoming aware and interested in using these techniques. Social Networking Sites have become a global phenomenon which assists in social connectivity and interpersonal communications. The Online Brand Communities that have come into existence within these Social Networking Sites have allowed marketers/brands to observe and understand the behaviour and intentions of the consumers who are engaging with such Online Brand Communities. In recent years, the importance of digital media and its use as a channel to establish a dialogue with the consumers has become apparent to the marketers and brands. Social Networking Sites and their increasing influence on people, especially on the younger generation are supported by previous studies. The Online Brand Communities that have evolved within these Social Networking Sites has become a phenomenon in the recent past. Previous studies focused on the importance and need

of integrating Online Brand Communities in business, along with the benefits they provided; such as understanding the behavioural intention of the consumers, their inclination towards the brand and their attitudes. Due to the widespread popularity of the Social Networking Site – Facebook, the study was focussed on the online brand pages present on Facebook. It tries to establish a relationship between using and engaging with the Facebook page that has been created by the brand and the intention to purchase the brand in the future.

Previous studies focused on the acceptance of Social Networking Sites and Online Brand Communities as a medium of communication by the consumers, and there are models likes TAM, TAM2, and UTAUT that encompass significant factors that were pivotal in engagement and thus influencing the behavioural intentions. This study focused on the factors that might play a major role for a marketer or a brand to create a stronger influence on the consumers for creating a behavioural intent towards purchasing. Due to widespread popularity, this study focused on Facebook, one of the most popular Social Networking Site, which is being used globally. This study seeks to understand whether the Facebook Brand Pages (created by FMCG brands) on Facebook have some effect on consumer's attitude towards a brand through the brand's online engagement, which in turn affects their intention to purchase the brand. The current study tries to establish a relationship between using and engaging with the Facebook page and the intention to purchase the brand in the future.

Marketers and brands can benefit from this study by changing their content and promotional strategies based on the results discussed in previous sections. The empirical results show that consumer is inclined towards the type and quality of the content of the posts on Facebook Brand Pages. The marketers and brands need to create and co-create meaningful, enjoyable and informational content. Similarly, the credible nature of information disseminated within the Facebook Brand Page to the users also plays an important part. The majority of focus should

be given to the content generation, and systematic content strategy should be made to exploit the capabilities of the Facebook Brand Pages to the maximum. Previous sections also discussed the group differences within the demographic variable; gender, age, annual income and Internet usage per week. This analysis can help marketers and brands to identify the demographics as per their needs, making segmentation of the population and targeting potential consumers much easy. In particular, the small and medium enterprises in India which do not have adequate resources to undertake extensive promotional campaigns may take advantage of the findings of this study.

8.3 Contribution of the study

The research has addressed the objectives stated in Chapter 1, Section 1.6. The study has made three significant contributions to consumer behaviour in the era of Social Media.

i. First, this is a premier study which links the online consumer engagement to the purchase intention of the consumer in the context with Facebook Brand Pages run by Indian FMCG Brands. Though, there is extant literature which conceptualises and define consumer engagement; the current study augments the understanding of consumer behaviour in Online Brand Communities and their effect on purchase intention.

ii. Second, the research has brought out a model which defines constructs that affect the consumer engagement in Online Brand Communities and Facebook Brand Pages and empirically provide evidence explaining the effects of the engagement within a Facebook Brand Page on purchase intention.

iii. Third and lastly, the study sheds light on the factors that make a Facebook Brand Page effective in engaging the consumer; thus, making it possible to integrate online engagement through Online Brand Communities by the marketers and brand. Since, in India, there exist a large number of small and medium enterprises who do not have enough budget for promotional

activities, Social Networking Sites and Online Brand Communities are an excellent resource for such enterprises. This research is highly appropriate for such enterprises for understanding the key factors of creating a successful Facebook Brand Page for marketing, promotion and consumer engagement.

8.4 Implications of the study

The study has been conducted to help brands, marketers to understand the important factors in Online Consumer Engagement on Facebook Brand Pages. It also addresses the small-medium enterprises that are still in their initial stages and wishes to engage consumers through cost-free capabilities of Social Media platform such as Facebook. This study also attempts to bridge the gaps in the existing literature and would be beneficial to researchers and. Following implications have been suggested by the study:

8.4.1 To practitioners/marketers/brands/SMEs

The study has following implications for brands, marketers and practitioners who are using Facebook brand pages for marketing and promotional purposes. Following implications are suggested to them:

i. Organisations that are planning to engage consumers using Social Media can understand more about the important antecedents to consumers' attitude and their engagement with the brand on Social Networking Site platform Facebook. This understanding should help them formulate strategies based on the proposed framework – especially, for the SMEs which cannot spend much on promotional costs. For them, Facebook provides a low cost but effective channel.

ii. This study can help the marketers in the country to work according to the mindset of Indian consumers since most of the studies done in this field are based on different cultural settings and the results based on these studies from different countries may or may not be generalized
in the Indian context. Consumer's adherence towards the functionality and benefits of a Facebook Brand Page can be understood from the study and thus provide a better vision to strategic promotional campaigns through Facebook Brand Pages.

8.4.2 To academicians/researchers

The study has following implications for researchers and academicians who study social media platforms and online brand communities similar to Facebook brand pages. Following implications are suggested to them:

i. As discussed in previous sections, there is a lack of literature that directly addresses the relationship between Online Consumer Engagement and Purchase Intention. The study attempts to bridge the said gap and add to the existing body of knowledge in the context of Facebook Brand Page based Online Consumer Engagement and its effects on the purchase intention of the consumers. The study is one of a kind to identify factors involved in Online Consumer Engagement in Facebook Brand Pages and empirically test their effect on purchase intention.

ii. This study further adds to the knowledge of Online Consumer Engagement through FMCG brand run Facebook Brand Page. It validates the proposed framework that measures the effects of predictor variables on consumer's purchase intention through consumers attitude towards the FMCG brand and their Brand engagement. Also, the study discusses the difference in the proposed model for different demographical groups to see their moderating effects on the proposed model.

8.5 Limitations of the study

There were six major limitations to this study, thus requiring caution while interpreting the results:

- i. First, literature is scarce when it comes to empirically tested studies on consumer engagement and its influences on purchase intention.
- ii. Second, a sample of 501 respondents is considered for this study, which might not be large enough.
- iii. Third, convenience sampling is used for the study, which is a non-random sampling technique.
- iv. Fourth, the study follows a cross-sectional design which was based on a questionnaire collected in a limited time frame.
- v. Fifth, the study was focussed and limited to the social networking platform, Facebook.
- vi. Sixth, the study was structured and designed, considering the population residing within India.

8.6 Future scope of the study

The study suggests further research in the following areas:

- i. There might be some other factors which also contribute towards the dependent variables which can be included in future extension studies. Further research can be conducted to identify such factor(s) and incorporate them into the proposed framework
- ii. A similar study can be conducted on other SNS platforms such as Twitter, YouTube, etc.to understand the dynamics of these platforms for the factors identified in this study.
- iii. This study can be replicated in other research settings to validate the socio-cultural impact on the model proposed, as well as the effects of the moderating variables. Therefore, similar studies in the context of a different country might provide new insight and findings.

References

Abramyk, H. (2017), "Top 10 Review Websites to Get More Consumer Reviews On". Available at: https://www.vendasta.com/blog/top-10-consumer-review-websites (accessed on 29 August 2017).

Afetian, S. (2013), "Rise Of The Interest-Based Network". Available at: http://www.cmo.com/features/articles/2013/11/27/rise_of_the_interest.html#gs.uAVjPF0 (accessed on 30 August 2017).

Affilorama, (n.d.), "What is a media sharing site?". Available at: https://www.affilorama.com/internet101/media-sharing (accessed om 29 August 2017).

Ahluwalia, R., Burnkrant, R. E., & Unnava, H. R. (2000), "Consumer response to negative publicity: The moderating role of commitment", *Journal of marketing research*, Vol. 37 No. 2, pp. 203-214.

Aichner, T. and Jacob, F. (2015), "Measuring the Degree of Corporate Social Media Use", *International Journal of Market Research*. Vol. 57 No. 2, pp. 257–275.

Ajzen, I., & Fishbein, M. (1980). Understanding attitudes and predicting social behaviour.

Alexa (2017), "The top 500 sites on the web", Available at: https://www.alexa.com/topsites/category/Computers/Internet/On_the_Web/Online_Communi ties/Social_Networking (accessed on 23 August, 2017).

Algesheimer, R., Dholakia, U. M., & Herrmann, A. (2005), "The social influence of brand community: Evidence from European car clubs", *Journal of marketing*, Vol. 69 No. 3, pp.19-34.

Amin, H. (2007), "Extending TAM to SMS banking: analyzing the gender gap among students", *International Journal of Business and Society*, Vol. 8 No. 1, pp. 24-44.

Appelbaum A. (2001), "The constant consumer", *Gallup Management Journal*. Available at http://gmj.gallup.com/content/745/Constant-Consumer.aspx (accessed 24 April 2018).

Aradhak, P. (2018), "430 acres allotted to Patanjali", Available at: https://www.thehindu.com/news/cities/Delhi/430-acres-allotted-to-Patanjali/article16984837.ece (accessed on 25 April, 2018).

Arbuckle, J.L. and Wothke, W. (1999), Amos 4.0 User's Guide, SPSS, Chicago, IL.

Ashley, C., & Tuten, T. (2015), "Creative strategies in Social Media marketing: An exploratory study of branded social content and consumer engagement", *Psychology & Marketing*, Vol. 32 No. 1, pp. 15-27.

Ashley, C., & Tuten, T. (2015), "Creative strategies in Social Media marketing: An exploratory study of branded social content and consumer engagement", *Psychology & Marketing*, Vol. 32 No. 1, pp. 15-27.

Aslam, S. (2017), "Twitter by the Numbers: Stats, Demographics & Fun Facts". Available at: https://www.omnicoreagency.com/twitter-statistics/ (accessed on 24 August 2017).

Assiouras, I., Liapati, G., Kouletsis, G., & Koniordos, M. (2015), "The impact of brand authenticity on brand attachment in the food industry", *British Food Journal*, Vol. 117 No. 2, pp. 538-552.

Bagozzi, R. P. and Dholakia, U. M. (2006), "Antecedents and purchase consequences of consumer participation in small group brand communities", *International Journal of Research Marketing*, Vol. 23 No. 1, pp. 45-61.

Bambauer-Sachse, S., and Mangold, S. (2011), "Brand Equity Dilution through Negative Online Word-of-Mouth Communication," *Journal of Retailing and Consumer Services* Vol. 18 No. 1, pp. 38–44.

Bansal, H. and Voyer, P.A. (2000), "Word of mouth processes within a services purchase decision context", Journal of Service Research, Vol. 3 No. 2, pp. 166-177.

Beal,V.(n.d.),"socialshopping".Availableat:http://www.webopedia.com/TERM/S/social_shopping.html (accessed on 30 August 2017).

Bearden, W., Netemeyer, R. and Teel, J. (1989), "Measurement of consumer susceptibility to interpersonal influence", Journal of Consumer Research, Vol. 15, pp. 473-481.

Beese, J. (2016), "Social Media Ecommerce: The Complete Guide to Social Shopping". Available at: https://sproutsocial.com/insights/social-media-ecommerce/ (accessed on 25 August 2017)

Belaid, S. and Temessek Behi, A. (2011), "The role of attachment in building consumer-brand relationships: an empirical investigation in the utilitarian consumption context", *Journal of Product & Brand Management*, Vol. 20 No. 1, pp. 37-47.

Bell, E., & Bryman, A. (2007), "The ethics of management research: an exploratory content analysis", *British Journal of Management*, Vol. 18 No. 1, pp. 63-77.

Bhushan, R. & Chakravarty, C. (2018), "Baba Ramdev's Patanjali made the FMCG sector better: Rakesh Kapoor, Reckitt CEO", Available at: https://economictimes.indiatimes.com/industry/cons-products/fmcg/baba-ramdevs-patanjalimade-the-fmcg-sector-better-rakesh-kapoor-reckitt-ceo/articleshow/63034899.cms (accessed on 27 April 2018).

Blackboard, (n.d.), "About Discussions, Forums, and Threads". Available at: https://help.blackboard.com/Learn/Instructor/Interact/Discussions/About_Discussions_Forum s_Threads (accessed on 29 August 2017).

Bloomberg (2018), "Company Overview of Patanjali Ayurved Limited", Available at: https://www.bloomberg.com/research/stocks/private/snapshot.asp?privcapId=130378187 (accessed on 25 April, 2018).

Boatin, C. (2017), "Social Media for Medical Marketing: More is More". Available at: https://www.outletcreativegroup.com/blog/social-media-for-medical-marketing-more-is-more (accessed on 24 August 2017).

Bonds-Raacke, J.& Raacke, J. (2010), "MySpace and Facebook: Identifying Dimensions of Uses and Gratifications for Friend Networking Sites," *Individual Differences Research*, Vol. 8 No. 1, pp. 27–34.

Bowden, J.L. (2009), "The process of consumer engagement: A conceptual framework", *Journal of Marketing Theory and Practice*, Vol. 17 No. 1, pp. 63-74.

Boyd, D.M. and Ellison, N.B. (2007), "Social Network Sites: Definition, History, and Scholarship", *Journal of Computer-Mediated Communication*, Vol. 13 No. 1, pp. 210–30.

Breuer, A., (2011), "Democracy Promotion in the Age of Social Media: Risks and Opportunities", German Development Institute, Briefing Paper.

Brodie, R. J., Hollebeek, L. D., Jurić, B., & Ilić, A. (2011), "Consumer engagement: Conceptual domain, fundamental propositions, and implications for research", *Journal of service research*, Vol. 14 No. 3, pp. 252-271. Brodie, R.J., Ilic, A., Biljana, J. and Hollebeek, L. (2011b), "Consumer engagement in a virtual brand community: an exploratory analysis", Journal of Business Research, available online Science Direct from 23rd August (in press).

Brown, J.J. and Reingen, P.H. (1987), "Social ties and word-of-mouth referral behavior", *Journal of Consumer Research*, Vol. 14 No. 3, pp. 350-362.

Browne, M.W. and Cudeck, R. (1993), Alternative Ways of Assessing Model Fit, Sage Publications, Newbury Park, C

Bruhn, M., Schoenmueller, V. and Scha⁻fer, D. B. (2012), "Are Social Media Replacing Traditional Media in Terms of Brand Equity Creation?" *Management Research Review* Vol. 35 No. 9, pp. 770–790.

Budzienski, J. (2017), "3 Ways to Be Constantly Recruiting Star Talent Through Social Media". Available at: https://hiring.monster.ca/hr/hr-best-practices/recruiting-hiring-advice/attracting-job-candidates/recruit-talent-through-social-media-ca.aspx (accessed on 26 August).

Burns, K.S. (2010, August), "Brands among friends: An examination of brand friending and engagement on Facebook", *Paper presented at the 94th Annual Conference of the Association for Education in Journalism and Mass Communication*, Denver, CO.

Byrne, B. M. (2010), "Structural equation modeling with AMOS: Basic concepts, applications, and programming". 2nd Edition. Routledge Taylor & Francis Group.

Cao, J., Knotts, T., Xu, J. and Chau, M. (2009), "Word of mouth marketing through online social networks", *Proceedings of the Fifteenth Americas Conference on Information Systems*, San Francisco, CA, 6-9 August.

Carroll, B. A., & Ahuvia, A. C. (2006), "Some antecedents and outcomes of brand love", *Marketing letters*, Vol. 17 No. 2, pp. 79-89.

Casaló, L. V., Flavián, C. and Guinalíu, M. (2013), "New members' integration: Key factor of success in online travel communities", *Journal of Business Research*, Vol. 66 No. 6, pp. 706-710.

Chauhan, K. and Pillai, A. (2013), "Role of content strategy in Social Media brand communities: a case of higher education institutes in India", *Journal of Product and Brand Management*, Vol. 22 No. 1, pp. 40-51.

Chen, Q. and Wells, W. (1999), "Attitude toward the site", *Journal of Advertising Research*, Vol. 39 No. 5, pp. 27-37.

Chen, Q., Clifford, S. and Wells, W. (2002), "Attitude toward the site II: new information", *Journal of Advertising Research*, Vol. 42 No. 2, pp. 33-44.

Cheng, X., Liu, J., Wang, H., & Wang, C. (2012), "Coordinate live streaming and storage sharing for Social Media content distribution", *IEEE Transactions on Multimedia*, Vol. 14 No. 6, pp. 1558-1565.

Chu, S.C. and Kim, Y. (2011), "Determinants of consumer engagement in electronic word-ofmouth (eWOM) in social networking sites", *International Journal of Advertising*, Vol. 30 No. 1, pp. 47-74.

Chung, D. (2008), "Interactive features of online newspapers: identifying patterns and predicting use of engaged readers", *Journal of Computer Mediated Communication*, Vol. 13 No. 3, pp. 658-679.

Cisnero, K. (2014), "A Beginner's Guide to Content Curation". Available at: https://blog.hootsuite.com/beginners-guide-to-content-curation/ (accessed on 28 August 2017).

Copp, E. (2017), "10 Benefits of Social Media for Business", https://blog.hootsuite.com/socialmedia-for-business/ (accessed on 20 July, 2017).

Coulter, K.S. and Roggeveen, A. (2012), "Like it or not: consumer responses to word-of-mouth communication in on-line social networks", *Management Research Review*, Vol. 35 No. 9, pp. 878-899.

Coursaris, C. K., van Osch, W., & Balogh, B. A. (2016), "Do Facebook likes lead to shares or sales? exploring the empirical links between Social Media content, brand equity, purchase intention, and engagement", In *System Sciences (HICSS), 2016 49th Hawaii International Conference on* (pp. 3546-3555). IEEE.

Cova, B. and Pace, S. (2006), "Brand community of convenience products: New forms of consumer empowerment – the case "My Nutella" community", *European Journal of Marketing*, Vol. 40 No. 9-10, pp. 1087-1105.

Coyle, J.R. and Thorson, E. (2001), "The effects of progressive levels of interactivity and vividness in web marketing sites", Journal of Advertising, Vol. 30 No. 3, pp. 65-77.

Crittenden, V. L., Peterson, R. A., & Albaum, G. (2010), "Technology and business-toconsumer selling: Contemplating research and practice", *Journal of Personal Selling & Sales Management*, Vol. 30 No. 2, pp. 103-109.

Cronbach, L.J. (1951), "Coefficient alpha and the internal structure of tests", *Psychometrika*., Vol 16, pp. 297–334.

cti09 (n. d.) Types of Virtual Communities, Available at: https://cti09.wikispaces.com/Types+of+Virtual+Communities (accessed on 15 March, 2018).

Cvijikj, I. P., & Michahelles, F. (2011), "Understanding Social Media marketing: a case study on topics, categories and sentiment on a Facebook brand page", In *Proceedings of the 15th international academic MindTrek conference: Envisioning future media environments, Tampere*, Finland, pp. 175-182.

Cvijikj, I. P., & Michahelles, F. (2013), "Online engagement factors on Facebook brand pages", *Social Network Analysis and Mining*, Vol. 3 No. 4, pp.843-861.

Dayal, S. (2016), "Interest-based networks are the way of future". Available at: https://yourstory.com/2016/11/interest-based-networks-future/ (accessed on 30 August 2017).

De Beule, S. (2013), "4 Types of Consumer Engagement to Build Die Hard Brand Advocates", Available at: https://cxsocial.clarabridge.com/4-types-of-consumer-engagement-to-build-diehard-brand-advocates/ (accessed on 10th January 2018).

De Bruyn, A. and Lilien, G.L. (2008), "A multi-stage model of word of mouth influence through viral marketing", International Journal of Research in Marketing, Vol. 25, pp. 151-164.

De Vries, L., Gensler, S. and Leeflang, P.S. (2012), "Popularity of brand posts on brand fan pages: an investigation of the effects of Social Media marketing", *Journal of Interactive Marketing*, Vol. 26 No. 2, pp. 83-91.

Dholakia, U. M., Bagozzi, R., & Pearo, L. K. (2004), "A social influence model of consumer participation in network and small group based virtual communities", *International Journal of Research in Marketing*, Vol. 21 No. 3, pp. 241–263.

Dholakia, U. M., Blazevic, V., Wiertz, C., & Algesheimer, R. (2009), "Communal service delivery: how consumers benefit from participation in firm-hosted virtual P3 communities", *Journal of Service Research*, Vol. 12 No. 2, pp. 208-226.

Douglas, J.Y., & Hargadon A. (2001), "The pleasures of immersion and engagement: Schemas, scripts and the fifth business", *Digital Creativity*, Vol. 12 No. 3, pp. 153–66.

Dunay, P. (2013), "10 E-Commerce Predictions For 2013". Available at: https://www.forbes.com/sites/gyro/2012/12/24/10-e-commerce-predictions-for-2013/ (accessed on 24 August 2017).

Dutta, A. (2017). "Ramdev's Patanjali shakes up FMCG order; next up 'nutritious restaurants", Available at: https://web.archive.org/web/20170715102827 /http://wap.business-standard.com/article-amp/markets/wap. business-standard.com/article/companies/ramdev-s-patanjali-doubles-sales-to-rs-10-561-crore-in-fy17-117050400652_1.html Business Standard 15th July 2017 (accessed on 25 April 2018).

Economist, (2013), "The rise of the sharing economy". Available at: https://www.economist.com/news/leaders/21573104-internet-everything-hire-rise-sharing-economy (accessed on 30 August 2017).

economictimes (2019a), "Hindustan Unilever Ltd.". Available at: https://economictimes.indiatimes.com/hindustan-unilever-ltd/yearly/companyid-13616.cms (accessd on 20 February, 2019).

economictimes (2019b), "Colgate-Palmolive (India) Ltd.". Available at: https://economictimes.indiatimes.com/colgate-palmolive-(india)ltd/profitandlose/companyid-13873.cms (accessd on 20 February, 2019).

economictimes (2019c), "ITC Ltd.". Available at: https://economictimes.indiatimes.com/itcltd/yearly/companyid-13554.cms (accessed on 20 February, 2019).

economictimes (2019d), "Nestle India Ltd.". Available at: https://economictimes.indiatimes.com/nestle-india-ltd/yearly/companyid-13330.cms (accessed on 20 February, 2019).

economictimes (2019e), "Britannia Industries Ltd.". Available at: https://economictimes.indiatimes.com/britannia-industries-ltd/yearly/companyid-13934.cms (accessd on 20 February, 2019).

economictimes (2019f), "Marico Ltd.". Available at: https://economictimes.indiatimes.com/marico-ltd/quarterly/companyid-5886.cms (accessd on 20 February, 2019). economictimes (2019g), "AProcter & Gamble Hygiene & Healthcare Ltd.". Available at: https://economictimes.indiatimes.com/procter-&-gamble-hygiene-&-healthcare-ltd/profitandlose/companyid-13253.cms (accessed on 20 February, 2019).

economictimes (2019h), "Godrej Consumer Products Ltd.". Available at: https://economictimes.indiatimes.com/godrej-consumer-products-ltd/yearly/companyid-1908.cms (accessed on 20 February, 2019).

economictimes (2019i), "Amul turnover grows 8% to Rs 29,085 crore in 2017-18". Available at: https://economictimes.indiatimes.com/industry/cons-products/food/amul-turnover-grows-8-to-rs-29085-crore-in-2017-18/articleshow/63583079.cms (accessed on 20 February, 2019).

economictimes (2019j), "Dabur India Ltd.". Available at: https://economictimes.indiatimes.com/dabur-india-ltd/yearly/companyid-11796.cms (accessed on 20 February, 2019).

Ellison, N.B. (2007), "Social Network Sites: Definition, History, and Scholarship", *Journal of computer-mediated communication*, Vol. 13 No. 1, pp. 210–230.

Entrepreneur (2017), "Competitive Analysis". Available at: https://www.entrepreneur.com/encyclopedia/competitive-analysis (accessed on 25 August 2016).

equitymaster (2018), "Fast Moving Consumer Goods Sector Analysis Report", Available at: https://www.equitymaster.com/research-it/sector-info/consprds/Consumer-Products-Sector-Analysis-Report.asp (accessed on 2 April, 2018).

Esch, F.R., Langner, T., Schmitt, B.H. and Geus, P. (2006), "Are brands forever? How brand knowledge and relationships affect current and future purchases", *Journal of Product & Brand Management*, Vol. 5 No. 2, pp. 98-104.

Evans, D., & McKee, J. (2010), "Social Media marketing. The next generation of business Engagement". Indianapolis, IN: Wiley Publishing, Inc.

F. Hair, Joseph & Black, William & Babin, Barry & Anderson, Rolph. (2010). Multivariate Data Analysis: A Global Perspective.

Featherman, M.S. and Pavlou, P.A. (2003), "Predicting e-services adoption: a perceived risk facets/perspective", Journal of Human-Computer Studies, Vol. 59 No. 4, pp. 451-474.

Fedorikhin, A., Park, C.W. and Thomson, M. (2008), "Beyond fit and attitude: the effect of emotional attachment on consumer responses to brand extensions", Journal of Consumer Psychology, Vol. 8 No. 4, pp. 281-291.

Ference, A. (2017), "7 Consumer Engagement Strategies That Marketers Can't Ignore". Available at: https://www.outbrain.com/blog/3-golden-consumer-engagement-strategies-that-marketers-cant-ignore/ (accessed on 20 August 2018).

FinancialExpress (2018a), "Rural FMCG market to reach \$220 billion by 2025: Study", Available at: https://www.financialexpress.com/industry/rural-fmcg-market-to-reach-220-billion-by-2025-study/934453/ Financial Express 15th November 2017, (Accessed on 4 April, 2018).

FinancialExpress (2018b), "Mukesh Ambani's Reliance Jio is 3rd most influential brand in India, and top 2 are not Indian companies", Available at: https://www.financialexpress.com/ industry/mukesh-ambanis-reliance-jio-3rd-most-influential-brand-in-india-and-top-2-are-not-indian-companies/1266031/lite/?__twitter_impression=true (accessed on 25 April, 2018).

Fiore, A.M., Jihyun, K., & Hyun-Hwa, L. (2005), "Effect of image interactivity technology on consumer responses toward the online retailer", *Journal of Interactive Marketing*, Vol. 19 No. 3, pp. 38–53.

Fishbein, M., & Ajzen, I. (2005), "Theory-based behavior change interventions: Comments on Hobbis and Sutton", *Journal of health psychology*, Vol. 10 No. 1, pp. 27-31.

Foreman, C. (2017), "10 Types of Social Media and How Each Can Benefit Your Business", https://blog.hootsuite.com/types-of-social-media/ (accessed on 20 July, 2017).

Fornell, C., & Larcker, D. F. (1981), "Structural equation models with unobservable variables and measurement error: Algebra and statistics", *Journal of marketing research*, pp. 382-388.

Fortin, D.R. and Dholakia, R.R. (2005), "Interactivity and vividness effects on social presence and involvement with a web-based advertisement", Journal of Business Research, Vol. 58 No. 3, pp. 387-396.

Ganguly, (2018), "India Crosses 500 Mn Internet Connections as Rural Users Surge". Available at: https://inc42.com/buzz/india-crosses-500-mn-internet-connections-as-rural-users-surge/ (accessed on 13 January 2019).

Gaskin, J. (2012). Confirmatory factor analysis. Gaskination's StatWiki.

Glass, R., & Li, S. (2010), "Social influence and instant messaging adoption", *Journal of Computer Information Systems*, Vol. 51 No. 2, pp. 24-30.

Godes, D., and Mayzlin. D. (2009), "Firm-Created Word-of-Mouth Communication: Evidence from a Field Test." *Marketing Science* Vol. 28 No. 4, pp. 721–739.

Goldenberg, J., Libai, B., & Muller, E. (2001), "Talk of the network: A complex systems look at the underlying process of word-of-mouth", *Marketing letters*, Vol. 12 No. 3, pp. 211-223.

Goldsmith, R. E. (2012)," Brand engagement and brand loyalty", In A. Kapoor, & C. Kulshrestha (Eds.), Branding and sustainable competitive advantage: Building virtual presence (pp. 121-135).

Greenland, S., Senn, S. J., Rothman, K. J., Carlin, J. B., Poole, C., Goodman, S. N., & Altman, D. G. (2016). Statistical tests, *P* values, confidence intervals, and power: a guide to misinterpretations. *European Journal of Epidemiology*, *31*, 337–350. http://doi.org/10.1007/s10654-016-0149-3.

Gretzel, U. and Dinhopl, A. (2014), "Breaking up is hard to do: why do travelers unlike travelrelated organizations?", *Information and Communication Technologies in Tourism*, p. 267.

Gretzel, U. and Fesenmaier, D.R. (2012), "Consumer relations 2.0 – implications for destination marketing", paper presented at the TTRA Annual International Conference, VA Beach.

Gummerus J., Liljander V., Weman, E., Pihlström, M. (2012), "Consumer engagement in a Facebook brand community", *Management Research Review*, Vol. 35 No. 9, pp. 857 – 877.

Gurd, J. (2016), "Understanding the role of organic vs paid Social Media". Available at: http://www.smartinsights.com/social-media-marketing/social-media-strategy/understanding-role-organic-paid-social-media/ (accessed on 25August 2017).

Guthrie, J.T, Wigfield, A., Barbosa P., Perencevich, K.C., Taboada, A., & Davis, M.H., et al. (2004), "Increasing reading comprehension and engagement through concept-oriented reading instruction", *Journal of Educational Psychology*, Vol. 96 No. 3, pp. 403–23.

Hair, J. F., Black, W. C., Rabin, B. J., & Anderson, R. E. (2010). Multivariate data analysis: A global perspective. New Jersey: Pearson Prentice Hall.

Hair, J.F., Black, W.C., Babin, B.J. and Anderson, R.E. (2010), Multivariate Data Analysis: A Global Perspective, 7th ed., Prentice Hall, NJ.

Hajli, N., Shanmugam, M., Powell, P. and Love, P. E. D. (2015), "A study on the continuance participation in on-line communities with social commerce perspective", *Technological Forecasting and Social Change*, Vol. 96, pp. 232-241.

Hanna, R., Rohm, A., and Crittenden, V. L. (2011), "We're all connected: The power of the Social Media ecosystem", *Business horizons*, Vol. 54 No. 3, pp. 265-273.

Harden, L., & Heyman, B. (2009), "Digital engagement: Internet marketing that captures consumers and builds intense brand loyalty", New York: AMACOM.

Hargittai, E. (2007), "Whose space? Differences among users and non-users of social network sites", *Journal of Computer-Mediated Communication*, Vol. 13 No. 1, pp. 276-297.

Harris, A. L., & Rea, A. (2009), "Web 2.0 and virtual world technologies: A growing impact on IS education", *Journal of Information Systems Education*, Vol. 20 No. 2, pp. 137.

Hassan, M., & Casaló Ariño, L. V. (2016), "Consumer devotion to a different height: How consumers are defending the brand within Facebook brand communities", *Internet Research*, Vol. 26 No. 4, pp. 963-981.

Haythornthwaite, C. (2005), "Social networks and Internet connectivity effects", *Information, Communication, & Society*, Vol. 8 No.2, pp. 125–147.

Heath, R. (2007), "How do we predict advertising attention and engagement?" School of Management University of Bath Working Paper, Vol 9.

Higgins, T.E., & Scholer, A. (2009), "Engaging the consumer: The science and art of the value creation process", *Journal of Consumer Psychology*, Vol. 19 No. 2, pp. 100-114.

Ho, K. (2017), "41 Up-to-Date Facebook Facts and Stats". Available at: https://blog.wishpond.com/post/115675435109/40-up-to-date-facebook-facts-and-stats (accessed on 24 August 2017).

Hollebeek, L. (2011), "Exploring consumer brand engagement: definition and themes", *Journal of strategic Marketing*, Vol. 19 No. 7, pp. 555-573.

Hollebeek, L. D. (2013), "The consumer engagement/value interface: An exploratory investigation", *Australasian Marketing Journal (AMJ)*, Vol. 21 No. 1, pp. 17-24.

Hollebeek, L.D., Glynn, M.S., Brodie, R.J. (2014), "Consumer Brand Engagement in Social Media: Conceptualization, Scale Development and Validation", *Journal of Interactive Marketing*, Vol. 28, pp. 149-164.

Hsu, C.L. and Lu, H.P. (2004), "Why do people play on-line games? An extended TAM with social influences and flow experience", *Information & Management*, Vol. 41 No. 7, pp. 853-68.

ibef (2018a), "Fast moving consumer goods (FMCG)", Available at: https://www.ibef.org/download/FMCG-February-2018.pdf (accessed on 26 March, 2018).

ibef (2018b), "FMCG Industry in India", Available at: https://www.ibef.org/industry/fmcg.aspx (accessed on 13 February 2018).

Jackson, D. (2017), "Ultimate Guide to Social Media Competitive Analysis". Available at: https://sproutsocial.com/insights/social-media-competitive-analysis/ (accessed on 25 August 2016).

Jahn, B. and Kunz, W. (2012), "How to transform consumers into fans of your brand", Journal of Service Management, Vol. 23 No. 3, pp. 344-361.

Jain, V. (2018), "Allowing 100% FDI in single brand retail will improve ease of doing biz: Industry", Available at: https://retail.economictimes.indiatimes.com/news/industry/allowing-100-fdi-in-sbrt-will-enhance-ease-of-doing-biz-industry/62443952, Economic Times 10th January 2018, (Accessed on 4 April, 2018).

Jang, H., Olfman, L., Ko, I., Koh, J., & Kim, K. (2008), "The influence of on-line brand community characteristics on community commitment and brand loyalty", *International Journal of Electronic Commerce*, Vol. 12 No. 3, pp. 57-80.

Jayasingh, S., & Venkatesh, R. (2015), "Consumer engagement factors in Facebook brand pages,", *Asian Social Science*, Vol. 11 No. 26, pp. 19.

Jin, B. and Suh, Y.G. (2005), "Integrating effect of consumer perception factors in predicting private brand purchase in a Korean discount store context", *Journal of Consumer Marketing*, Vol. 22 No. 2, pp. 62-71.

Kabadayi, S., Price, K. (2014), "Consumer – brand engagement on Facebook: liking and commenting behaviors", *Journal of Research in Interactive Marketing*, Vol. 8 No. 3 pp. 203 – 223.

Kane, G.C., Fichman, R.G., Gallaugher, J. and Glaser, J. (2009), "Community relations 2.0", Harvard Business Review, Vol. 87 No. 11, pp. 45-50.

Kang, M.S. (2004), "The research regarding the introduction and an activity of community concept from marketing," *The Korea Society of Management Education Research*, Vol. 7 No. 2, pp. 77-9

Kaplan, A. M. (2012), "If you love something, let it go mobile: Mobile marketing and mobile Social Media 4x4", *Business Horizons*, Vol. 55 No. 2: pp. 129–139.

Kaplan, A. M., & Haenlein, M. (2010), "Users of the world, unite! The challenges and opportunities of Social Media", *Business horizons*, Vol. 53 No. 1, pp. 59-68.

Kaplan, A. M., & Haenlein, M. (2012), "Two hearts in three-quarter time: How to waltz the Social Media/viral marketing dance", *Business Horizons*, Vol. 54 No. 3, pp. 253-263.

Keller, K. L. (2001), "Building consumer based brand equity: A blueprint for creating strong brands", *MSI Report*, pp. 1-107.

Kemp, S. (2017), "Digital in 2017: Global Overview" Available at: https://wearesocial.com/special-reports/digital-in-2017-global-overview (accessed on 24 August 2017).

Khang, H., Ki, E.J. and Ye, L. (2012), "Social Media research in advertising, communication, marketing, and public relations", Journalism and Mass Communication Quarterly, Vol. 89 No. 2, pp. 279-298.

Kim, A. J., & Johnson, K. K. (2016), "Power of consumers using Social Media: Examining the influences of brand-related user-generated content on Facebook", *Computers in Human Behavior*, Vol. 58, pp. 98-108.

Kirkpatrick, D. (2011), "The Facebook effect: the real inside story of Mark Zuckerberg and the world's fastest-growing company". London: Virgin.

Lakhi, A (2018), "List of Top 10 FMCG Leading Companies in India (2019)". Available at: https://www.reviewsxp.com/blog/fmcg-companies-in-india/ (accessed on 13 January, 2018).

Laskey, H. A., Day, E., & Crask, M. R. (1989), "Typology of main message strategies for television commercials", *Journal of advertising*, Vol. 18 No. 1, pp. 36-41.

Lee, D.H. (2010), "Korean consumer & society: growing popularity of Social Media and business strategy", SERI Quarterly, Vol. 3 No. 4, pp. 112-1179.

Leong, L.Y., Hew, T.S., Ooi, K.B. and Lin, B. (2011), "Influence of gender and English proficiency on Facebook mobile adoption", International Journal of Mobile Communication, Vol. 9 No. 5, pp. 495-521.

Leonidou, L.C., Leonidou, C.N. and Kvasova, O. (2010), "Antecedents and outcomes of consumer environmentally friendly attitudes and behavior", *Journal of Marketing Management*, Vol. 26 No. 13/14, pp. 1319-1344.

LePage, E. (2017), "How to Create a Social Media Marketing Plan in 6 Steps". Available at: https://blog.hootsuite.com/how-to-create-a-social-media-marketing-plan/ (accessed on 28 August 2017).

Lewis, K., Kaufman, J., Gonzalez, M., Wimmer, A., & Christakis, N. (2008), "Tastes, ties, and time: A new social network dataset using Facebook. Com", *Social networks*, Vol. 30 No. 4, pp. 330-342.

Lewis, K.N., Palmer, A. and Moll, A. (2010), "Predicting young consumer's take up of mobile banking services", *International Journal of Bank Marketing*, Vol. 28 No. 5, pp. 410-432.

Liébana-Cabanillas, F.J., Sánchez-Fernández, J. and Muñoz-Leiva, F. (2014), "Role of gender on acceptance of mobile payment", *Industrial Management & Data Systems*, Vol. 114 No. 2, pp. 220-240.

Limbu, Y.B., Wolf, M. and Lunsford, D. (2012), "Perceived ethics of online retailers and consumer behavioral intentions: the mediating roles of trust and attitude", *Journal of Research in Interactive Marketing*, Vol. 6 No. 2, pp. 133-154.

Lin, H-F. (2008), "Determinants of successful virtual communities: Contributions from system characteristics and social factors", *Information and Management*, Vol. 45 No. 8, pp. 522-527.

Lin, K.-Y. and Lu, H.-P. (2011), "Intention to continue using Facebook fan pages from the perspective of social capital theory", *Cyberpsychology, Behavior, and Social Networking*, Vol. 14 No. 10, pp. 565-570.

Logan, K. (2014), "Why Isn't Everyone Doing It? A Comparison of Antecedents to Following Brands on Twitter and Facebook", *Journal of Interactive Advertising*, Vol. 14 No. 2, pp. 60-72.

Lou, H., Luo, W. and Strong, D. (2000), "Perceived critical mass effect on groupware acceptance", *European Journal of Information Systems*, Vol. 9 No. 2, pp. 91-104.

Louis, D. and Lombart, C. (2010), "Impact of brand personality on three major relational consequences (trust, attachment, and commitment to the brand)", *Journal of Product & Brand Management*, Vol. 19 No. 2, pp. 114-130.

Louis, D., & Lombart, C. (2010), "Impact of brand personality on three major relational consequences (trust, attachment, and commitment to the brand)", *Journal of Product & Brand Management*, Vol. 19 No. 2, pp. 114-130.

Luarn, P., Lin, Y. F., & Chiu, Y. P. (2015), "Influence of Facebook brand-page posts on online engagement", *Online Information Review*, Vol. 39 No. 4, pp. 505-519.

Lusch, R.F., & Vargo, S.L. (2010), "S-D logic: Accommodating, integrating, transdisciplinary", Grand Service Challenge: University of Cambridge.

Madan, K., & Yadav, R. (2016), "Behavioural intention to adopt mobile wallet: a developing country perspective", *Journal of Indian Business Research*, Vol. 8 No. 3, pp. 227-244.

Malhotra, A., Malhotra, C.K. and See, A. (2013), "How to crate brand engagement on Facebook", *MIT Sloan Management Review*, Vol. 54 No. 2, pp. 18-20.

Mangold, W. G., and D. J. Faulds. (2009), "Social Media: The New Hybrid Element of the Promotion Mix." *Business Horizons* Vol. 52 No. 4, pp. 357–364.

Marci, C.D. (2006), "A biologically based measure of emotional engagement: Context matters", *Journal of Advertising Research*, Vol. 46, pp. 381-387.

Markus, U. (2002), "Characterising the virtual community", Available at: http://www.sapdesignguild.org/editions/edition5/print_communities.asp (accessed on 15 March, 2018).

Martínez-López, F. J., Anaya-Sánchez, R., Aguilar-Illescas, R., & Molinillo, S. (2015), "Online Brand Communities: Using the Social Web for Branding and Marketing". Springer.

Martínez-López, F. J., Anaya-Sánchez, R., Molinillo, S., Aguilar-Illescas, R., & Esteban-Millat, I. (2017), "Consumer engagement in an online brand community", *Electronic Commerce Research and Applications*, Vol. 23, pp. 24-37. Mashele, F., & Chuchu, T. (2018), "An empirical investigation into the relationship between sustainability and supply chain compliance within the South African Public and the private sector", *Journal of Business and Retail Management Research*, Vol. 12 No. 2.

Mathwick, C., & Rigdon, E. (2004), "Play, flow, and the online search experience", *Journal of Consumer Research*, Vol. 31, pp. 324-333.

McAlexander, J.H., Kim, S.K. and Roberts, S.D. (2003), "Loyality: the influences of satisfaction and brand community integration", *Journal of Marketing Theory and Practice*, Vol. 11 No. 4, pp. 1-11.

McKnight, D. H., Choudhury, V., & Kacmar, C. (2002), "Developing and validating trust measures for e-commerce: an integrative typology", *Information Systems Research*, Vol. 13 No. 3, pp. 334-359.

Md Husin, M., Ismail, N., & Ab Rahman, A. (2016), "The roles of mass media, word of mouth and subjective norm in family takaful purchase intention", *Journal of Islamic Marketing*, Vol. 7 No. 1, pp. 59-73.

Merriam-Webster, (2017), "Social Media", https://www.merriamwebster.com/dictionary/social%20media (accessed on 20 July, 2017).

Mohamed, N., Hussein, R., Ahmad Zamzuri, N.H. and Haghshenas, H. (2014), "Insights into individual's online shopping continuance intention", Industrial Management & Data Systems, Vol. 114 No. 9, pp. 1453-1476.

Mollen, A., & Wilson, H. (2010), "Engagement, telepresence and interactivity in online consumer experience: Reconciling scholastic and managerial perspectives", *Journal of Business Research*, Vol. 63, pp. 919-925.

MSI (2010), "2010-2012 research priorities". Available at: http://www.msi.org/pdf/MSI_RP10-12.pdf (accessed on 30 August 2017).

Muniz, A. M., & O'guinn, T. C. (2001), "Brand community", *Journal of consumer research*, Vol. 27 No. 4, pp. 412-432.

Muntinga, D.G., Moorman, M. and Smit, E.G. (2011), "Introducing COBRAs", *International Journal of Advertising*, Vol. 30 No. 1, pp. 13-46.

Murdough, C. (2009), "Social Media measurement: It's not impossible", *Journal of Interactive Advertising*, Vol. 10, pp. 94–99.

Nations, D. (2017), "6 Top Social Shopping Websites You Need to Check Out". Available at: https://www.lifewire.com/top-social-shopping-websites-3486565 (accessed on 30 August 2017).

Netemeyer, R. G., Bearden, W. O., & Sharma, S. (2003), "Scaling procedures: Issues and application". London: Sage.

Netemeyer, R. G., Krishnan, B., Pullig, C., Wang, G., Yagci, M., Dean, D., Ricks, J. & Wirth, F. (2004). Developing and validating measures of facets of consumer-based brand equity. *Journal of Business Research*, *57*(2), 209-224.

Networksolutions (n.d.), "Social Bookmarking – What is it and how can it help promote my site?". Available at: http://www.networksolutions.com/support/social-bookmarking-what-is-it-and-how-can-it-help-promote-my-site/ (accessed on 29 August 2017).

Newberry C. (2017), "Social Listening: What it is, Why You Should Care, and How to Do it Well". Available at: https://blog.hootsuite.com/social-listening-business/ (accessed on 25 August 2017).

Nielsen (2015), "Social Networks Blogs Now Account for One in Every Four and a Half Minutes Online", Available at: http://www.nielsen.com/us/en/insights/news/2010/social-media-accounts-for-22-percent-of-time-online.html (accessed on 24 August 2017).

Nowlin, A. (2013), "30 Tools Social Media Professionals Can't Live Without ". Available at: http://watsonnowlin.com/top-social-tools/ (accessed on 24 August 2017).

Nunnally, J. C. (1978). Psychometric theory. New York: McGraw-Hill.

Nysveen, H. (2005), "Intentions to use mobile services: antecedents and cross-service comparisons", Journal of the Academy of Marketing Science, Vol. 33 No. 3, pp. 330-346.

O'Brien, H. (2010), "The influence of hedonic and utilitarian motivations on user engagement: The case of online shopping experiences", *Interacting with Computers*, Vol. 22, pp. 344-352.

O'Brien, H., & Toms, E. (2008), "What is user engagement? A conceptual framework for defining user engagement with technology", *Journal of American Society for Information Science and Technology*, Vol. 59 No. 6, pp. 938-955.

O'Brien, H., & Toms, E. (2010), "The development and evaluation of a survey to measure user engagement", *Journal of American Society for Information Science and Technology*, Vol. 61 No. 1, pp. 50-69.

Obar, J.A., and Wildman, Steve (2015). "Social Media definition and the governance challenge: An introduction to the special issue", *Telecommunications policy*. Vol. 39 No. 9, pp. 745–750.

Obar, Jonathan A. and Wildman, Steve (2015). "Social Media definition and the governance challenge: An introduction to the special issue", *Telecommunications policy*, Vol. 39 No. 9, pp. 745–750.

Olczak, A. B., & Sobczyk, R. K. (2013), "Brand engagement on Facebook based on mobile phone operators' activity within four European countries", *Studia Ekonomiczne*, No 150. Pp. 97-108.

Oliver, M. (2014), "4 Social Media Coupon Campaigns to Successfully Promote an Ecommerce Business". Available at: http://www.rignite.com/blog/4-social-media-coupon-campaigns-to-successfully-promote-an-ecommerce-business (Accessed on 28 August 2017).

Palazon, M., Sicilia, M., & Lopez, M. (2015). The influence of "Facebook friends" on the intention to join brand pages. *Journal of Product & Brand Management*, Vol. 24 No. 6, pp. 580-594.

Palazon, M., Sicilia, M., & Lopez, M. (2015). The influence of "Facebook friends" on the

Park, C. W., & MacInnis, D. J. (2006), "What's in and what's out: Questions on the boundaries of the attitude construct", *Journal of Consumer Research*, Vol. 33 No. 1, pp. 16-18.

Park, N., Kee, K.F. and Valenzuela, S. (2009), "Being immersed in social networking environment: Facebook groups, uses and gratifications, and social outcomes", *CyberPsychology & Behavior*, Vol. 12 No. 6, pp. 729-734.

Park, W.C., MacInnis, D.J., Priester, J., Eisingerich, A.B. and Iacobucci, D. (2010), "Brand attachment and brand attitude strength: conceptual and empirical differentiation of two critical brand equity driver", Journal of Marketing, Vol. 74 No. 6, pp. 1-17.

Parker, S. (2015), "Top 7 Benefits of Social Recruiting". Available at: http://www.jobcast.net/top-7-benefits-of-social-recruiting/ (accessed on 25 August 2017).

Parker, S. (2016), "A Long List of Instagram Statistics That Marketers Need to Know". Available at: a https://blog.hootsuite.com/instagram-statistics/ (accessed on 24 August 2017).

Patanjali (2018), "Overview", Available at: https://patanjaliayurved.org/companyoverview.html (accessed on 25 April, 2018).

Patterson, P., Yu, T., & De Ruyter, K. (2006), "Understanding consumer engagement in services", In *Advancing theory, maintaining relevance, proceedings of ANZMAC 2006 conference*, Brisbane, pp. 4-6.

Payton, S. (2011), "5 Ways to Promote Sales Through Social Media". Available at: https://www.futuresimple.com/blog/5-ways-to-promote-sales-through-social-media/ (Accessed on 28 August 2017).

PcMag (n.d.), "Definition of: media sharing site". Available at: https://www.pcmag.com/encyclopedia/term/63612/media-sharing-site (accessed om 29 August 2017).

Peng, K. F., Fan, Y. W., & Hsu, T. A. (2004), "Proposing the content perception theory for the online content industry–a structural equation modelling", *Industrial Management & Data Systems*, Vol. 104 No. 6, pp. 469-489.

Peng, K. F., Fan, Y. W., & Hsu, T. A. (2004), "Proposing the content perception theory for the online content industry-a structural equation modelling", *Industrial Management & Data Systems*, Vol. 104 No. 6, pp. 469-489.

Platformed, (n.d.), "From Social Media to the Sharing Economy: the Three Drivers of Business Disruption". Available at: http://platformed.info/social-media-sharing-economy-platforms/ (accessed on 30 August 2017).

Podsakoff, P.M., MacKenzie, S.B., Lee, J.Y. and Podsakoff, N.P. (2003), "Common method biases in behavioral research: a critical review of the literature and recommended remedies", Journal of Applied Psychology, Vol. 88 No. 5, pp. 879-903.

Poorrezaei, M (2016) "Consumer engagement: conceptualisation, measurement and validation", (Doctoral Thesis) University of Salford. Retreived from http://usir.salford.ac.uk/39377/

Porter, C. E. and Donthu, N. (2008), "Cultivating trust and harvesting value in virtual communities", *Management Science*, Vol. 54 No.1, pp. 113-128.

Prahalad, C. K., & Ramaswamy, V. (2004), "Co-creation experiences: The next practice in value creation", *Journal of interactive marketing*, Vol. 18 No. 3, pp. 5-14.

Pressault, A. (2016), "How to Conduct a Competitive Analysis on Social Media: A Quick Guide". Available at: https://blog.hootsuite.com/competitive-analysis-on-social-media/ (accessed on 25 August 2016).

Punyatoya, P. (2015), "Effect of perceived brand environment-friendliness on Indian consumer attitude and purchase intention: An integrated model", *Marketing Intelligence & Planning*, Vol. 33 No. 3, pp. 258-274.

Raacke, J. & Bonds-Raacke, J. (2008), "MySpace and Facebook: Applying the Uses and Gratifications Theory to Exploring Friend-Networking Sites," *CyberPsychology and Behavior*, Vol. 11 No. 2, pp. 169–74.

Raney, A.A., Arpan, L.M., Pashupati, K. and Brill, D.A. (2003), "At the movies, on the web: an investigation of the effects of entertaining and interactive web content on site and brand evaluations", *Journal of Interactive Marketing*, Vol. 17 No. 4, pp. 38-54.

Rappaport, S.D. (2007), "Lessons from online practice: New advertising models", *Journal of Advertising Research*, Vol. 47 No. 2, pp. 135-141.

Rauniar, R., Rawski, G., Yang, J., & Johnson, B. (2014), "Technology acceptance model (TAM) and Social Media usage: an empirical study on Facebook", *Journal of Enterprise Information Management*, Vol. 27 No. 1, pp. 6-30.

Reitz, A. R. (2012), "Online consumer engagement: Understanding the antecedents and outcomes" (Doctoral dissertation, Colorado State University).

Reutemann, J. (2016), "The Rockbridge Series - Part 4", http://patricktrawick.com/blog-1/2016/11/28/the-rockbridge-series-part-4 (accessed on 20 July, 2017).

Rhea (2018), "How digital marketing for FMCG brands will revolutionise the sector" Available at: https://www.socialbeat.in/blog/digital-marketing-for-fmcg/ (accessed on 22 April, 2018).

Samson, E. (2015), "5 Social-Media Opportunities Businesses Mostly Overlook", https://www.entrepreneur.com/article/254046 (accessed on 20 July, 2017).

Sashi, C. M. (2012), "Consumer engagement, buyer-seller relationships, and Social Media". *Management decision*, Vol. 50 No. 2, pp. 253-272.

Schejter, A.M. and Tirosh, N. (2015), ""Seek the meek, seek the just": Social Media and social justice", *Telecommunications Policy*, Vol. 39 No. 9, pp. 796–803.

Schivinski, B., & Dabrowski, D. (2015), "The impact of brand communication on brand equity through Facebook", *Journal of Research in Interactive Marketing*, Vol. 9 No. 1, pp. 31-53.

Schivinski, B., & Dabrowski, D. (2016), "The effect of Social Media communication on consumer perceptions of brands", *Journal of Marketing Communications*, Vol. 22 No. 2, pp. 189-214.

Shi, S., Chen, Y., & Chow, W. S. (2016), "Key values driving continued interaction on brand pages in Social Media: An examination across genders", *Computers in Human Behavior*, Vol. 62, pp. 578-589.

Shi, S., Chen, Y., & Chow, W. S. (2016), "Key values driving continued interaction on brand pages in Social Media: An examination across genders", *Computers in Human Behavior*, Vol. 62, pp. 578-589.

Shimp, T.A. (2010), Advertising, Promotion, and Other Aspects of Integrated Marketing Communications, 8th ed., South-Western, Boston, MA.

Simplus Information Services (2015), "10 stunning facts about Baba Ramdev's Patanjali Ayurved", Available at: https://in.finance.yahoo.com/news/10-stunning-facts-baba-ramdev-053501185.html (accessed on 25 April, 2018).

Singh, S. (2005), "Web 2.0: Connectedness, sharing, and the social", Available at: http://www.siliconindia.com/magazine_articles/Web_20Connectedness_sharing_and_the_soc ial-NLF651991104.html (accessed on 24 August 2017).

Solis, B. (2010), "Engage: The complete guide for brands and businesses to build, cultivate, and measure success in the new web", Hoboken, NJ: John Wiley & Sons, Inc.

Somvanshi K. K. (2017), "Patanjali: Indian FMCG's new Baahubali", Available at: https://economictimes.indiatimes.com/industry/cons-products/fmcg/patanjali-indian-fmcgs-new-baahubali/articleshow/58523174.cms (accessed on 27 April 2018).

Song, S., Yoo, M. (2016), "The role of Social Media during the pre-purchasing stage", *Journal* of Hospitality and Tourism Technology, Vol. 7 No. 1, pp. 84–99.

Spears, N., & Singh, S. N. (2004), "Measuring attitude toward the brand and purchase intentions". *Journal of Current Issues & Research in Advertising*, Vol. 26 No. 2, pp. 53-66.

Statista (2019a), "Leading Countries based on Number of Instagram Users as of January 2019 (in Millions)", Available at: https://www.statista.com/statistics/578364/countries-with-most-instagram-users/ (accessed on 20 February, 2019).

Statista (2019b), "Number of Twitter users in India from 2012 to 2019 (in millions)", Available at: https://www.statista.com/statistics/381832/twitter-users-india/ (accessed on 20 February, 2019).

Statista (2019c), "Leading Country based on number of Facebook users as of January 2019 (in millions)", Available at: https://www.statista.com/statistics/268136/top-15-countries-based-on-number-of-facebook-users/ (accessed on 20 February, 2019).

Statista (2019d), "Number of Facebook users across India as of January 2018, by age and gender (in millions)". Available at: https://www.statista.com/statistics/717615/ India-number-of-facebook-users-by-age-and-gender/ (accessed on 12 February, 2019).

Statista (2019e), Number of Social Media users worldwide from 2010 to 2021 (in billions)". Available at: https://www.statista.com/statistics/278414/number-of-worldwide-socialnetwork-users/ (accessed on 12 February, 2019).

Sterling, G. (2016), "Nearly 80 percent of Social Media time now spent on mobile devices", Available at: http://marketingland.com/facebook-usage-accounts-1-5-minutes-spent-mobile-171561 (accessed on 24 August 2017).

Stewart, D.W. and Pavlou, P.A. (2002), "From consumer response to active consumer: measuring the effectiveness of interactive media", Journal of the Academy of Marketing Science, Vol. 30 No. 4, pp. 376-396.

Sweeney, J.C., Soutar, G.N. and Mazzarol, T. (2008), "Factors influencing word of mouth effectiveness: receiver perspectives", European Journal of Marketing, Vol. 42 Nos 3/4, pp. 344-364.

Syncapse. (2013). *Why consumers become brand fans*. Available at: http://freshbridge.nl/blog/ wp-content/uploads/2013/09/Syncapse_Report.Why_Consumers_Become_Brand_Fans_June. pdf

Tabachnick, B.G. and Fidell, L.S. (2013) Using Multivariate Statistics. Pearson, Boston.

Tafesse, W. (2015), "Content strategies and audience response on Facebook brand pages", Marketing Intelligence & Planning, Vol. 33 No. 6, pp. 927 – 944.

Takaoka, K. (2016), "10 Types of Social Media Platform", http://sbtvirtual.ca/10-types-social-media-platform/ (accessed on 20 July, 2017).

Tang, A. (2017), "How much time do people spend on Social Media?", Available at: http://www.marketing-interactive.com/how-much-time-do-people-spend-on-social-media-infographic/ (accessed on 24 August 2017).

Tarhini, A., Hone, K., & Liu, X. (2014), "Measuring the moderating effect of gender and age on e-learning acceptance in England: A structural equation modeling approach for an extended technology acceptance model", *Journal of Educational Computing Research*, Vol. 51 No. 2, pp. 163-184.

Taylor, D.G., Lewin, J.E. and Strutton, D. (2011), "Friends, fans, and followers: do ads work on social networks? How gender and age shape receptivity", *Journal of Advertising Research*, Vol. 51 No. 1, pp. 258-274.

thedailyrecords, (2019), "The 10 Best FMCG Companies in India". Available at: http://www.thedailyrecords.com/2018-2019-2020-2021/world-famous-top-10-list/india/best-fmcg-companies-india-biggest/8504/ (accessed on 13 January, 2018).

Thompson, S. A. and Sinha, R. K. (2008), "Brand communities and new product adoption: the influence and limits of oppositional loyalty". *Journal of Marketing*, Vol. 72 No. 6, pp. 65-80.

Thomson, M., MacInnis, D.J. and Park, C.W. (2005), "The ties that bind: measuring the strength of consumers' emotional attachments to brands", *Journal of Consumer Psychology*, Vol. 15 No. 1, pp. 77-91.

Tickle, M., Adebanjo, D., & Michaelides, Z. (2011), "Developmental approaches to B2B virtual communities", *Technovation*, Vol. 31 No. 7, pp. 296–308.

Tsai, H-T., Huang, H-C. and Chiu, Y-L. 2012, "Brand community participation in Taiwan: Examining roles of individual, group, and relationship-level antecedents", *Journal of Business Research*, Vol. 65 No. 5, pp. 676-684.

Uribe, F., Rialp, J., & Llonch, J. (2010), "El uso de las redes sociales digitales como herramienta de marketing-Un estudio de casos", *Libro de Actas del XXII Congreso Nacional de Marketing*, AEMARK 2010.

Uslaner, E. M. (2000). Social capital and the net", *Communications of the ACM*, Vol. 43 No. 12, pp. 60-64.

Van Doorn, J., Lemon, K. N., Mittal, V., Nass, S., Pick, D., Pirner, P., & Verhoef, P. C. (2010), "Consumer engagement behavior: Theoretical foundations and research directions", *Journal of service research*, Vol. 13 No. 3, pp. 253-266.

Van Slyke, C., Ilie, V., Lou, H., & Stafford, T. (2007), "Perceived critical mass and the adoption of a communication technology", *European Journal of Information Systems*, Vol. 16 No.3, pp. 270-283.

Venkatesh, V. and Davis, F.D. (2000), "A theoretical extension of the technology acceptance model: four longitudinal field studies", Management Science, Vol. 46 No. 2, pp. 186-204.

Venkatesh, V., Morris, M.G., Davis, G.B. and Davis, F.D. (2003), "User acceptance of information technology: toward a unified view", *MIS Quarterly*, Vol. 27 No. 3, pp. 425-78.

Vision Critical (2013), "From Social to Sale: 8 Questions to Ask Your Consumers" [White Paper], Vancouver, BC, Canada, http://www.visioncritical.com/social2sale.

Vivek, S. D., Beatty, S. E., & Morgan, R. M. (2012), "Consumer engagement: Exploring consumer relationships beyond purchase", *Journal of Marketing Theory and Practice*, Vol. 20 No. 2, pp. 122-146.

Von Loewenfeld, F. (2006), "Brand communities", Ph.D. Thesis, Johannes Gutenberg University, Mainz, Germany.

Vorvoreanu, M. (2009), "Perceptions of corporations on Facebook: An analysis of Facebook social norms", *Journal of New Communications Research*, Vol. 4 No. 1, pp. 67-86.

Walker, L. (2016), "Psst! The Secret List of Anonymous Social Networks". Available at: https://www.lifewire.com/list-of-anonymous-social-networks-2654864 (accessed on 30 August 2017).

Walmsley, A. (2010), "New media needs new PR", Available at: http://www.campaignlive.co.uk/article/andrew-walmsley-digital-new-media-needs-new-pr/985566 (accessed on 24 August 2017).

Wang, A. (2006), "Advertising engagement: A driver of message involvement on message effects", *Journal of Advertising Research*, Vol. 46, pp. 355-368.

Wang, R. (2011), "Moving from transaction to engagement". Available at: http://blogs.hbr.org/cs/2011/10/moving_from_transaction_to_eng.html (accessed on 28 August 2017).

Webdunia (2015), "बाबा रामदेव की हर्बल फैक्टरी अब नेपाल में", Available at: http://hindi.webdunia.com/international-hindi-news/baba-ramdev-herbal-factory-herbalproducts-nepal-kathmandu-115041800030_1.html (accessed on 25 April 2018).

Weber, L. (2007), "Marketing to the Social Web. How digital consumer communities build your business". Hoboken, NJ: Wiley.

Weman, E. (2011), "Consumer motivations to join a brand community on Facebook", *Master* of Marketing Thesis, Hanken School of Economics, Helsinki.

Wiegandt, P. (2009), "Value creation of firm-established brand communities". Wiesbaden, Germany: Gabler.

Wiertz, C. and de Ruyter, K. (2007), "Beyond the call of duty: why consumers contribute to firm-hosted commercial online communities", Organization Studies, Vol. 28 No. 3, pp. 347-76.

Wilcox, K. and Stephen, A.T. (2013), "Are close friends the enemy? Online social networks, self-esteem, and self-control", Journal of Consumer Research, Vol. 40 No. 1, pp. 90-104.

Wilimzig, B. J. (2011), "Online communities: Influence on members brand loyalty and purchase intent". *Retrieved from Google Scholar*.

Wilson, D. J. (1990), "Science, community, and the transformation of American philosophy". Chicago: University of Chicago Press.

Wirtz, B. W., Piehler, R., & Ullrich, S. (2013), "Determinants of Social Media website attractiveness", *Journal of Electronic Commerce Research*, Vol. 14 No. 1, pp. 11.

Wirtz, J., Den Ambtman, A., Bloemer, J., Horváth, C., Ramaseshan, B., Van De Klundert, J., & Kandampully, J. (2013), "Managing brands and consumer engagement in online brand communities", *Journal of Service Management*, Vol. 24 No. 3, pp. 223-244.

Woisetschläger, D. M., Hartleb, V. and Blut, M. (2008), "How to make brand communities work: Antecedents and consequences of consumer participation", *Journal of Relationship Marketing*, Vol. 7 No. 3, pp. 237-256.

Wong, C.H., Tan, G.W.H., Loke, S.P. and Ooi, K.B. (2014), "Mobile TV: a new form of entertainment?", Industrial Management & Data Systems, Vol. 114 No. 7, pp. 1050-1067.

Wortham, J. (2007), "After 10 Years of Blogs, the Future's Brighter Than Ever". Avalable at: https://www.wired.com/2007/12/after-10-years-of-blogs-the-futures-brighter-than-ever/ (accessed on 30 August 2017).

wpmu, (2017), "ICT and Social Media: Definitional Issues and the Relationship" http://wpmu.mah.se/nmict11group1/2011/10/30/2-ict-and-social-media-definitional-issues-and-the-relationship/ (accessed on 20 August, 2017).

Wu, J.H. and Wang, S.C. (2005), "What drives mobile commerce? An empirical evaluation of the revised technology acceptancemodel", *Information & Management*, Vol. 42 No. 5, pp. 719-729.

Xiang, Z. and Gretzel, U. (2010), "Role of Social Media in online travel information search", *Tourism Management*, Vol. 31 No. 2, pp. 179-188.

Xu, D.H., Huo, B.F. and Sun, L.Y. (2014), "Relationships between intra-organizational resources, supply chain integration and business performance", Industrial Management & Data Systems, Vol. 114 No. 8, pp. 1186-1206.

Yu, J., Ha, I., Choi, M. and Rho, J. (2005), "Extending the TAM for a t-commerce", *Information & Management*, Vol. 42 No. 7, pp. 965-76.

Zaglia, M. E. (2013), "Brand communities embedded in social networks", *Journal of Business Research*, Vol. 66 No. 2, pp. 216-224.

Zhou, T. (2014), "Understanding the determinants of mobile payment continuance usage", Industrial Management & Data Systems, Vol. 114 No. 6, pp. 936-948.

Zhou, Z., Zhang, Q., Su, C., & Zhou, N. (2012), "How do brand communities generate brand relationships? Intermediate mechanisms", *Journal of Business Research*, Vol. 65 No. 7, pp. 890-894.

Zorzini, C. (2015), "3 Ways Ecommerce Companies Should Use Social Media for Marketing". Available at: https://ecommerce-platforms.com/articles/3-ways-ecommerce-companiesshould-use-social-media-for-marketing (Accessed on 26 August 2017).

Appendix A

Survey Questionnaire

Consumer engagement on Facebook Brand Pages

Dear Respondent,

We are conducting a study to understand the online consumer engagement by FMCG brands owned official Facebook Brand Pages. Nowadays Major FMCG brands (Pepsi, Cadbury, Nestle, Dabur etc.) have created brand pages on Facebook and they engage with their consumers by posting images, videos regarding useful information about their products. These Facebook Pages are also used for promotion and to communicate with consumers. This study focuses on the understanding what encourages consumers to like a brand page and whether the promotion on these brand pages effect their intentions to purchase the brand.

Following Question are to be answered on a scale of 1 to 7 with

- 1 Strongly Disagree
- 2 Disagree
- 3 Somewhat Disgree
- 4 Neutral
- 5 Somewhat Agree
- 6 Agree
- 7 Strongly Agree

* 5	э.	-	2	a ni	ie.	-	d
	N		ч	13	Ł		-

1. Gender *

Mark only one oval.

T	7	Male
Ē	5	Female

2. Age *

Mark only one oval.

C) Beow 20
\subset) 21-25
\square) 26-30
\square) 31-35
$\overline{\mathbb{C}}$) 36-40
\bigcirc) above 40

 Annual Family Income * Mark only one oval.

C	Below 5,00,000
Ć	5,00,00-7,50,000
Ċ	7,50,000-10,00,000
Ć	Above 10,00,000

4. Occupation *

Mark only one oval.

\bigcirc	Student
\bigcirc	Govt. / Private sector employee
\bigcirc	Self-employed professional
$\overline{\bigcirc}$	Entrepreneur/Trader
0	Other:

1 2 3 4 5 6 7 Strongly Disagree 0 0 0 Strongly Agree I am attracted to this brand.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disagree 0 0 0 Strongly Agree 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disagree 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disagree 0 0 Strongly Agree Strongly Agree Mark only one oval. Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 Strongly Agree Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 Strongly Agree Strongly Agree I 1 2 3 4 5 6 7 Strongly Disgree 0 <th>Mark only one ova</th> <th>14:</th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th> <th></th>	Mark only one ova	1 4:							
About 15 hours/week Nore than 15 hours/week Thave lot of affection for this brand.* Mark only one oval. T 2 3 4 5 6 7 Strongly Disagree T 1 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 2 3 4 5 6 7 Strongly Disagree T 3 4 5 6 7 Strongly Disagree T 4 4 5 6 7 Strongly Disagree T 4 5 6 7 Strongly Disagree T 5 6 7 Strongly Disagree T 5 6	Less than 5	5 hours/w	eek						
More than 15 hours/week a Have lot of affection for this brand.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree <	About 10 h	ours/wee	k						
More than 15 hours/week I have lot of affection for this brand.* Mark only one oval.	About 15 h	ours/wee	k						
Mark only one oval. 1 2 3 4 5 8 7 Strongly Disagree 0 <th>More than</th> <th>15 hours</th> <th>week</th> <th></th> <th></th> <th></th> <th>ал. С</th> <th></th> <th></th>	More than	15 hours	week				ал. С		
1 2 3 4 5 6 7 Strongly Disagree 0 0 0 0 Strongly Agree 7. I am attracted to this brand.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disagree 0 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disagree 0 0 Strongly Agree 3 4 6 8 7 3. I remain involved with a brand on Facebook due to the affection I have for the brand.* Mark only one oval. 1 2 3 4 6 8 7 Strongly Disgree 0 0 Strongly Agree 0 Strongly Agree 3 1 2 3 4 5 6 7 1 2 3 4 5 6 7 Strongly Agree 3 5 5 7 5 5 5 7 5 5 5 7 5 5 5 7 5 5 6<	에너지 않는 성격 방송 전 가지 않	gen 230. (this bra	nd.*					
Strongly Disagree Strongly Agree 1 2 3 4 5 6 7 1 2 3 4 5 6 7 Strongly Disagree Image: Strongly Agree Strongly Agree Strongly Agree 3. I remain involved with a brand on Facebook due to the affection I have for the brand.* Mark only one oval. Image: Strongly Agree 3. I remain involved with a brand on Facebook due to the affection I have for the brand.* Mark only one oval. Image: Strongly Agree 3. I remain involved with a brand on Facebook due to the affection I have for the brand.* Mark only one oval. Image: Strongly Agree 3. I think, brand pages provide a convenient platform for the customers to voice their cor and suggestions to the brand.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree Image: Strongly Agree Image: Strongly Agree Image: Strongly Agree Image: Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree Image: Strongly Agree Image: Strongly Agree Image: Strongly Agree 1 2 3 4 5 6	Mark only one ova	L,							
1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 6 7 Strongly Disagree 1 2 3 4 5 8 7 Strongly Disgree 1 2 3 4 5 8 7 Strongly Disgree 1 2 3 4 5 8 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Str		1	2	3	4	5	6	7	
Mark only one oval. 1 2 3 4 5 6 7 Strongly Disagree Image: Strongly Disagree Strongly Disagree Strongly Disagree Strongly Disagree Strongly Disagree 1 2 3 4 5 6 7 Strongly Disgree Image: Disagree Image: Disagree Strongly Disgree Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree Image: Disgree Strongly Disgree Strongly Disgree Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree Image: Disgree Strongly Disgree Strongly Disgree Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree Image: Disgree Strongly Disgree Strongly Disgree Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree Image: Disgree Strongly Disgree Strongly Disgree Strongly Agree 1 2	Strongly Disagree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
1 2 3 4 5 6 7 Strongly Disagree 0 0 1 2 3 4 5 6 7 3. Iremain involved with a brand on Facebook due to the affection I have for the brand. * Mark only one oval. 1 2 3 4 5 8 7 1 2 3 4 5 8 7 Strongly Disgree 0 0 0 Strongly Agree 2. I think, brand pages provide a convenient platform for the customers to voice their cor and suggestions to the brand. * Mark only one oval. 1 2 3 4 5 8 7 Strongly Disgree 0 0 Strongly Agree Strongly Disgree Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 Strongly Agree Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 Strongly Agree Strongly Agree 1 2 3	. I am attracted to t	this bran	d.*						
Strongly Disagree Strongly Agree 3. I remain involved with a brand on Facebook due to the affection I have for the brand.* Mark only one oval. 1 2 3 4 5 8 7 Strongly Disgree 1 2 3 4 5 8 7 Strongly Disgree 1 2 3 4 5 8 7 Strongly Disgree 1 2 3 4 5 8 7 Ithink, brand pages provide a convenient platform for the customers to voice their cor and suggestions to the brand.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 <td>Mark only one ova</td> <td>I,</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Mark only one ova	I,							
3. I remain involved with a brand on Facebook due to the affection I have for the brand.* 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Ark only one oval. 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2 3 4 5 6 7 Strongly Disgree 1 2<		1	2	3	4	5	6	7	
Mark only one oval. 1 2 3 4 5 8 7 Strongly Disgree 0 0 0 0 Strongly Agree 2. I think, brand pages provide a convenient platform for the customers to voice their cor and suggestions to the brand. * Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 0. I think it is possible to communicate instantly with brands through brand pages without time and space boundaries. * Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 2 3 4 5 6 7 I think, brand pages make it easy to connect and communicate with the brand	Strongly Disagree	0	0	0	0	0	0	0	Strongly Agree
Mark only one oval. 1 2 3 4 5 8 7 Strongly Disgree 0 0 0 0 Strongly Agree 1 1 2 3 4 5 8 7 Strongly Disgree 0 0 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Disgree Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Str									
Strongly Disgree Strongly Agree I think, brand pages provide a convenient platform for the customers to voice their corrand suggestions to the brand.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree Image: Constraint of the customers to voice their corrand suggestions to the brand.* Mark only one oval. Strongly Disgree Strongly Agree I think it is possible to communicate instantly with brands through brand pages without time and space boundaries.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree Image: Constraint of the brand.* Strongly Agree Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree Image: Constraint of the brand.* Strongly Agree Image: Constraint of the brand.* 1 2 3 4 5 6 7 Strongly Disgree Image: Constraint of the brand.* Image: Constraint of the brand.* 1 2 3 4 5 6 7				T due b		to uic	uncouo		for the brund.
2. Ithink, brand pages provide a convenient platform for the customers to voice their contand suggestions to the brand. * Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 0. Ithink it is possible to communicate instantly with brands through brand pages without time and space boundaries. * Mark only one oval. 1 2 3 4 5 6 7 1 2 3 4 5 6 7 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 Strongly Agree Strongly Agree 1 1 2 3 4 5 6 7 1 2 3 4 5 6 7		1	2	3	4	5	6	7	
and suggestions to the brand.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree Image: Communicate instantly with brands through brand pages without time and space boundaries.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree Image: Communicate instantly with brands through brand pages without time and space boundaries.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree Image: Communicate and communicate with the brand.* Strongly Agree Image: Communicate and communicate with the brand.* 1 2 3 4 5 6 7									
Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree O O Strongly Agree 1 1 2 3 4 5 6 7 1 1 2 3 4 5 6 7 1 2 3 4 5 6 7 Strongly Disgree O O Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree O O Strongly Agree Strongly Agree 1 1 2 3 4 5 6 7 Strongly Disgree O O Strongly Agree Strongly Agree 1 2 3 4 5 6 7	Strongly Disgree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
1 2 3 4 5 6 7 Strongly Disgree 0 0 0 0 Strongly Agree 1 1 2 3 4 5 6 7 Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 0 Strongly Agree 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree 1 1 2 3 4 5 6 7 Strongly Disgree 0 0 0 Strongly Agree Strongly Agree 1 1 2 3 4 5 6 7). I think, brand pag			nvenier	Ont platfo	m for t	C the cust	O omers to	
Strongly Disgree Strongly Agree I think it is possible to communicate instantly with brands through brand pages without time and space boundaries.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree Image: Communicate and communicate with the brand.* Mark only one oval. Strongly Disgree Image: Communicate and communicate with the brand.* Mark only one oval. 1 2 3 4 5 6 7). I think, brand pag and suggestions	to the br		nvenier	nt platfo	rm for t	he cust	omers to	
1 1 2 3 4 5 6 7 1 2 3 4 5 6 7 Strongly Disgree Image: Construction of the second of t). I think, brand pag and suggestions	to the br		nvenier	ont platfo	rm for t) he cust	omers to	
time and space boundaries.* Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree O O O O Strongly Agree 1 1 2 3 4 5 6 7 1 1 2 3 4 5 6 7). I think, brand pag and suggestions	to the br I.	and. *		877.) -				
Mark only one oval. 1 2 3 4 5 6 7 Strongly Disgree O O O Strongly Agree 1 1 2 3 4 5 6 7	 I think, brand pag and suggestions Mark only one ova 	to the br I.	and. *		877.) -				o voice their cor
1 2 3 4 5 6 7 Strongly Disgree O O O O Strongly Agree 1. I think, brand pages make it easy to connect and communicate with the brand. * Mark only one oval. 1 2 3 4 5 6 7	 I think, brand pag and suggestions Mark only one ova Strongly Disgree I think it is possib 	to the br / 1	2	3	4	5	6	7	o voice their cor Strongly Agree
Strongly Disgree O Strongly Agree 1. I think, brand pages make it easy to connect and communicate with the brand. * Mark only one oval. 1 2 3 4 5 6 7	 I think, brand pag and suggestions Mark only one ova Strongly Disgree I think it is possib time and space b 	to the br / 1 ole to co oundarie	2	3	4	5	6	7	o voice their con Strongly Agree
1. I think, brand pages make it easy to connect and communicate with the brand. * Mark only one oval. 1 2 3 4 5 6 7	 I think, brand pag and suggestions Mark only one ova Strongly Disgree I think it is possib time and space b 	to the br / 1 ole to co oundarie	2	3	4	5	6	7	o voice their con Strongly Agree
Mark only one oval.	 I think, brand pag and suggestions Mark only one ova Strongly Disgree I think it is possib time and space b 	to the br / 1 ole to co oundarie	2	3	4	5	6	7	o voice their con Strongly Agree
	 I think, brand pag and suggestions Mark only one ova Strongly Disgree I think it is possib time and space b Mark only one ova 	to the br / 1 ole to co oundarie	2	3	4	5	6	7	o voice their cor Strongly Agree
	 I think, brand pag and suggestions Mark only one ova Strongly Disgree I think it is possib time and space b Mark only one ova Strongly Disgree I think, brand pag 	to the bi 1 1 0 1 0 1 0 1 1 1 1 1 1 1 1 1 1 1 1	2 	3 cate inst 3	4	5 ith bran 5	6 ids thro 6	7 ugh brai 7	Strongly Agree
	 I think, brand pag and suggestions Mark only one ova Strongly Disgree I think it is possib time and space b Mark only one ova Strongly Disgree I think, brand pag 	to the br 1 1 0 1 0 1 0 1 0 1 1 1 1 1 1 1 1 1 1	2 mmunic es. * 2 it easy	3 ate inst 3 to conr	4 antly w 4	5 ith bran 5	6 nds thro 6	7 ugh bra 7 with the	Strongly Agree

	1	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
I like the content	posted	(Images	Video	s Articl	es) on t	he bran	d page	*
Mark only one ova	5. A.	ges					- p-8	
	1	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
Games and / or vi Mark only one ova		eated b	y brand	s are fu	in and k	eeps m	e engag	ed *
	્ય	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	0	$\overline{\circ}$	0	0	$\overline{\bigcirc}$	0	Strongly Agree
I think the enterta	inina co	ontent (i	images	videos	article	a) on br	nea bae	e nositively infl
brand's image.*		vitterit (i	inages,	VIGEOS	, arucie	5/ 011 01	and pag	e positively init
Mark only one ova	6							
	1	2	3	4	5	6	7	
	10 T 1	ables m	e to get		cted and) stay ir	touch	Strongly Agree
	10 T 1	ables m	e to get	t conner	Cted and	d stay ir 8	T touch a	
Facebook brand (L		12	4		223		
Facebook brand j Mark only one ova	1	2	3	4	5	8	7	with brand. * Strongly Agree
Facebook brand ; Mark only one ova Strongly Disgree Facebook brand ;	1	2	3	4	5	8	7	with brand. * Strongly Agree
Facebook brand ; Mark only one ova Strongly Disgree Facebook brand ;	1 Object market	2	3	4	5 Stay inf	6 Ormed	7	with brand. * Strongly Agree
Facebook brand ; Mark only one ova Strongly Disgree Facebook brand ; Mark only one ova	1 Dage ma 1 1 n about ny store	2 skes it e 2 the dise	3 counts :	4 or me to 4	5 stay inf 5	e ormed e	7 about th 7	with brand. * Strongly Agree e brand and its Strongly Agree
Facebook brand (Mark only one ova Strongly Disgree Facebook brand (Mark only one ova Strongly Disgree Latest informatio without visiting a	1 Dage ma 1 1 n about ny store	2 skes it e 2 the dise	3 counts ;	4 or me to 4	5 stay inf 5	e ormed e	7 about th 7	with brand. * Strongly Agree e brand and its Strongly Agree
Facebook brand (Mark only one ova Strongly Disgree Facebook brand (Mark only one ova Strongly Disgree Latest informatio without visiting a	1 Dage ma 1 1 n about 1	2 akes it e 2 the disk es and/o	3 counts or shops	4 or me to 4	5 stay inf 5 motion:	6 ormed 8 0	7 about th 7 obtaine	with brand. * Strongly Agree e brand and its Strongly Agree
Facebook brand (Mark only one ova Strongly Disgree Facebook brand (Mark only one ova Strongly Disgree Latest informatio without visiting a Mark only one ova	1 coage ma coage ma 1 1 1 1 1 1 coages o	2 akes it e 2 the dise es and/o	3 casier for 3 counts : or shops 3	4 or me to 4 on and pro	5 stay inf 5 motion: 5	6 ormed 6 s can be	7 about th 7 obtaine 7	with brand. * Strongly Agree e brand and its Strongly Agree ed from brand p Strongly Agree
Facebook brand ; Mark only one ova Strongly Disgree Facebook brand ; Mark only one ova Strongly Disgree Latest informatio without visiting a Mark only one ova Strongly Disgree Following brand ; etc.). *	1 coage ma coage ma 1 1 1 1 1 1 coages o	2 akes it e 2 the dise es and/o	3 casier for 3 counts : or shops 3	4 or me to 4 on and pro	5 stay inf 5 motion: 5	6 ormed 6 s can be	7 about th 7 obtaine 7	with brand. * Strongly Agree e brand and its Strongly Agree ed from brand p Strongly Agree

20. I think that the brand page provides reliable information.*

Mark only one oval.

	1	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
I think brand page between brands a Mark only one ova	and con			ation s	ource th	nat enab	oles a tra	insparent chann
main only one ora	E.							
	1	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
I think that the inf Mark only one ova		on offer	ed by m	y brand	l page is	s sincer	e and ho	onest.*
	1	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
People who influe Mark only one ova	Concerned as	r behavi	our thir	ık that l	should	Engage	e with th	e brand page. *
	्य	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	0	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc	Strongly Agree
	a	2	3	4	5	6	7	
Strongly Disgree	0	\bigcirc	\bigcirc	0	0	\bigcirc	\bigcirc	Strongly Agree
People in my env brand page. * Mark only one ova		nt (frien 2		colleagu 4			hat I shi 7	ould Engage with
Strongly Disgree	0	0	\square	\cap	0		\cap	Strongly Agree
Many people I co Mark only one ova		ate with	n, are ac	tive on	brand j	pages. 1		1000
	а	2	3	4	5	6	7	
Strongly Disgree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
The people I com Mark only one ova		te with,	will con	itinue to	o use br	ad page	es in the	future.*
	1	2	3	4	5	6	7	
Strongly Disgree	0	0		0	0	\cap	0	Strongly Agree

1	2	3	4	5	6	7			
Bad	\bigcirc	0	\bigcirc	\bigcirc	0	0	Good	Ē	
	Ce	63 - 63							
* Mark only one o	wal.								
	1	2	3	4	5	6	7		
Unfavourable	0	C		C		$) \subset$		Fa	vourable
). *									
Mark only one o	wal.								
1		2	3	4	5	6	7		
Unlikable) (\supset	0	\supset	0	\bigcirc	0	Likable	• • •
.*									
Mark only one o	wal.								
	1	2	3	4	5	6	7		
Unappealing	\square	0	0	C			0	App	pealing
Specifik Arsonal Cele			· · · · · · · · · · · · · · · · · · ·						
* Mark only one o	wal.								
Mark only one o		2	3	2	5	6	7		
Mark only one o	oval. 1	2	3	4	5	6	7	1000	
Mark only one o		2	3	4	5	6	7	Plea	sant
Mark only one o	1	0	0	0	0	0	7	Plea	sant
Mark only one o	1	0	0	0	0	0	7	Plea	sant
Mark only one o	1	0	nore abo	0	o brand	0	0	Plea 7	sant
Mark only one o	1	C)	nore abo	out this	o brand	•	0	17.632	sant Strongly Agr
Mark only one of Unpleasant (3. I am interested Mark only one of Strongly Disagn	1 Din lea wal.	O rning r 1	nore abo	out this	s brand	*	6	17.632	
Mark only one of Unpleasant (3. I am interested Mark only one of Strongly Disagn	1 I in lea wal.	O rning r 1	nore abo	out this	s brand	*	6	17.632	
Mark only one of Unpleasant (3. I am interested Mark only one of Strongly Disagn	1 I in lea wal.	O rning r 1	nore abo	out this	s brand	*	6	17.632	
Mark only one of Unpleasant (3. I am interested Mark only one of Strongly Disagn	1 l in lea wal. rested wal.	1 I in oth	2 er produ	3 out this	s brand 4	5 • this br	6	7	
Mark only one of Unpleasant (a. I am interested Mark only one of Strongly Disagn Mark only one of Strongly Disagn	1 lin lea wal. ee (wal.	I in oth	nore abo 2 er produ 2	aut this 3 cts off 3	s brand 4	5 • this br	6	7	Strongly Agr
Unpleasant (3. I am interested Mark only one o Strongly Disagn 4. I would be inter Mark only one o	1 l in lea oval. ee (oval. ee (ee (I in oth	nore abo 2 er produ 2	aut this 3 cts off 3	s brand 4	5 • this br	6	7	Strongly Agr
Mark only one of Unpleasant (3. I am interested Mark only one of Strongly Disagn 4. I would be inter Mark only one of Strongly Disagn 5. I like to visit the	1 l in lea oval. ee (oval. ee (ee (I in oth	nore abo 2 er produ 2	aut this 3 cts off 3	s brand 4	5 • this br	6	7	Strongly Agr

38. I will definitely purchase product in the future, for a brand that I follow on Facebook * Mark only one oval.

Strongly Disagree	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Strongly Agree
7. I will continue to I Mark only one ova	10. 20	orand th	at I folk	ow on F	aceboo	k*		
	3	2	3	4	5	6	7	
Strongly Disagree	0	0	0	\bigcirc	\bigcirc	\bigcirc	0	2012/00/77/70/27/202
Strongly Disagree 8. I will refer the bra Mark only one ovai		friends	/family/	Colleag	oues, tha	C I follo	w on Fa	Strongly Agree
3. I will refer the bra		friends 2	/family/ 3	Colleag	ues, tha	t I follo	w on Fa	



Appendix B

Mahalanobis Distance (D²) Values

Observation number	Mahalanobis d-squared	p1	p2
63	85.548	0	0.001
330	84.876	0	0
327	84.516	0	0
435	82.147	0	0
446	81.585	0	0
289	80.808	0	0
42	80.038	0	0
114	77.98	0	0
111	76.941	0	0
112	76.083	0	0
133	76.052	0	0
383	74.449	0	0
172	74.291	0	0
28	73.165	0	0
442	72.473	0	0
140	72.301	0	0
313	72.166	0	0
486	71.948	0	0
5	71.789	0	0
170	70.158	0	0
58	69.112	0	0
12	69.105	0	0
168	69.038	0	0
394	68.613	0	0
106	68.218	0	0
142	67.342	0	0
154	67.127	0	0
10	67.092	0	0
217	66.165	0.001	0
193	66.052	0.001	0
348	65.216	0.001	0
447	64.463	0.001	0
264	64.446	0.001	0
148	63.406	0.001	0
414	63.338	0.001	0

223	62.53	0.001	0
382	62.446	0.001	0
188	62.247	0.002	0
489	61.86	0.002	0
136	61.817	0.002	0
85	61.794	0.002	0
165	61.544	0.002	0
41	61.298	0.002	0
210	60.388	0.003	0
244	59.219	0.003	0
456	58.932	0.004	0
281	58.69	0.004	0
46	58.505	0.004	0
187	58.38	0.004	0
389	58.011	0.005	0
451	57.943	0.005	0
20	57.509	0.005	0
407	56.966	0.006	0
98	56.674	0.006	0
498	56.566	0.007	0
247	56.386	0.007	0
251	56.223	0.007	0
102	55.713	0.008	0
131	55.537	0.008	0
381	55.281	0.009	0
145	55.243	0.009	0
390	55.17	0.009	0
119	54.898	0.01	0
249	53.919	0.012	0
448	53.759	0.013	0
141	53.659	0.013	0
395	53.203	0.014	0
419	53.1	0.015	0
81	52.686	0.016	0
84	52.591	0.017	0
411	52.589	0.017	0
379	52.494	0.017	0
146	52.315	0.018	0
105	51.863	0.019	0
340	51.607	0.021	0

406 51.399 0.022 0 246 51.393 0.022 0 317 51.385 0.022 0 325 51.359 0.022 0 403 51.136 0.023 0 197 51.084 0.023 0 40 51.046 0.023 0 302 50.009 0.029 0 263 49.621 0.032 0 129 49.603 0.032 0 99 49.559 0.032 0	
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	
32551.3590.022040351.1360.023019751.0840.02304051.0460.023030250.0090.029026349.6210.032012949.6030.03209949.5590.0320	
40351.1360.023019751.0840.02304051.0460.023030250.0090.029026349.6210.032012949.6030.03209949.5590.0320	
19751.0840.02304051.0460.023030250.0090.029026349.6210.032012949.6030.03209949.5590.0320	
4051.0460.023030250.0090.029026349.6210.032012949.6030.03209949.5590.0320	
30250.0090.029026349.6210.032012949.6030.03209949.5590.0320	
26349.6210.032012949.6030.03209949.5590.0320	
12949.6030.03209949.5590.0320	
99 49.559 0.032 0	
10.7	
186 49.449 0.033 0	
55 49.298 0.034 0	
11 48.947 0.036 0	
173 48.669 0.039 0	
369 48.499 0.04 0	
59 48.478 0.04 0	
232 48.144 0.043 0	
100 47.841 0.046 0	
24 47.814 0.046 0	
227 47.746 0.047 0	
13 47.698 0.047 0	

Appendix C

Curriculum Vitae

Name: Anurag Tiruwa

Date of Birth: 20th October 1988

Educational Qualifications:

M.B.A. (Marketing Management and Supply Chain Management), Delhi School of Management, Delhi Technological University, 2013, CGPA -7.41/10.0

B.Tech. (Electronic Instrumentation and Control), Govt. Engineering College Ajmer, Rajasthan Technical University, 2010, Percentage - 64.72

Professional Experience:

July 2018 onwards: Assistant Professor at Rukmini Devi Institute of Advanced Studies

Journal Papers

• Tiruwa, A., Yadav, R. and Suri, P. K., (2018) "Modelling Facebook usage for collaborative learning in higher education", *Journal of Applied Research in Higher Education* (Emerald), Vol. 10 No. 3, pp. 357-379.

• Tiruwa, A., Yadav, R. and Suri, P. K. (2018), "Moderating Effects of Age, Income and Internet Usage on Online Brand Community Induced Purchase Intention", *Journal of advances in Management Research* (Emerald), https://doi.org/10.1108/JAMR-04-2017-0043.

· Yadav, R., Tiruwa, A. and Suri, P. K. (2017), "Internet Based Learning (IBL) in Higher Education: A literature Review", *Journal of International Education in Business* (Emerald), Vol. 10 No. 2, pp. 102-129.

• Tiruwa, A., Yadav, R. and Suri, P. K. (2016), "An Exploration of Online Brand Community (OBC) Engagement and Consumer's Intention to Purchase", *Journal of Indian Business Research* (Emerald), Vol. 8 No. 4, pp. 295-314.

Conference Papers

• Tiruwa, A. and Yadav, R. (2017), "Understanding Consumers' Inclination Towards Pepsi® Facebook Brand Page (FBP) Posts: A Sentiment Analysis Approach", Proceedings of GLOGIFT 17 December 11-13, Delhi School of Management, Delhi Technological University, Delhi, pp. 548-559.

• Tiruwa, A. and Yadav, R. (2015), "Social CRM - An Emerging Medium", *Advances in Computer Science and Information Technology*, Vol. 2 No. 11, pp. 48-52.

• Tiruwa, A., Yadav, R. and Suri, P.K. (2014), "Going Viral-an Effective Strategy for Marketing", *International Conference on Research and Sustainable* Business, IIT Roorkee, pp. 598-603.

• Tiruwa, A., Yadav, R. and Suri, P.K. (2014), "Consumer Engagement Through Digital Media - A study on FMCG sector", *IBAIC-2014: International Conference, Indus Business Academy, Noida*, pp. 628-637.

Book Reviews

• Tiruwa, A. (2014), "Siddhartha by Herman Hesse", *Values-Based Management*, Vol. 4 No. 2, pp. 56-58.

• Tiruwa, A. and Yadav, R. (2014), "Awakening India by Swami Vivekanand", *Values-Based Management*, Vol. 4 No. 1, pp. 73-76,

Abstracts

• Tiruwa, A., Yadav, R., & Suri, P. K. (2017). "Blog Marketing - Potentials and practices", 2nd Global Leadership Research Conference, pp. 131.

• Tiruwa, A., Yadav, R., & Suri, P. K. (2017). "A Sentiment Analysis Approach to understand the consumers' inclination towards the Facebook Brand Page (FBP) Posts", *17th Global Conference on Flexible Systems Management*, pp. 45.